

MONETARIA

English Edition Volume VI, Number 1 January-June 2018



Monetaria

Editorial Board

Andrés R. Bianchi Larre (CHILE) Jorge Braga de Macedo (PORTUGAL) Guillermo Calvo (Argentina) Vittorio Corbo (CHILE) José Fernando De Gregorio R. (CHILE) Jordi Galí (SPAIN) Francisco de Paula Gutiérrez Gutiérrez (Costa RICA) Arnold C. Harberger (USA) Gustavo Loyola (BRAZIL) Miguel Mancera Aguayo (MEXICO) Enrique Mendoza (MEXICO) Guillermo Ortiz (MEXICO) Manuel Ramos-Francia (MEXICO) Miguel Urrutia Montoya (COLOMBIA) Carlos Végh (USA) Richard Webb (PERU) Marion Williams (BARBADOS)

Editorial Committee

Ex officio Members José Contreras, BANCO CENTRAL DE VENEZUELA, *President of the Editorial Committee* Elías Albagli, BANCO CENTRAL DE CHILE Adrián Armas Rivas, BANCO CENTRAL DE RESERVA DEL PERÚ

Óscar Carvallo, CEMLA Daniel Isaac Chiquiar Cikurel, Banco de México André Minella, Banco Central do Brasil Pablo Andrés Neumeyer, Banco Central de la República Argentina Jorge Hernán Toro Córdoba, Banco de la República, Colombia

Nonpermanent Members

Enrique Alberola Ila, BIS Fausto Hernández Trillo, Centro de Investigación y Docencia Económicas (Mexico) Isabel Horta Correia, Banco de Portugal Emanuel Kohlscheen, BIS

Technical Committee

Sergio Afonso Lago Alves, Banco Central do Brasil Alberto Ortiz, cemla Carolina Pagliacci, IESA (Venezuela) María José Roa, cemla Martín Tobal, Banco de México Marco Vega, Banco Central de Reserva del Perú

MONETARIA

Volume VI, number 1, January-June 2018

TABLE OF CONTENTS



Monetaria is an academic journal published by CEMLA. It provides an international forum for publishing original current high level economic research, both in Spanish and English (edited in two volumes), covering both theoretical and empirical issues and, preferably, using these jointly to possibly influence the design of public policies. It is available printed and electronically <www.cemla.org>. For articles and essays submission: cpublicaciones@cemla.org>. The opinions expressed in Monetaria are those of the authors, and do not necessarily reflect those of the Center and its authorities, the Editorial Committe, or the organization to which the authors are affiliated. Monetaria material may not be reproduced in any form without the written permission of the publisher. Indexed in: EBSCO and Research Papers in Economics (RePec).

Inventory Adjustments to Demand Shocks under Flexible Specifications

Carlos R. Barrera Chaupis

Asymmetries of the Exchange Rate Pass-through to Domestic Prices in Costa Rica during the Exchange Rate Flexibility Period

> Carlos Brenes Soto Manfred Esquivel Monge

Abstract

This article analyses the exchange rate pass-through to domestic prices in Costa Rica during the current exchange rate flexibility period and tests whether there is evidence of asymmetry. To this end, we estimate structural distributed lag models that encompass symmetric and asymmetric data generating process in line with Kilian and Vigfusson (2011). We found evidence of sign asymmetry in the bivariate relationship between inflation and exchange rate and when controlling for interest rate differential and output gap.

Keywords: pass-through asymmetry, exchange rate, exchange rate flexibility.

JEL classification: E31, E37, E58.

C. Brenes Soto <bre>crease@bccr.fi.cr> and M. Esquivel Monge<esquivelmm@bccr.fi.cr>, researchers, Economic Research Department, Banco Central de Costa Rica.

1. INTRODUCTION

A nenvironment of free capital movement under an inflation targeting regime demands the monetary authority adopt exchange rate flexibility. Together with inflation commitments, said regime requires appropriate knowledge of the magnitude and time with which exchange rate (ER) movements are transmitted to domestic prices, i. e., the exchange rate pass-through (ERPT). Properly understanding ERPT requires determining whether it exhibits sign or magnitude asymmetries. Abstracting this type of nonlinearities can result in the estimation of pass-through levels different from those actually occurring.

This article analyzes ERPT to prices in Costa Rica from March 2006 to April 2017 and tests the hypothesis that it presents asymmetries. We estimate structural distributed lag models that encompass symmetric and asymmetric data generating process in line with Kilian and Vigfusson (2011), employing data exclusively for the exchange rate flexibility period.

The importance of knowing the magnitude of the ERPT to prices lies in the predictive capacity of such changes and the time it takes the economy to transmit them to domestic prices. Besides determining the magnitude and lag with which they appear, it is important to establish the presence of sign and magnitude asymmetries in said phenomenon. Positive asymmetry means domestic prices react more to domestic currency depreciations, while negative asymmetry would imply a stronger response to appreciations. On the other hand, if the ERPT shows magnitude asymmetries, the response of domestic prices to ER shocks would depend on the size of such shocks.

The amount of ERPT can be related to many factors, including an economy's level of openness, the organizational structure of import sectors, the level and volatility of inflation, the level of flexibility in the exchange rate regime, etc. The exchange rate regime in Costa Rica varied significantly towards the end of 2006 when the fixed rate regime (crawling peg exchange rate) was replaced by increasingly more flexible regimes. In light of the fact that the aforementioned factors upon which the magnitude of the ERPT could depend are not fixed over time, it is reasonable to propose a hypothesis that there are asymmetries in said phenomenon.

Although the ERPT in Costa Rica has been studied previously, in most cases the models employed have assumed that the magnitude of the ERPT is constant over time. Moreover, the data samples employed always include observations from two very different exchange rate regimes. Hence, quantifying and verifying the presence of asymmetries only using data extracted from the exchange rate flexibility period (last 11 years) is relevant given that it could provide estimates for the phenomenon more in line with the current economic situation. Furthermore, before 2006, when the period of exchange rate flexibility began, the exchange rate regime in force (crawling peg) fostered very few episodes of nominal appreciation, meaning the data were not optimal for studying sign asymmetries in the ERPT. Since the end of 2006 there has been a larger degree of freedom in exchange rate movements, there is a relatively greater number of appreciation periods and, therefore, more data for studying asymmetries.

The paper is organized as follows: after this introduction, Section 2 describes the most important background literature and the evolution of methodologies employed in its analysis. Section 3 details the conceptual framework of the methodological approximating used for testing the proposed hypothesis. Next, Section 4 examines methodological aspects, the data and the econometric approach used. Section 5 presents the main results and, finally, Section 6 lists the most important conclusions.

2. BACKGROUND

Empirical literature on the ERPT generally presents more evidence of symmetry for industrialized countries (see Taylor, 2000; Goldfang and Werlang, 2000; Choudhri and Hakamura, 2001; and Engel, 2002), while for emerging economies the linearity assumption does not seem appropriate [see Winkelried (2003), Wang and Guo (2016) and Mendoza (2012)].

Among recent studies that make flexible the linearity assumption, Przystupa and Wróbel (2011) analyze the case of Poland. The authors observe that pass-through varies according to the stage of the business cycle, identifying it as smaller during contractionary periods and larger during expansions. Moreover, for ER fluctuations below a certain magnitude (2%), the pass-through differs from the other observations. They also find that the ERPT is greater during periods of low volatility (understood as a standard deviation of the daily variation below 4.32%).

Pérez and Vega (2016), meanwhile, find evidence for sign asymmetry in the ERPT of Peru. The authors also provide evidence of a different behavior for each exchange rate regime in the period studied.

Lariau, El Said and Takebe (2016) review evidence for the cases of Angola and Nigeria. They find that the ERPT is higher over the long term for the less diversified more import-dependent economy (Angola). They also demonstrate that dedollarization in Angola led to a decline in the ERPT. Furthermore, over the short term the ERPT is not statistically different from zero, which according to the authors reveals distortions caused by protectionism afforded to certain industries. For Nigeria, they show that the food and drinks component of the CPI is not affected by changes on the ER given the large share of domestic production in that index grouping. The research reflects the importance of countries' domestic consumption structure for determining the ERPT. Angola and Nigeria are similar countries with regard to their dependence on crude oil exports and they

also implement similar actions to offset possible price shocks in that commodity; despite this, the results reveal different ERPTs.

The Banco Central de Costa Rica has made significant research efforts to improve understanding of the ERPT. Such endeavors span from the fledgling estimations of León, Morera, and Ramos (2001) and León, Laverde, and Durán (2002), up to more recent papers such as those of Rodríguez (2009), Esquivel and Gómez (2010) and Orane (2016). Most of those studies employ the implicit assumption of linearity in the ERPT, estimating it with VAR models. The exception is Esquivel and Gómez (2010), who address the matter using an alternative methodology (LSTVAR) that considers the possibility of some variables inducing sign or magnitude asymmetries in the pass-through. The authors find that the lagged variation of oil prices is the variable most likely to induce asymmetries. Nevertheless, they conclude that there is little evidence of statistically significant sign or magnitude asymmetries.

Meanwhile, Esquivel and Gómez (2010) use a data sample between January 1991 and June 2009. In Costa Rica, the fixed exchange rate regime (crawling peg) was substituted in October 2006 by a flexible regime (exchange rate band), which was subsequently replaced by a managed float regime in February 2015. In view of this, there are at least three events to justify and make important a new study on the ERPT and its possible asymmetries.

First, the observations used in Esquivel and Gómez (2010) combine some (the majority) extracted from the period of fixed ER with others from the flexible phase. It should be taken into account that the crawling peg regime implied a systematic bias towards positive variations in the nominal ER (colones per US dollar). Only 15% of the observations used in that study are not affected by said bias. At present, the abundance of observations for the period after adoption of the flexible ER regime allows for considering estimations of the pass-through and statistical tests for asymmetry that use a sample with observations exclusively from the flexible regime. Second, there is a large body of documented evidence that the series of variation of the CPI in Costa Rica underwent a structural change during 2009. It is possible that said structural change has influenced the magnitude and characteristics of the ERPT. The data set used in the paper of 2010 evidently does not allow for capturing said phenomenon.

Finally, to provide additional robustness to the test for asymmetries in the ERPT, it is wise to apply alternative estimation methodologies. A traditional approach for measuring asymmetries uses censored VAR models. Applied to the topic of ERPT, the aforementioned method would imply estimating a VAR model where ER variations with a negative sign are censored from the sample and another where positive variations are censored. Subsequently, the impulse response (IR) functions of both models would be compared in order to conclude whether they are statistically different or not.

It is well documented in the literature on static models that censoring explanatory variables causes ordinary least square estimators to be biased, as described in Rigobon and Stoker (2009) or Greene (2003).

Although the bias observed in those procedures is clear when the data generating process (DGP) is symmetric, asymptotic bias continues even when the DGP is asymmetric. Just as stated by Kilian and Vigfusson (2011), only when the DGP is such that it does not exercise an impact on the dependent variable when the explanatory variable decreases can one guarantee that the censored linear model is not biased. In their study, those authors demonstrate that censored VAR models generate asymptotic biases and propose a structural model to prevent them. Their model encompasses symmetric and asymmetric data generating processes as special cases. Combined with the proposal of Lee, Ni and Ratti (1995), in which shocks should be rescaled by a volatility measure before performing an estimation of the pass-through, it is not only possible to diagnose the presence of sign and magnitude asymmetries, but also to determine whether the pass-through is smaller in periods of high volatility. Álvarez and Esquivel (2016) apply this method to assess the presence of asymmetries in the pass-through of commodity prices to domestic prices in Costa Rica.

In the original work of Kilian and Vigfusson (2011), the authors estimate the impact of energy price shocks on economic growth, proposing two statistical tests for applying to the hypothesis of symmetry in the response of growth. One of them is conducted on regression coefficients and is a variation of that proposed by Mork (1989) but with higher statistical power. The other is applied directly to the IR functions. The latter is based on the fact that, as postulated in Koop, Pesaran and Potter (1996), in nonlinear VAR models the magnitude of shocks can influence the dynamic response of the variables. Moreover, under this same context, the dynamic response of a variable can exhibit asymmetries even if the coefficients do not exhibit departures from symmetry.

In addition to this problem, traditional empirical literature on censored VAR models also has the disadvantage of ignoring that, by being nonlinear models, IR functions depend on the history of observations [see Koop, Pesaran, and Potter (1996), and Gallant, Rossi, and Tauchen (1993)]. IR functions in this type of models require a Monte Carlo simulation in order to include possible data histories and different sizes of shocks.

3. CONCEPTUAL FRAMEWORK

Kilian and Vigfusson (2011) show that when the DGP is not symmetric it cannot be represented as a bivariate VAR model for x_t^+ and y_t . A DGP where only positive shocks to x_t have an impact on y_t can be denoted with the following system:

$$x_{t} = a_{1} + \rho x_{t-1} + e_{1t},$$

$$y_{t} = a_{2} + \gamma x_{t}^{+} + e_{2t}.$$

The contemporaneous effect on y_t of a positive shock to x_t in System 1 is given by γ . The impact in the subsequent period

1

would be $\rho\gamma$, and then $\rho^2\gamma$, and so on successively thereafter. Thus, estimation of coefficients γ and ρ of Model 1 would be unbiased. By using a censored VAR model such as Model 2, estimation of ρ would be asymptotically biased despite the fact that the estimation of γ would be unbiased. This would be reflected in the IR function.

2

$$x_{t}^{+} = a_{1} + \rho x_{t-1}^{+} + \epsilon_{1t},$$

$$y_{t} = a_{2} + \gamma x_{t-1}^{+} + \epsilon_{2t}.$$

The problem with System 2 is that it is not a true representation of the DGP. Use of a full structural model would avoid that drawback. Kilian and Vigfusson (2011) propose the following model:

3

$$x_{t} = a_{1}x_{t-1} + a_{2}y_{t-1} + \dots + \epsilon_{1t},$$

$$y_{t} = \beta_{1}x_{t}^{+} + \beta_{2}x_{t-1}^{+} + \beta_{3}y_{t-1} + \dots + \epsilon_{2t},$$

System 3 is a structural model where, unlike Model 2, negative shocks to x_i can affect the future path of y_i if such shocks eventually lead to positive shocks in the future path of x_i .

System 4 is the reduced form of 3. The IR functions of a structural model such as 3 cannot be identified from a Cholesky decomposition of the variance-covariance matrix and its reduced version because such a composition does not discriminate between positive and negative shocks. Hence, applying Cholesky in 4 to $Var[\epsilon_{1t}, u_{2t}]$ is not appropriate given that u_{2t} should only reflect positive shocks.

4
$$x_{t} = a_{1}x_{t-1} + a_{2}y_{t-1} + \dots + \epsilon_{1t},$$
$$y_{t} = \beta_{1}x_{t-1}^{+} + \beta_{2}y_{t-1} + \dots + u_{2t},$$

where $u_{2t} = \beta_1 \epsilon_{1t} + \epsilon_{2t}$.

Additional technical details on the conceptual proposal and tests for the absence of asymptotic bias in Model 3 can be

l

consulted in the paper referred to (Kilian and Vigfusson, 2011). The points summarized here motivate the use of the methodology proposed by those authors to verify the presence of asymmetries in the exchange rate pass-through.

4. METHODOLOGY

4.1 Estimation of Impulse Response Functions in Asymmetric Structural Models

We propose a structural model where the endogenous variables in an equation system are used to allow exchange rate shocks to have a varied impact on prices in an economy depending on whether the currency is appreciating or depreciating.

In an initial approach using a bivariate model, the structure would be written as follows:

$$\begin{aligned} x_t &= a_1 x_{t-1} + a_2 y_{t-1} + \ldots + \epsilon_{1t}, \\ y_t &= \beta_1 x_t^+ + \beta_2 x_{t-1}^+ + \beta_3 y_{t-1} + \ldots + \epsilon_{2t}, \end{aligned}$$

where

5

- x_t is the level or variation of the ER in period t.
- y_t is the level or variation of the CPI in period t.

•
$$x_t^+ = \begin{cases} x_t, \text{ si } x_t > 0\\ 0, \text{ si } x_t \le 0 \end{cases}$$

In contrast to a censored VAR, in which the endogenous variables correspond to x_t^+ and y_t , in the proposed Model 5 negative shocks to x_t can affect the future path of y_t if they eventually lead to positive shocks in the future path of x_t . The authors of the reference study demonstrate that the estimators of this model are asymptotically unbiased, unlike those obtained using censored VAR models, regardless of whether the DGP is symmetric or not.

According to different studies (see Gallant, Rossi, and Tauchen, 1993; and Koop, Pesaran, and Potter, 1996), in nonlinear models such as 5, the dynamic response of y_t could be magnified or reduced by the accumulated effect of previous shocks. Hence, IR functions should be estimated as an average of the impulse responses generated based on a data set that is both diverse and representative of initial conditions. We estimate IR functions following the sequence of steps shown below:

- 1) Random selection of a *history* (Ω_i) composed of consecutive *p* values of x_i and y_i .¹
- 2) Given Ω_i , simulate two-time paths for H data after the last observation available for x and y. That is, for x we generate $\begin{bmatrix} x_{t+1}, x_{t+2}, \dots, x_{t+H} \end{bmatrix}$ and $\begin{bmatrix} x_{t+1}^*, x_{t+2}^*, \dots, x_{t+H}^* \end{bmatrix}$, while for y we generate $\begin{bmatrix} y_{t+1}, y_{t+2}, \dots, y_{t+H} \end{bmatrix}$ and $\begin{bmatrix} y_{t+1}^*, y_{t+2}^*, \dots, y_{t+H}^* \end{bmatrix}$. For the first paths of de x and y, as well as the second of y, stochastic disturbances $[\epsilon_{1t}, \epsilon_{1t+1}, \dots, \epsilon_{1t+H}]$ and $[\epsilon_{2t}, \epsilon_{2t+1}, \dots, \epsilon_{2t+H}]$ are randomly selected from their respective marginal empirical distributions. Furthermore, for the second sequence of x, the value (δ) is assigned to the first component of the sequence of disturbances, $(\epsilon_{1t} = \delta)$, while the rest of the sequence is randomly extracted from its marginal empirical distribution.
- 3) Random sequences of ϵ_{1t} and ϵ_{2t} can be treated as independent given that they are obtained from the marginal distribution generated by estimated structural Model 5.
- 4) We proceed to obtain the difference between two paths of y for t=1, 2, ..., H, defining each difference as y_i^{δ} , where i=1, 2, ..., H.
- 5) Steps 2 and 4 are repeated (n_{boot}) times.
- 6) Steps 1 to 5 are repeated 1 to 5 (n_{hist}) times. We, therefore, obtain a number $n_{hist} * n_{hist}$ for different series y_i^{δ} that are then averaged.

¹ p corresponds to the number of lags used for each model estimated.

The result obtained from steps 2 to 5 is the response of y to a shock of size δ , over a horizon of H periods and conditional on Ω_i . Following the same nomenclature of Kilian and Vigfusson (2011), we can define this response as $I_y(\delta, H, \Omega_i)$. Repeating the exercise for all possible histories and averaging the responses, we obtain the response of y unconditional on Ω_i , that is, $I_y(\delta, H)$.

To more clearly differentiate the proposal of Kilian and Vigfusson (2011) regarding the traditional way of obtaining the IR functions, we define the response of y conditional on the historical paths of x and y (that is $x_{t-i} = y_{t-i} = 0$ for i=1, 2, ...) as follows:

$$I_{y}(\delta,H,\underline{0}).$$

Relaxing the assumption of $x_{t-i} = y_{t-i} = 0$ and allowing a history (Ω_i) for *x* and *y*, besides inducing a shock of magnitude δ in the *t*-th observation of disturbance term ϵ_1 , we can alternatively define the response:

$$I_{y}^{*}(\delta, H, \Omega_{i}) = E\left\{y_{t+h} \mid \Omega_{i}, \epsilon_{1t} = \delta, \left[\epsilon_{1t+j}\right]_{j=1}^{h}, \left[\epsilon_{2t+j}\right]_{j=0}^{h}\right\} - E\left\{y_{t+h} \mid \Omega_{i}, \left[\epsilon_{1t+j}\right]_{j=0}^{h}, \left[\epsilon_{2t+j}\right]_{j=0}^{h}\right\}.$$

As mentioned previously, by averaging 7 for all possible histories, we obtain the unconditional response in Ω_i , which corresponds to $I_y^*(\delta, H)$. The impulse response normally obtained in the literature corresponds to $I_y^*(\delta, H, \underline{0})$. This IR does not allow future shock dynamics (at least in disturbances) and does not condition history. In linear systems, this type of configuration for the calculations does not present any drawbacks. However, they do present them when computing IR in nonlinear systems: The response may not converge to zero even when the DGP is stationary (see Koop, Pesaran, and Potter, 1996). Moreover, Potter (2000) opts for considering future shocks as

6

7

random rather than fixing them at zero when estimating nonlinear IRs. Finally, due to the lack of realism in conditioning an IR estimation at zero, this is not very useful.

In reduced-form VAR equations the errors are correlated. Based on this we use a method for orthogonalizing the impulses. The usual approach is to employ an inverse Cholesky factorization of the variance-covariance matrix of the estimation residuals. A structural model such as 5 used in this research becomes more attractive for estimating IR functions given that in $I_y(\delta, H, \Omega_i)$ and $I_y(\delta, H)$ calculations, an exchange rate shock (x_i) is orthogonal to other shocks.

Kilian and Vigfusson (2011) show that, for small shocks, the difference between the IR estimated considering possible histories as well as the behavior of errors $\begin{bmatrix} I_y^*(\delta, H) \end{bmatrix}$, and that estimated without considering those two items $\begin{bmatrix} \overline{I}_y^*(\delta, H, \underline{0}) \end{bmatrix}$, is substantial. Nonetheless, this difference declines as the size of the shock increases, i. e., the authors demonstrate that

$$\lim_{n\to\infty}\frac{1}{n}I_{y}(n\delta,H)=I_{y}^{*}(\delta,H,\underline{0})$$

For exchange rate shocks of a sufficiently large magnitude, we would expect that the importance of Ω_i and the randomness of ϵ_{1i} decrease until reaching the point at which the IR estimated using the traditional VAR approach is a good approximation to correct estimation. This is, therefore, the explanation of how the traditional VAR method can generate estimations for the response of domestic prices to exchange rate shocks that are very different from those correctly estimated through a nonlinear specification.

This inverse relationship between the size of shocks and the estimated response of domestic prices is important given that, for series where the variation (in this case of the exchange rate) exhibits a small standard deviation, the advantage of using $I_y(n\delta, H)$, in terms of reducing asymptotic bias in IR function measurement, is greater.

8

4.2 Symmetry Tests

Despite solving the problem of asymptotic bias with respect to a censured VAR, structural model 5 is asymptotically inefficient compared to a VAR when the DGP is symmetric. Hence, efficient ERPT estimation requires a prior statistical test to evaluate the hypothesis of symmetry in the DGP.

Those defined below as tests of symmetry in parameters assess the equality of the magnitude of coefficients associated with appreciations and depreciations.

Kilian and Vigfusson (2011) show that these tests are useful for reduced-form models to identify asymmetries in parameter responses. Nonetheless, they are not useful for identifying asymmetries in the IR functions of asymmetric structural models. This is due to the fact that they could obtain parameters associated with appreciations and depreciations that are not statistically different, while the IRs are indeed so. The latter because IR functions can be a nonlinear function of both the slope parameters and the variance of the innovations.

In light of this problem, Eldstein and Kilian (2007) suggest an alternative approximation based on the IR functions obtained according to the method explained in Section 4.1 to test the symmetry hypothesis. We refer to this second group of tests as tests of symmetry in the IRs.

4.2.1 Tests of Symmetry in Parameters

Tests for symmetry in parameters, or slope-based symmetry tests, are attractive given their simplicity and because they do not require the computation of IR functions. According to this method, after estimating the regression of y_t on its own lags as well as those on x_t^+ and x_t^- , we test the equality of the coefficients by means of Wald test statistics that, under the null hypothesis of symmetry, have distribution Ji^2 [see Mork (1989)].

Kilian and Vigfusson (2011) show that this approximation does not exploit all the restrictions implied by the null hypothesis of symmetry. They demonstrate that, by working with a reduced model, Mork (1989) omits the equality restriction of the contemporaneous terms of x_t^+ and x_t^- . The authors, therefore, propose, in terms of Model 5, working with the null hypothesis

$$H_0: \beta_1 = \beta_2 = 0.$$

The same authors argue that this hypothesis has higher statistical power than that of Mork (1989). They test this hypothesis in a model such as 5, and by means of parameter exclusion Wald tests seek to determine whether the fit of the model improves with the inclusion of regressors x_t^+ , x_{t-1}^+ ,..., x_{t-b}^+ .

4.2.2 Tests of Symmetry in IR Functions

The proposal of Kilian and Vigfusson (2011), adapted for testing sign symmetry in IR functions for prices in the presence of exchange rate shocks to h over different horizons can be summarized in the following steps:

- 1) Estimate structural Model 5.
- 2) Calculate IR *h* periods ahead (in this case it was performed with a horizon of 24 periods) for both positive and negative shocks. That is, calculate $I_y^*(\delta,h)$ and $-I_y^*(-\delta,h)$.
- 3) Construct a Wald test of the joint null hypothesis of symmetry in positive and negative IRs up to a horizon of *h* periods in the future. The statistic, therefore, takes the form: $W = \sum_{i=0}^{h} \left[I_{y}^{*}(\delta,i) + I_{y}^{*}(-\delta,i) \right]^{2} = 0.$
- 4) Estimate the variance-covariance matrix of the vector sum of response coefficients by bootstrap simulation.

The W statistic, therefore, has distribution $\int_{h+1}^{2} given the as$ ymptotic normality of the parameter estimators of the model.

4.3 Data

The database employed in the estimations corresponds to series published by the Banco Central de Costa Rica on its official online data portal.² Basic exchange rate data sets have a daily frequency, but a monthly series was constructed by taking the average between the purchase and sale references on every business day each month. Meanwhile, the series for the CPI are originally monthly.

As controls in the estimations, we included indicators on output gap and interest rate differentials. The base information for the output gap is the seasonally adjusted series of the monthly economic activity index (IMAE). We applied a Hodrick-Prescott filter to this with smoothing parameter $\lambda = 23.000$ in line with Segura and Vásquez (2011).

Finally, the series for interest rate differentials considers the United States Treasury federal funds effective rate³ and the monetary policy rate of the Banco Central de Costa Rica. The sample period spans from January 2006 to March 2017.

5. RESULTS

5.1 Evaluation of Stationary Properties

The stationary properties of the series employed are determined in order to define the type of econometric method with which to perform the prior analysis. The results of the unit root tests applied are displayed in Table 1. It can be seen that both under the Dickey-Fuller (DF) test and that of Phillips-Perron (PP), it is not possible to reject the null hypothesis of a unit root for all the series at levels, except for the IMAE gap. In the case of the first difference, the null hypothesis of a unit root is

² <https://www.bccr.fi.cr/seccion-indicadores-economicos/indicadores-econ%C3%B3micos>.

³ <https://fred.stlouisfed.org/series/FEDFUNDS>.

			Variable				
Variable in:	Type of test	Specification	CPI	ER	Interest rate differential	IMAE gap	
Levels	ADF	Const	0.99	0.27	0.72	0.00	
		Const and trend	0.99	0.55	0.91	0.00	
	РР	Const	0.98	0.31	0.47	0.00	
		Const and trend	1.00	0.60	0.77	0.00	
First difference	ADF	Const	0.00	0.00	0.00	0.00	
		Const and trend	0.00	0.00	0.00	0.00	
	PP	Const	0.00	0.00	0.00	0.00	
		Const and trend	0.00	0.00	0.00	0.00	
Source: Author	s' calcula	ations.					

Table 1P VALUES IN UNIT ROOT TESTS (H₀: X_T HAS UNIT ROOT)

rejected for all the series. Based on these results, all the variables in the estimations were used in first differences, except the IMAE gap, which was kept at levels.

5.2 Lag Order

We proceeded to determine the most appropriate lag order for estimating Model 5 in two ways. Firstly, based on VAR model lag selection criteria and secondly using goodness-of-fit criteria for the equation of y_t (price equation in the application of this paper) in asymmetric structural Model 5. The selection was made for three different model specifications: one bivariate model (consisting of the first difference of the CPI and the exchange rate); two models of three variables constructed based on the bivariate model adding the IMAE gap and interest rate differential, respectively. Table 2 displays the results for those models under five different criteria.

Table 2

		Model				
Specification	Criteria	Bivariate	Bivariate + interest rate differentials	Bivariate + IMAE gap		
	LR	5	1	3		
	FPE	1	1	1		
VAR	AIC	1	1	1		
	SC	1	1	1		
	нQ	1	1	1		
Asymmetric prices	AIC	5	5	5		
equation	SC	1	1	1		

OPTIMAL NUMBER OF LAGS ACCORDING TO DIFFERENT CRITERIA

Note: LR stands for likelihood ratio, FPE to final prediction error, AIC to Akaike information criterion, SC to Shwarz's criterion, and HQ to that of Hannan-Quinn.

Source: Authors' calculations.

In general, the specification that includes only one lag tends to dominate both in the criteria for the VAR model and for the equation of y_t in the asymmetric structural model, regardless of whether the model is bivariate or incorporates interest rate differentials or the IMAE gap. It should be emphasized, however, that based on the AIC, the model with five lags dominates all the cases for the equation of y_t in the asymmetric structural model.

The results presented here are useful for assessing the evidence on asymmetric effects shown in the following section, where tests of symmetry in parameters and in IR functions for models with up to 12 lags are revealed. Furthermore, the IR functions presented below for measuring the exchange rate pass-through correspond precisely to the specifications with lag order selection based on the evidence in Table 2.

Table 3

		Type of model	
Lags	Bivariate	Trivariate with interest rate differentials	Trivariate with IMAE gap
1	0.29	0.43	0.19
2	0.64	0.85	0.46
3	0.48	0.71	0.44
4	0.71	0.87	0.58
5	0.55	0.61	0.38
6	0.58	0.56	0.41
7	0.33	0.28	0.39
8	0.24	0.25	0.23
9	0.07	0.13	0.15
10	0.07	0.11	0.10
11	0.10	0.20	0.08
12	0.11	0.32	0.07

P VALUE IN TEST OF PARAMETER SYMMETRY (H_a: SYMMETRIC PASS-THROUGH)

Note: Cases with the rejection of the H_0 at 10% are highlighted in bold. Source: Authors' calculations.

5.3 Symmetry Tests

5.3.1 Test of Symmetry in Parameters

The results of the test of symmetry in the parameters, explained in Section 4.2.1, are shown in Table 3. As mentioned previously, they include the models that consider from 1 up to 12 lags. As can be seen, for models identified as having better goodness-of-fit (with 1 and 5 lags) there is not sufficient evidence to reject the null hypothesis of symmetric pass-through either in the bivariate case or trivariate ones. Nonetheless, it is interesting to see that the inclusión of additional lags (above 9) tends to increase the evidence against the hypothesis of symmetry, at least for the bivariate and trivariate models that include IMAE gap.

5.3.2 Test of Symmetry in Impulse Response Functions

The results from applying the test of symmetry on IR functions, the methodology for which was described in Section 4.2, can be seen in Table 4. The results were obtained by simulating 40,000 forecasts of structural Model 5 with a horizon of up to 24 months.⁴ It is worth remembering that the variables involved are, alternatively, the first difference of the CPI and the first difference of the nominal ER (bivariate case), adding IMAE gap and interest rate differentials for the models denominated trivariate. In view of the fact that the nonlinearity of IR functions may appear on any horizon, the table contains p values for each forecasting horizon from 1 up to 24 months.

In general, the results do not lead to very different conclusions than those obtained from the tests of symmetry in parameters. For the models with better goodness-of-fit (those that include 1 and 5 lags), the evidence against the symmetry hypothesis is scarce in all models and for all horizons. Table 4 also displays the results for the model with most evidence against the symmetry hypothesis (the version that includes up to 12 lags). In this case, and at 10% significance, the bivariate model at horizons of between four and six months, and the trivariate model with interest rate differentials for horizons above ten months, exhibit some evidence in favor of the alternative hypothesis of an asymmetric response in domestic prices to exchange rate shocks. Nevertheless, it should be emphasized that goodness-of-fit criteria do not favor this specification.

The fact that the greatest evidence of asymmetric passthrough is found when the model estimated includes 12 lags (trivariate model with interest rates differentials) might be because the estimations do not take into account seasonal factors.

⁴ See procedure explained in Section 4.1.

4	(
٩)
	2
5	1
h	

P VALUE IN TEST OF SYMMETRY IN IMPULSE RESPONSE FUNCTIONS (H₀: SYMMETRIC PASS-THROUGH)

				W	odel Specificat	ion			
		Bivariate		Bivariate wi	ith interest rate	differentials	Bivan	iate with IMA	$E \ gab$
Horizon	1 lags	5 lags	12 lags	I lags	5 lags	12 lags	I lags	5 lags	12 lags
1	0.19	0.10	0.16	1.00	0.95	0.35	0.97	0.97	0.96
0	0.35	0.16	0.10	1.00	0.98	0.12	1.00	1.00	0.98
3	0.55	0.29	0.17	1.00	1.00	0.07	1.00	1.00	1.00
4	0.68	0.30	0.04	1.00	1.00	0.12	1.00	1.00	1.00
ũ	0.78	0.41	0.03	1.00	1.00	0.20	1.00	1.00	1.00
9	0.86	0.47	0.05	1.00	1.00	0.25	1.00	1.00	1.00
7	0.92	0.57	0.08	1.00	1.00	0.34	1.00	1.00	1.00
×	0.96	0.68	0.12	1.00	1.00	0.30	1.00	1.00	1.00
6	0.98	0.76	0.12	1.00	1.00	0.38	1.00	1.00	1.00
10	0.98	0.83	0.10	1.00	1.00	0.11	1.00	1.00	1.00

11	0.99	0.89	0.12	1.00	1.00	0.07	1.00	1.00	1.00
12	1.00	0.93	0.16	1.00	1.00	0.05	1.00	1.00	1.00
13	1.00	0.95	0.20	1.00	1.00	0.05	1.00	1.00	1.00
14	1.00	0.97	0.25	1.00	1.00	0.07	1.00	1.00	1.00
15	1.00	0.98	0.29	1.00	1.00	0.08	1.00	1.00	1.00
16	1.00	0.99	0.35	1.00	1.00	0.08	1.00	1.00	1.00
17	1.00	0.99	0.42	1.00	1.00	0.04	1.00	1.00	1.00
18	1.00	1.00	0.48	1.00	1.00	0.05	1.00	1.00	1.00
19	1.00	1.00	0.55	1.00	1.00	0.02	1.00	1.00	1.00
20	1.00	1.00	0.40	1.00	1.00	0.03	1.00	1.00	1.00
21	1.00	1.00	0.46	1.00	1.00	0.01	1.00	1.00	1.00
22	1.00	1.00	0.52	1.00	1.00	0.01	1.00	1.00	1.00
23	1.00	1.00	0.54	1.00	1.00	0.02	1.00	1.00	1.00
24	1.00	1.00	0.60	1.00	1.00	0.02	1.00	1.00	1.00
Note: Cases wit	h rejection of t	he $ m H_{_0}$ at 10%	are highlighte	d in bold.					

Nonetheless, visual examination of the correlograms, as well as simple tests in which the variables analyzed are regressed in fictitious seasonal variables, do not suggest the presence of this type of effects (see Figure A.1 and Table A.1 in the Annex).

5.4 Quantification of Exchange Rate Pass-through to Prices

In this section, we quantify the ERPT estimated using structural Model 5. For each model (bivariate and the two model variations with three endogenous) IR function estimations were performed following the procedure described in Section 4.1, fixing $n_{bool} = n_{hist} = 200$, i. e., averaging 40,000 estimations at each horizon from 1 up to 24 months. The magnitude of these functions is shown as a proportion of the size of the original shock. Moreover, those corresponding to negative exchange rate shocks are shown multiplied by -1 to allow their magnitude to be easily compared with those corresponding to positive shocks. The confidence bands shown are empirical and correspond to percentiles 5 and 95 of the distribution of the 40,000 forecast simulations performed for each horizon and for each model specification.

They also display IR functions for four different sizes of ER shock (1, 2, 4 and 10 standard deviations), in order to analyze whether sign asymmetry could be associated to the size of the shocks, a matter that would not be evident in the tables presented in the previous section.

Figure 1 displays the IR functions obtained from the bivariate model that includes only one lag. The first point that should be mentioned is that the proportional magnitude of the pass-through during positive shocks (appreciations) ends up being between 22% and 35%, which is consistent with the most recent estimations based on linear methods.⁵ However,

⁵ See Orane (2016).

the pass-through in negative shocks is estimated to be around 15% for small shocks and close to 0% for larger shocks.

Meanwhile, with respect to matters of asymmetry, it can be seen that, for the case of small shocks (one standard deviation), the evidence is consistent with that shown in Table 4 in the sense that the dynamic response of prices is not statistically different in positive or negative ER shocks. Furthermore, in accordance with the size of the shock confidence bands for the estimations cease to overlap. Thus, for mid-sized and large shocks the response of prices does appear statistically different.

Figure 2 shows the IR functions obtained when the additionalvariable is incorporated into the model, specifically interest rate differentials. In terms of the proportional magnitude of the long-term pass-through we estimate, there is not much difference from the bivariate case. The pass-through is between 20% and 30% in depreciations, and between 0% (large shocks) and 15% (small shocks) in the case of appreciations.

Just as in the bivariate case, when the ER shock is small (one standard deviation), there is no significant difference in the dynamic response of domestic prices. Nonetheless, for larger shocks (four and ten standard deviations) the spaces between the confidence bands move apart during positive and negative shocks, indicating sign asymmetry in the response.

One pattern that can be extracted from the IR functions in Figure 1 and Figure 2 is that when ER shocks are small, the response of domestic prices is no different in the presence of appreciations or depreciations. However, when the shocks are mid-sized and large, the response during appreciations tends to decrease in proportional magnitude, eventually differing from the response during depreciations. One possible explanation for this behavior is that economic agents may interpret large appreciations as temporary phenomena that do not merit price adjustments. This could be caused by the historical trend (which has reverted during recent years) of inflation in Costa Rica being higher than in the country's main trading partners. The aforementioned meant the public became accustomed to Figure 1







Figure 2

increases in the nominal ER, and episodes of appreciations, particularly very large ones, tend to be seen as exceptions to the trend and therefore temporary.

Figures A.2 and A.3 in the Annex display the IR functions for the case of bivariate and trivariate models (with interest rate differentials) with five lags. Except for being necessary a horizon of over 18 months to illustrate convergence, the dynamic response pattern is similar to that observed in the figures mentioned here.

One item that can be extracted from the estimations performed, but that is not easily appreciable in Figure 1 or Figure 2, is that the magnitude of the pass-through is a growing function of the shocks when they are depreciations, but a decreasing function if they are appreciations. This is illustrated in Figure 3 corresponding to estimations using the trivariate model that includes interest rate differentials (the trend is the same in the case of the bivariate model). Note that for positive exchange rate shocks (upper panel of the figure) the dynamic response of domestic prices is larger than for smaller shocks. On the other hand, for negative shocks (lower panel of the figure), the smaller the shock, the larger the proportional response (in absolute value).⁶

As mentioned, this phenomenon could be explained by economic agents' expectations being rooted in considering episodes of appreciation in the domestic currency as unusual. If this were the case, negative exchange rate shocks, especially the largest ones, would be considered temporary and, possibly due to items such as menu costs, would not generate downward adjustments in prices in domestic currency.

⁶ As shown, IR functions appear multiplied by -1 in the presence of appreciations.



6. CONCLUSIONS

In general, the magnitude of exchange rate pass-through to prices is calculated to be between 20% and 35% in the case of depreciations. This estimation is similar in size to the most recent ones obtained by the Banco Central de Costa Rica employing linear methods. Nevertheless, those linear methods assume sign symmetry in the estimation. In this paper, we calculate that in the case of appreciations the magnitude of the pass-through is between 0% and 15 percent.

The dynamic response of the CPI to exchange rate shocks exhibits evidence of sign asymmetry only when the shocks are mid-sized or large.

For more common unexpected appreciations or depreciations (of one standard deviation), tests for asymmetry in parameters and in IR functions do not find sufficient evidence to reject the hypothesis of symmetry. Meanwhile, the empirical confidence bands for IR functions indicate that when the size of the appreciation or depreciation is mid-sized or large (four or more standard deviations), the response of domestic prices is greater (in absolute value) during a depreciation. Hence, it is not correct to assume a response of similar magnitude in domestic prices to appreciations than to deprecations when these are relatively large.

The size of the shock influences the proportional magnitude of the pass-through

When it comes to unexpected depreciations in the domestic currency, those of greatest magnitude are transmitted to a larger extent than smaller ones. Moreover, during unexpected appreciations, the largest ones are transmitted less to domestic prices. The evidence found in this research indicates that considering a constant pass-through regardless of the direction or magnitude of exchange rate shocks possibly leads to erroneous estimates for the impact of exchange rate variations on domestic prices.

ANNEX

Figure A.1

CORRELOGRAM AND PARTIAL CORRELOGRAM OF LOGARITHMIC FIRST DIFFERENCE OF THE CPI

Sample: 2006M1-2017M4 Observations: 13

Autocorrelation	Partial correlation		AC	PAC	Q statistic	Prob.
		1	0.493	0.493	33.504	0.000
1	1 1	2	0.300	0.076	46.029	0.000
	т 🗖	3	0.312	0.183	59.655	0.000
	1	4	0.174	-0.071	63.926	0.000
1	1	5	0.332	0.312	79.605	0.000
1	1 1	6	0.316	0.020	93.880	0.000
I .	1 1	7	0.167	-0.045	97.929	0.000
1	1	8	0.116	0.094	99.882	0.000
1	1 1	9	0.102	0.057	101.41	0.000
	1 1	10	0.147	0.054	104.59	0.000
1	I 🗌	11	0.277	0.184	116.01	0.000
1	1 1	12	0.230	-0.014	123.95	0.000
I	I I	13	0.172	0.050	128.45	0.000
1	I I	14	0.180	0.026	133.40	0.000
	1 1	15	0.156	0.044	137.16	0.000
1		16	0.174	-0.041	141.86	0.000
1	1 1	17	0.198	0.027	147.98	0.000
- I -	I I	18	0.252	0.151	158.02	0.000
	I I	19	0.164	-0.058	162.33	0.000
1	I I	20	0.074	-0.084	163.21	0.000
1 1	1 I I	21	0.065	-0.056	163.89	0.000
1 1	I I	22	0.023	-0.051	163.97	0.000
1	т <u>і</u> і	23	0.088	0.029	165.26	0.000
I 🗖 I	I I	24	0.155	0.076	169.27	0.000
-						

Source: Author's calculations.

Table A.1

STATIONARITY TEST WITH DICHOTOMOUS VARIABLES

Dependent variable: DLOGIPC Method: least squares Sample (adjusted): 2006M2-2017M4 Included observations: 135, after adjustments

Variable	Coefficient	Standard	Statistical probability t	Prohability
	0 5011	0.1950	4 7205	<u>1 100000111y</u>
C	0.5911	0.1250	4.7505	0.0000
DUMCE	0.6718	0.0818	8.2114	0.0000
(@SEAS(2))	-0.2532	0.1718	-1.4732	0.1433
@seas(3)	-0.6152	0.1718	-3.5801	0.0005
@SEAS(4)	-0.3096	0.1718	-1.8018	0.0740
@SEAS(5)	-0.1398	0.1756	-0.7959	0.4276
@SEAS(6)	-0.3344	0.1756	-1.9041	0.0592
@SEAS(7)	-0.1404	0.1756	-0.7994	0.4256
@SEAS(8)	-0.2920	0.1756	-1.6627	0.0989
@SEAS(9)	-0.7188	0.1756	-4.0935	0.0001
@seas(10)	-0.6174	0.1756	-3.5159	0.0006
@seas(11)	-0.2678	0.1756	-1.5249	0.1299
@seas(12)	-0.1626	0.1755	-0.9267	0.3559
\mathbb{R}^2		0.4498	Mean of the dependent variable	0.4377
Adjusted R ²		0.3956	Standard deviation of the dependent variable	0.5293
Standard error of the regression		0.4115	Akaike criteria	1.1532
Residual sum of squares		20.6554	Schwarz criteria	1.4329
Log likelihood		-64.8392	Hannan-Quinn criteria	1.2669
Statistical me	asure of F	8.3102	Durbin-Watson statistic	1.3304
Probability (s measure of	statistical F)	0.0000		

Source: Authors' calculations.



Figure A.2

Figure A.3

IMPULSE RESPONSE FUNCTIONS OF PRICES TO EXCHANGE RATE BY SHOCK SIZE


References

- Álvarez, Cristian, and Manfred Esquivel (2016), Efecto de cambios de precio en materias primas sobre los precios domésticos en Costa Rica, ¿influye el nivel de competencia en los mercados?, Documento de Investigación, No. DEC-DIE-06-2016, Banco Central de Costa Rica.
- Bernanke, Ben, Mark Gertler, and Mark Watson (1997), Systematic Monetary Policy and the Effects of Oil Price Shocks, Brookings Papers on Economic Activity, No. 1, pp. 91-157.
- Choudhri, E., and D. Hakamura (2001), Exchange Rate Pass Through to Domestic Prices: Does the Inflationary Environment Matter?, FMI Working Paper, No. 01/194.
- Eldstein, Paul, and Lutz Kilian (2007), "The Response of Business Fixed Investment to Changes in Energy Prices: A Test of Some Hypotheses About the Transmission of Energy Price Shocks," *The B. E. Journal of Macroeconomics*, Vol. 7, No. 1, pp. 1-41.
- Engel, Charles (2002), "Expenditure Switching and Exchange-rate Policy," in M. Gertler y K. Rogo (eds.), *NBER Macroeconomics Annual*, pp. 231-272.
- Esquivel, Manfred, and José Fabio Gómez (2010), Asymmetries of the Exchange Rate Pass Through to Domestic Prices: The Case of Costa Rica, Documento de Investigación, No. DEC-DIE-043-2010, Banco Central de Costa Rica.
- Gallant, Ronald, Peter E. Rossi, and George Tauchen (1993), "Nonlinear Dynamic Structures, "*Econometrica*, vol. 61, núm. 4, pp. 871-907.
- Goldfang, I., and S. Werlang (2000), The Pass-through from Depreciation to Inflation: A Panel Study, Working Paper, No. 5, Banco Central do Brasil.
- Greene, William H. (2003), *Econometric Analisys*, Prentice Hall, Upper Saddle River, N.J.
- Hamilton, James D. (1996), *Time Series Analisys*, Princeton University Press, Chichester, United Kingdom.
- Kilian, Lutz, and Robert J. Vigfusson (2011), "Are the Responses of the U.S. Economy Asymmetric in Energy Price Increases and Decreases?," *Quantitative Economics*, Vol. 2, pp. 419-453.
- Koop, Gary, M. Hashem Pesaran, and Simon M. Potter (1996), "Impulse Response Analysis in Nonlinear Multivariate Models," *Journal of Econometrics*, Vol. 74, No. 1, pp. 119-147.

- Lariau, A., M. El Said, and M. Takebe (2016), An Assessment of the Exchange Rate Pass-through in Angola and Nigeria, FMI Working Paper, No. 16/191.
- Leduc, Sylvain, and Keith Sill (2004), "A Quantitative Analysis of Oil Price Shocks, Systematic Monetary Policy and Economic Downturns," *Journal of Monetary Economics*, Vol. 51, No. 4, pp. 781-808.
- Lee, Kiseok, Shawn Ni, and Ronald A. Ratti (1995), "Oils Shocks and the Macroeconomy: The Role of Price Volatility," *Energy Journal*, Vol. 16, No. 39-56.
- León, Jorge, Ana P. Morera, and Welmer Ramos (2001), El pass through del tipo de cambio: Un análisis para la economía costarricense de 1991 al 2001, Documento de Investigación, No. DIE-DM-11-2001-DI, Banco Central de Costa Rica.
- León, Jorge, Bernal Laverde, and Rodolfo Durán (2002), *El pass through del tipo de cambio en los precios de bienes transables y no transables en Costa Rica*, Documento de Investigación, No. DIE-05-2002-DI, Banco Central de Costa Rica.
- Mendoza, Omar (ed.) (2012), El efecto transferencia (pass-through) del tipo de cambio sobre los precios de Latinoamérica, Banco Central de Venezuela, Caracas.
- Mork, Knut A. (1989), "Oil and the Macroeconomy When Prices Go Up and Down: An Extension of Hamilton's Results," *Journal of Political Economy*, Vol. 97, No. 3.
- Orane, Angelo (2016), Estimación del traspaso del tipo de cambio hacia distintos componentes el índice de precios al consumidor, Documento de Investigación, No. DI-06-2016, Banco Central de Costa Rica.
- Pérez, Fernando J., and Marco Vega (2016), Asymmetric Exchange Rate Pass-through: Evidence from Nonlinear SVARs, Working Paper, No. 63, Asociación Económica Peruana.
- Potter M., Simon (2000), "Nonlinear Impulse Response Functions," *Journal of Economic Dynamics and Control*, Vol. 24, No. 10, 1425-1446.
- Przystupa, J., and E. Wróbel (2011), Asymmetry of the Exchange Rate Pass-through: An Exercise on the Polish Data, MPRA Paper, No. 17660, Munich Personal RePEC Archive, pp. 1-29.
- Rigobón, Roberto, and Thomas M. Stocker (2009), "Bias from Censored Regressors," *Journal of Business & Economic Statistics*, Vol. 27, No. 3, pp. 340-353.

- Rodríguez, Adolfo (2009), Evaluación del modelo lineal de pass-through para la proyección de inflación dentro del régimen de banda cambiaria, Documento de Investigación, No. DEC-DIE-DI-07-2009, Banco Central de Costa Rica.
- Segura, Carlos, and José Pablo Vásquez (2011), Estimación del parámetro de suavizamiento del filtro de Hoddrick y Precott para Costa Rica, Documento de Trabajo, No. DEC-DIE-DT 006-2011, Banco Central de Costa Rica.
- Taylor, John B. (2000), "Low Inflation, Pass Through, and the Pricing Power of Firms," *European Economic Review*, Vol. 44, No. 7, pp. 1389-1408.
- Wang, Sheng, and Rufei Guo (2016), "Asymmetric Exchange Rate Pass-through and Monetary Policy in Open Economy," *Annals* of Economics and Finance, Vol. 17, No.1, pp. 33-53.
- Winkelried, Diego (2003), "¿Es asimétrico el pass-through en el Perú? Un análisis agregado," VIII Reunión de Red de Investigadores de Bancos Centrales del Continente Americano, CEMLA.

Corporate Firms' Financial Conditions and Investment in Latin America: Determinants and Measurement

Óscar Carvallo Valencia Jonathan Barboza Pineda Ignacio Garrón Vedia

Abstract

For our research, we used a large dataset of nonfinancial firms from ten Latin American countries to assess leverage determinants and their dynamics. The results seem to be consistent with elements of both the trade-off and pecking order views. Also, the regression results show the presence of significant adjustment costs. According to our results, a firm's leverage is significantly reduced in the face of rising interest rates, with feed-back effects. Furthermore, we observed that reducing tangible assets induces more volatility in the interest rates paid by firms in the future. Essentially, when we separate firms according to leverage level, it appears that these effects are stronger for the highly leveraged

O. Carvallo <ocarvallo@cemla.org>, Deputy Manager for Financial Research, CEMLA, corresponding author; J. Barboza Pineda <jbarboza@cemla.org>, Researcher, CEMLA; I. Garrón Vedia <igarron@ bcb.gob.bo>, Economist, Banco Central de Bolivia. This article's findings, interpretations, and conclusions are entirely those of the authors and do not necessarily reflect the views of the CEMLA, their central bank members, nor the Banco Central de Bolivia. enterprises. Dynamically, in the case of increasing rates, there seems to be more risk associated with higher leverage. Our results show that this effect is manifested in higher volatility of interest rates and reduced collateral levels, potential asset liquidation and rapid deleveraging. The segments most likely affected are medium size firms and large firms with high costs of liquidation and high sunk costs, especially in the service sector. Firms operating in markets with unique products would also suffer. Traditional market-based indexes of financial conditions could be complemented by corporate indicators underlying the role of collateral, cash flows, and risk. Based on these findings we propose and calculate an index of corporate financial conditions for the region.

Keywords: corporate finance; Latin American firms, pecking order, trade-off theory, financial distress.

JEL classification: G3, G30, G31.

1. INTRODUCTION

ver the last decade, the patterns of financial intermediation have dramatically changed in emerging economies. First, there has been a change in the characteristics of financial intermediation between bank and market, the base source of corporate funding. This process happened after the global financial crisis and in tandem with the increase in global liquidity, which was a result of nonorthodox monetary policy in advanced economies. Also, many emerging economies (EMEs) shifted their source of funding to corporate deposits (a less stable form of funding that tends to emerge from debt issuance), which translates into a close relation between nonfinancial firms' leverage and banks' funding.

There has been substantial growth in the number of total debt securities, in particular those of foreign ownership. In parallel, emerging economies have become more financially integrated with the rest of the world, especially regarding global corporate debt markets. While this is seemingly a welcoming phenomenon, some market observers and researcher have warned about potential pitfalls in the process of monetary policy reversal. Easier access to funding may have distorted corporate investment decisions. Also, currency mismatches might be exposed. Even if firms are naturally or financially hedged, they might be still exposed to changes in global financial conditions, directly by interest-rate shocks or indirectly by falling commodity prices (Hattori and Takáts, 2015).

For example, Fuertes and Serena (2014) examined after-crisis financial vulnerabilities for 2,773 debt-issuer nonfinancial firms in 36 EMEs, for 2000-2014. They do not find in general issuers financial ratios to have dramatically worsened. However, they did find particular segments, high leveraged, low profitability, low interest coverage ratio (ICR), and low liquidity firms, to be worrisome. Latin American trends do not differ from these global trends. As a consequence, their potential exposure to some risks, related to profitability, currency mismatches, rollovers, and global markets conditions, might have had risen.

In this article, we examine these issues by modelling the possible determinants of nonfinancial firms leverage ratios, by using a firm-level dataset of ten Latin American economies and then assessing the influence of firm-level indicators reflective of market financial conditions. Further research on these patterns showed how this model could contribute to informing the creation of a better calibrated, higher frequency financial condition indexes, comprised of both financial and nonfinancial information. After this step, we evaluated leverage determinants in a panel data frame and estimate in a more dynamic framework the effects of financial factors proxies on a firm's leverage, using a panel VAR methodology (Abrigo and Love, 2016; Love and Zicchino, 2006).

Overall, our results show that Latin American nonfinancial firm's leverage determinants are stable across countries, coherent with the standard theory and other cross-sectional studies on the topic. Our more-dynamic approach give us preliminary evidence on the existence of significant and robust interactions between the fundamental determinants of nonfinancial firms' capital structure and the firm-level proxy indicators of financial conditions. These new findings support the fact that nonfinancial firm's indicators yield useful information to construct better calibrated, high-frequency indexes of financial conditions.

To that end, we calculate a simple index of financial conditions in the corporate sector. We also extend our dynamic analysis by including investment as an endogenous variable in our dynamic panel model. Implicit in our exercise is to representing financial variables in terms of their contribution to creating real investment impulses. By controlling for fundamental factors in the investment equation, we use the coefficients for the financial variables as factor loadings in the construction of a financial condition index for nonfinancial firms.

The rest of the paper is structured as follows. In Section 2, we review the related literature and present our research hypotheses. Section 3 explains the methodological aspects of the empirical exercise. The data elaborations are presented in Section 4. The empirical results are contained in Section 5, for the financial panel VAR, and 6, for our investment panel model. Section 7 concludes the research study.

2. LITERATURE REVIEW

According to Jensen and Meckling (1976), firm behavior should be seen as a conundrum of conflicting objectives in equilibrium, with a nexus of complex contractual relations as the outcome. The principal-agent problems are of primary importance in those equilibria in the context of pervasive asymmetric information environments. The literature on modelling nonfinancial debt ratio determinants has been done according to two prevailing approaches: *trade-off* and *pecking order* hypotheses.

Under the static version of the *trade-off* hypothesis, the optimal leverage reflects a single period trade-off between the benefits of debt tax shields and the deadweight costs of financial distress caused by an excessive debt ratio (DeAngelo and Masulis, 1980; Bradley et al., 1984). Meanwhile, under the dynamics *trade-off* view, firms exhibit dynamic target adjustment behavior, with the presence of short-term costs of adjustment, as deviations from individual target levels of leverage are gradually removed over time (Flannery and Rangan, 2006; Lemmon et al., 2008; Frank and Goyal, 2007, Huang and Ritten, 2007).

On the other hand, under the *pecking order* hypothesis, the costs of issuing risky debt or equity overwhelm the forces that determine optimal leverage in the trade-off model. To minimize asymmetric information costs and other financing costs, firms establish a hierarchy over their sources of funding: financing investments first with internal funds (i.e., retained earnings), then with safe debt, followed by risky debt, and finally equity (Myers, 1984; Myers and Majluf, 1984). Table 1 summarizes the implications for several leverage determinant variables, regarding the two competing views.

A very important implication of the pecking order view is that firms would prefer internal rather than external sources of funding. Regarding external funding, firms would prefer debt financing over equity financing. In this regard, the variable "Internal financing deficit" (IFD) is quite relevant, as it indicates the firm's needs for external funding. Thus, the equilibrium corporate financing mix for any firm, at any point in time, would depend critically on where the firm is located in the hierarchy of funding. Thus, cross-sectional estimates would be unable to capture funding optimal patterns. Indeed, we find evidence suggesting that the internal financing deficit is a critical determinant of leverage for forms in the region. In a final section, we use these findings to propose and calculate an index of financial conditions for the corporate sector in the region.

	OMPETING HYPOTHESES	Pecking order hypothesis			Since firms may be constrained due to asymmetric information problems, and thus adopt a hierarchy in selecting sources of funding, using retained profits firstly, there is a negative relation between profitability and leverage.	Firms with high growth opportunities must undertake major investment projects, so their need for funding is greater. When internal funding is exhausted, firms prefer financing through debt rather than equity. Therefore, a positive relation is forecasted between growth opportunities and leverage (Shyam-Sunder and Myers, 1999; Ramalho and Silva, 2009).
Table 1 I EVERACE DETERMINANTS ACCORDING TO CO	LEVERAGE DETERMINANTS ACCORDING TO C	Trade off hypothesis	Since there is a debt tax shield, a positive relation is expected between corporate tax rates and leverage (Fama and French, 2002).	Considering this is a substitute for the debit tax shields, there is a negative relation between nondebt tax shields and leverage (DeAngelo and Masulis, 1980).	Given that highly profitable firms are likely more able to payback their debts, this justifies a positive relation between profitability and leverage (Fama and French, 2002). However, in a dynamic model, profitability could proxy growth opportunities, so in this context the relation is ambiguous.	Since firms with high expectations for growth opportunities tend to be likely more exposed to bankruptcy and agency costs, and therefore may not use higher amounts of debt, there is a negative relation between growth opportunities and leverage (Myers, 1977, 1984; Rajan and Zingales, 1995).
		Variable	Tax rate	Other nondebt tax shields	Profitability	Growth opportunities

Collateral value (tangible or intangible fixed assets)	Tangible assets can be used as collaterals in the case of a firm's bankruptcy, so the firms with tangible assets tend to have easier access to external finance. Therefore, there is a positive relation expected between tangibility and leverage (Titman and Wessels, 1988; Michaelas et al., 1999).	On the one hand, tangible/intangible assets can be used as collaterals, which can help diminish asymmetric information problems between managers and creditors (Michaelas et al., 1999; Sogorb-Mira, 2005). In contrast, Harris and Raviv (1991) argue that this contributes to make equity less costly. Therefore, the relation between tangibility and leverage is ambiguous.
Size	Larger firms tend to increase their leverage and to take advantage of debt tax shields (Smith and Stulz, 1985) which results in a lower likelihood of bankruptcy (Titman and Wessels, 1988).	On the one hand, since larger firms tend to accumulate retained earnings, making debt less necessary (López-García and Sogorb-Mira, 2008). On the other hand, larger firms tend to have fewer problems with asymmetric information, and can obtain external funding on more favorable terms (Myers, 1984). Therefore, the relation between size and leverage is ambiguous.
Age	Older firms with better reputations tend to have lower costs of external funding. Therefore, there is a positive relation between age and debt (Ramalho and Silva, 2009).	Older firms with better reputation tend to have more capacity to retain and accumulate earnings. Therefore, a negative relation between age and leverage is expected (La Rocca et al., 2011).
Mean reversion	There is an optimal leverage ratio, where tax shield benefits are comparable to financial distress costs. Whenever firms deviate from their optimal ratios, the existence of adjustment costs prevents firms from making a total adjustment to that ratio. Therefore, firms make short-term partial adjustments of leverage towards the optimal ratio (López-García and Sogorb-Mira, 2008).	

	LEVERAGE DETERMINANTS ACCORDING TO	COMPETING THEORIES
Variable	Trade off hypothesis	Pecking order hypothesis
Internal financing deficit		Firms with high internal financing deficit tend to rely more heavily on external debt to finance their investment projects. Therefore, there is a positive relation between internal financing deficit and leverage.
Volatility of earnings	For <i>reasonable</i> parameter values, Bradley et al. (1984) state that a firm's optimal leverage is a decreasing function of the volatility of its earnings.	When net cash flows are low, firms with more volatile net cash flows are likely to have lower dividend payouts and less leverage (Fama and French, 2002).
Uniqueness	Customers, workers, and suppliers of firms that produce unique or specialized products probably suffer from relatively high costs (imposed by the firm) in the event of a firm's liquidation. This attribute may be negatively related to the observed debt since it is correlated with non-debt tax shields and collaterals.	
Industry classification	Firms manufacturing machines and equipment should be financed relatively less, since they will find liquidation relatively costly.	
Sources: Titman an	d Wessels (1988), Frank and Goyal (2007, 2008), and Serrasqu	iro y Caetano (2012).

44

Table 1 (cont.)

3. METHODOLOGY

3.1 Breaking Down the Debt Ratio Model

As suggested by the literature, we use a dynamic partial adjustment model to capture the cost of adjustments and other leverage determinants. The introduction of a lagged dependent variable among the right-hand side variables creates an endogeneity problem since the lagged dependent variable might be correlated with the disturbance term. To solve this problem, Arellano and Bond (1991) developed a difference GMM estimator for the coefficients in the equation mentioned above, where the lagged levels of the regressors are the instruments for the first differential equation. Further, Arellano and Bover (1995) and Blundell and Bond (1998) suggest differentiating the instruments instead of the regressors in order to make them exogenous from fixed effects. This leads to the differences between the GMM and the system GMM estimator, which is a joint estimation of the equation in levels and in first differences. Hence, we use the two-step system GMM estimators, with Windmeijer (2005) corrected standard error.

3.2 Examining How the Debt Ratio Model Is Influenced by Financial Conditions

Further, and considering the results from the previous partial adjustment model, we examine how equilibrium leverage ratios are impacted by financial conditions in a more dynamic setting. For doing so, we implement a panel vector autoregression (panel VAR) methodology. This approach treats all variables as endogenous (VAR) and incorporates the unobserved individual heterogeneity in the panel. We present the results of the panel VAR estimations as well as well as the corresponding impulse-response functions.

Following closely the instrumental variables system-GMM methodology suggested by Love and Zicchino (2006) and

Abrigo and Love (2015), we estimate a first order panel VAR as follows:

$$Y_{it} = \alpha + \theta Y_{it-1} + f_i + d_{ct} + e_{it},$$

where Y_{it} and Y_{it-1} are (5×1) vectors of variables (profitability, tangibility, leverage, tax shield and a proxy of financial conditions), for firm *i*, at a time *t* and *t*-1, respectively; θ is a (5×5) matrix of coefficients which are homogeneous for all firms; f_i denotes for firms' fixed effects and d_{ct} are country effects which are homogeneous for each firm in country *c* at time *t*. Finally, e_{it} is the vector of the respective white-noise disturbances.

Eliminating firms fixed effect f_i by differencing will create correlation with the lagged dependant variables, generating bias in the estimators. Also, the specification include country effects d_{ct} to account for country-specific macro shocks that affect all firms in country c at the same time, wich also would create estimators' bias. Thus, following Love and Zicchino (2006), we perform a two-way standardization of the variables used in the panel VAR, in order to eliminate f_i and d_{cl} effects. First, with regard to the country effects, we subtract the means of each variable for every country and year. Secondly, regarding the endogeneity of fixed effects and lagged dependent variables, we use the Helmert procedure for each variable by forward mean-differencing (Arellano and Bover, 1995). This method removes from the regressors the mean of all available future observations, thus preserving orthogonality between the resulting transformed variables and lagged regressors.

Also, following Abrigo and Love (2015), we also perform a Granger-causality Wald test for each equation of the panel VAR, to check for the empirical order of the VAR. As in a standard VAR model, we check for the presence of eigenvalues outside the unitary circle, thus assessing the stability of the panel VAR system. Also, we calculate and show Cholesky impulse-response functions and forecast-error variance decompositions. Then,

we use the evidence from the Granger-Wald causality tests to inform the ordering of the variables in the Cholesky decompositions. The confidence intervals for the impulse-response exercises are generated by Monte-Carlo random generation of $\hat{\theta}$ and its corresponding estimated variance-covariance matrix. We present 90% confidence intervals, with 1,000 repetitions. Lastly, for the construction of a financial condition index in the final section of the paper, we extend our initial dynamic panel model, in order to incorporate investment dynamics and the role of financial conditions.¹

4. DATA

The data we used in this study was Orbis BvD corporate dataset for ten Latin American countries: Argentina, Bolivia, Brazil, Chile, Colombia, Ecuador, Mexico, Peru, Uruguay, and Venezuela. After checking the data for extreme outliers and inconsistencies, we obtained leverage information for 10,005 firms in 17 economic sectors, in the period 2006-2015. Next, we aggregated those sectors in manufacturing, services, primary sector, utilities, and public sector.² We counted, on average, approximately 2.03 years of observations of each of the 10,005 firms (20,315 observations). Figure 1 shows leverage distributions for the 17 sectors represented in our sample. Notably, and as reflected in our results, sectoral patterns are a clear determinant of leverage. For the panel VAR exercise, a data subset is used, comprised of 1,939 firms with information with an average period of 5.92 years. Depending on the variables used in the regression, N could be reduced. Tables A.1 and A.2 in Annex A, show descriptive statistics for the samples.

¹ All calculations were perfomed using the following Stata's user-written comands *pvar*, *pvarsoc*, *pvargranger*, *pvarirf* and *pvfevd*, developed by Abrigo and Love (2015).

² We show table results only for the manufacturing, services, and primary sectors, the bulk of our sample.



Monetaria, January-June, 2018



5. RESULTS

Tables B.1 and B.2³ show Blundell-Bond system-generalized moments method (GMM) estimation results for the determinants of leverage in manufacturing, services, and the primary sectors.⁴ Similar to other corporate finance studies the results obtained seem to be consistent with the elements of the two main theories (Rajan and Zingales, 1995). Regression results show the presence of significant adjustment costs. To the extent that firms have unobservable target levels, firms face low speed of adjustment. This would be consistent with the trade-off dynamic theories. Additional evidence supporting the tradeoff theory is provided by variable's tax shield results, which is positively correlated with leverage.

For manufacturing and service firms, the ratio of tangible assets to total assets is negatively correlated with leverage. Also, tangible assets are found to be correlated with growth opportunities.

On the other hand, these assets can be used as collateral. Thus, this piece of evidence seems to be supportive of both the *trade-off* and the *pecking order* hypotheses. Furthermore, our results suggest that medium-sized firms in manufacturing and services tend to be significantly more leveraged than the small firms in these sectors, while very large and large companies in the services sector are significantly more leveraged than their counterparts in the medium-sized-firms group (see Annex A.3 for variables description). This is in line with the trade-off hypothesis, as well as with Myers (1984). Regarding the uniqueness indicator,⁵ we found that it affects leverage positively and

³ Henceforth, all the statistical tables not displayed in the body of this document can be located in the Annex B.

⁴ In Table B.2 we use ROAA as measure of cash flow effects, instead of our IFD variable.

⁵ Uniqueness, measured as costs of goods sold to operating revenue, is related to the extent to which the market for a good depends on retaining a significant customer base. To that regard,

significantly only for firms in the primary and service sectors, which is contrary to the trade-off hypothesis. Uniqueness, as pointed out by Gilchrist et al. (2016), is critical to understanding a firm's ability to increase prices; thus, it is connected to the financial distress during episodes of aggregated shocks. Firms that produce unique products are more vulnerable to interest rate shocks while being highly leveraged, since they tend to have less flexibility to increase prices.

Three variables' estimates are quite consistent with the pecking order hypothesis, namely the internal financing deficit, the dichotomic variable equal to one if the firm is listed (and zero otherwise), and the pofitability variable (return on average assets, or ROAA). Leverage is higher for firms with a larger financing deficit. On the one hand, listed firms or firms with higher profitability tend to have lower leverage ratios, likewise for smaller firms, so they are also consistent with this hypotheses.

In order to examine the possibility of multiple endogeneity of the regressors, our empirical strategy also includes estimating panel VARs and impulse-response functions for the subsample of firms with larger time series dimension.

In this regard, we reproduce previous specifications as much as possible, considering panel VAR stability conditions.⁶ Then we augment the regressions in two variables to show the effects of financial conditions at the individual firm's level. In one case, we include the previous year's implicit interest rate paid on liabilities. In the other, we calculated the rate's previous 3-year rolling window standard deviation. Figures 2a and 2b show the evolution of median and inter-quartile ranges for the implicit interest rates and its standard deviation for the ten countries examined. Most countries have experienced episodes of high

firms deploy marketing and sale forces resources to convey the special and unique nature of their product. In that regard, the customer base becomes a valuable asset in this kind of markets, with price competition playing a secondary role.

⁶ We use instead tangibility in this set of results, calculated as the ratio of fixed to total assets.





Figure 2b

AVERAGE AND INTER-QUARTILE VOLATILITY 3-Year Rolling SD of Implicit Interest Rate



interest rate volatility and level, especially in the immediate post global financial crisis.

Table 2 shows the panel VAR results for the interest rate variable. In Table 3 and Figure 4, the corresponding variance decomposition and impulse-responses are displayed. Our results suggest the presence of a bidirectional causality between interest rates and leverage. Previous year interest rates reduce leverage at a time t, whereas a rise in the previous year's leverage reduces the future rate charged on a firm's liabilities. The impulse-response functions (figure 4) shows that a shock increasing *Leverage* tend to have negative and significant effects over the future interest rate lasting about four years, while a shock increasing the implicit interest rate has negative and significant effects over *Leverage* lasting about five years.

When including the volatility (standard deviation) of the implicit interest rate as an endogenous component of the panel var (Table 4), we find that firms with larger collateral (tangible assets) face lower future interest rate volatility. Also, under this specification, higher previous leverage seems to be associated with higher future profitability (ROAA). As shown by impulse-response functions (Figure 5), a shock increasing leverage has an immediate negative effect on profitability, compensated onwards with a significant increase in the second year wich lasts for about the fifth year.

Overall, our results seem to indicate that leverage is affected by previous interest rates, an obvious result, but with feedback effects involved. Conversely, collateral values seem to be important determinants of the future interest rate volatility facing firms. As shown by variance-decomposition results (Table 3), around 10% of the implicit interest variance is explained by leverage. Also, the tangibility of assets explains about 45% of volatility-of-interest-rate variance. The impulse responses for the effect of previous interest rates on leverage last for at least five years. Of similar duration is the reverse causality effect. Also, the effect of the tangibility of the future volatility of interest rates lasts for five years Figure 5). Table 2

	Response to						
Response of	$\begin{array}{c} ROAA\\ (t-1) \end{array}$	Leverage (t-1)	Tangibility (t-1)	Imp. int. rate (t-1)	Tax shield (t-1)		
ROA (t)	0.3744°	0.0609	-0.0417	0.0004	0.2178		
	(0.0686)	(0.0379)	(0.0346)	(0.0331)	(0.1412)		
Leverage (t)	-0.1891	0.8051°	-0.0135	-0.0857^{a}	0.1794		
	(0.0793)	(0.0644)	(0.0607)	(0.0459)	(0.2139)		
Tangibility (t)	-0.1252	-0.0660	0.8286°	-0.0910	0.0068		
	(0.0777)	(0.0769)	(0.0837)	(0.0587)	(0.2075)		
Imp. int. rate (t)	0.0291	-0.1209°	-0.0005	0.2944^{b}	-0.0557		
	(0.0432)	(0.0378)	(0.0311)	(0.1157)	(0.0916)		
Tax shield (t)	0.0601^{b}	-0.0042	0.0126	-0.0034	0.3312 ^c		
	(0.0240)	(0.0156)	(0.0142)	(0.0103)	(0.0738)		

PANEL VECTOR AUTOREGRESSIOM FOR DETERMINANTS OF CORPORATE FINANCING AND THE PREVIOUS IMPLICIT INTEREST RATE

Number of observations (N): 2,400

Number of firms (N): 669

Average number of years: 3.587

Final GMM criterion Q(b): 7.52e-34

Initial weight matrix: identity

GMM weight matrix: robust

^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Standard errors in parenthesis. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. This panel VAR satisfies the stability condition proposed by Hamilton (1994) and Lütkepohl (2005).

Table 3

VARIANCE DECOMPOSITION OF THE PANEL VAR FOR DETERMINANTS OF CORPORATE FINANCING AND THE PREVIOUS IMPLICIT INTEREST RATE

	Impulse variable						
Response variable	ROAA	Tangibility	Tax shield	Imp. int. rate	Leverage		
ROAA	0.8911	0.0508	0.0086	0.0017	0.0477		
Tangibility	0.0271	0.9515	0.0001	0.0012	0.0201		
Tax shield	0.2379	0.0160	0.7457	0.0002	0.0003		
Imp. int. rate	0.0213	0.0023	0.0006	0.8721	0.1036		
Leverage	0.1293	0.0239	0.0010	0.0427	0.8030		

Percent of variation in the row variable (10 years ahead) explained by the column variable. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. The variables were sorted following Granger-Wald causality test criteria.

IMPULSE-RESPONSES OF THE PANEL VECTOR AUTOREGRESSION FOR DETERMINANTS OF CORPORATE FINANCING AND THE PREVIOUS IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.



IMPULSE-RESPONSES OF THE PANEL VECTOR AUTOREGRESSION FOR DETERMINANTS OF CORPORATE FINANCING AND THE PREVIOUS IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

Figure 4 (cont.)

IMPULSE-RESPONSES OF THE PANEL VECTOR AUTOREGRESSION FOR DETERMINANTS OF CORPORATE FINANCING AND THE PREVIOUS IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

Table 4

			Response to		
Response of	ROAA $(t-1)$	Leverage (t-1)	Tangibility (t-1)	sD imp. int. rate (t-1)	Tax shield (t-1)
ROAA (t)	0.3420°	0.1058 ^b	-0.0330	-0.0006	-0.0398
	(0.0790)	(0.0457)	(0.0472)	(0.0675)	(0.2213)
Leverage (t)	-0.1181	0.7694°	-0.0470	-0.0662	-0.0626
	(0.1049)	(0.0775)	(0.0782)	(0.0906)	(0.3259)
Tangibility (t)	-0.1359	-0.0889	0.8497°	-0.0315	0.0724
	(0.1172)	(0.0890)	(0.1093)	(0.0938)	(0.3358)
SD. imp. int. rate (t)	0.0105	-0.0120	-0.0586°	0.8131°	0.0742
	(0.0293)	(0.0208)	(0.0224)	(0.0940)	(0.0680)
Tax shield (t)	0.0334	-0.0071	-0.0011	-0.0104	0.3727°
	(0.0321)	(0.0194)	(0.0214)	(0.0176)	(0.1208)

PANEL VECTOR AUTOREGRESSION FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF THE IMPLICIT INTEREST RATE

Number of observations (N): 1,745

Number of firms (N): 537

Average number of years: 3.25

Final GMM criterion Q(b): 4.24e-34

Initial weight matrix: identity

GMM weight matrix: robust

^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Standard errors in parenthesis. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. This panel VAR satisfies the stability condition proposed by Hamilton (1994) and Lütkepohl (2005).

Table 5

VARIANCE DECOMPOSITION OF THE PANEL VAR FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF THE IMPLICIT INTEREST RATE

	Impulse variable					
Response variable	Tax shield	Leverage	Tangibility	ROAA	sD imp. int. rate	
Tax shield	0.9912	0.0017	0.0005	0.0059	0.0008	
Leverage	0.0240	0.9523	0.0156	0.0045	0.0036	
Tangibility	0.0013	0.1063	0.8895	0.0027	0.0002	
ROA	0.1607	0.1610	0.0737	0.6042	0.0004	
SD imp. int. rate	0.0063	0.0190	0.4584	0.0020	0.5143	

Percent of variation in the row variable (10 years ahead) explained by the column variable. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. The variables were sorted following Granger-Wald causality test criteria.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF THE IMPLICIT INTEREST RATE



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.



IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF THE IMPLICIT INTEREST RATE

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

Figure 5 (cont.)

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF THE IMPLICIT INTEREST RATE



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

We further look at threshold effects in the leverage distribution by dividing firms into above and below median leverage. Results are shown in Tables B.3-B.4 and Figures B.1-B.2,⁷ for previous implicit interest rates; and Tables and Figures B.5-B.6 and Figures B.3-B.4 show results for the volatility of implicit interest rates.

The first part of Table B.3 reports the panel VAR estimates for firms whose mean leverage ratio is lower than the median and where the bidirectional causal relation between leverage and implicit interest rate found in the baseline model is reproduced. Interchangeably, results for the highly leveraged group of firms are presented in the bottom part of Table B.3. As the opposite of low-leveraged firms, in the case of high-leveraged the feed-back effect between implicit interest rates is no longer held, since only the one-year lagged leverage impacts the implicit interest rate significantly and negatively. Also important is the fact that, for this group of firms, implicit interest rates are negatively associated with the future collateral measured by tangibility, which means that an increases in previous rates reduces significantly the tangible assets of the firm in the next five years (with regards to impulse-response functions presented in Figure B.2). We presume this result is driven by the fact that already highly leveraged firms tend to face relevant price effects in their balance sheets when interest rates increase, and additionally, they are induced to liquidate asset positions in the face of interest rate shocks.

Regarding the impulse-response functions for highly leveraged firms (Figure B.2), then the future profitability grows up significantly from the second year after the leverage increases, up to about to the fifth year (Figure B.2). In turn, a positive shock of the implicit interest rate at year t leads to a significant decrease of the future collateral values, while the collateral itself is found to cause an increase in the future volatility of

⁷ Henceforth, all figures not displayed in the body of this document can be located in the Annex B.

rate (as shown at the bottom of Table B.5, for highly leveraged firms). This fact constitutes a negative spiral, in which financial conditions for firms are further deteriorated. The compensating mechanism to end up this harmful process seems to operate at lower leverage and profitability, as firms engage in a new leverage cycle. This is reflected in the negative and significant coefficient of the lagged profitability on future leverage.

6. AN AGGREGATED INDEX OF CORPORATE FINANCIAL CONDITIONS FOR TEN LATIN AMERICAN COUNTRIES

In this section, we extend our previous analysis by including investment as an endogenous variable in our dynamic panel model. Implicit in our exercise is representing of financial variables in terms of their contribution with the goal of creating real investment impulses. By controlling for fundamental factors in the investment equation, we use the coefficients for the financial variables, as factor loading in the construction of a corporate financial condition index.

We derive our intuition for our proposed index from the literature on micro-level real investment measuring. Investment dynamics within a PVAR firm-level have been estimated with the inclusion of financial variables (Love and Zicchino, 2006; Gilchrist and Himmelberg, 1998). Love and Zicchino (2006) estimated an investment PVAR using firm-level data from 36 countries. In their model, they included fundamental factors such as a measure of the marginal productivity of capital and Tobin's q. Their financial factors variable is represented by cash flows scaled by capital. Thus, their exercise is determining the dynamic function of investment that is augmented by a financial variable. They found the friction effect of the financial variable on investment to be larger for the group of countries with less developed financial systems. Also, a series of papers have looked at the elasticity of investment to cash flow and other financial variables, generally in a static or dynamic panel data context (Gomes, 2001; Balfoussia and Gibson, 2016; Hernando and Martínez-Carrascal, 2008) analyzed the impact of alternative measures of firms' financial conditions on investment and employment by using a large-scale panel dataset of Spanish firms over the period 1985-2001. They then used the estimated coefficients of the investment equation as factor loadings in the construction of a corporate financial conditions index. As Hernando and Martínez-Carrascal (2008), we estimate an error-correction investment model, as suggested by Bond et al. (1999). We follow this latest approach in the construction of our index of corporate financial conditions.

In this sense, we estimate a dynamic system-GMM panel model for fixed investment ratio at firm-level assuming the existence of additive year-specific effects, μ_t , country-specific effects, τ_k , and industry specific effects, γ_f , which could be expressed as follows:

$$\begin{split} \frac{I_{it}}{K_{i,t-1}} &= \rho_1 \Biggl(\frac{I_{i,t-1}}{K_{i,t-2}} \Biggr) + \omega_0 \Delta y_{it} + \omega_1 \Delta y_{i,t-1} + \theta(k-y)_{i,t-2} + \beta_0 Lev_{it} + \\ &+ \beta_1 Lev_{i,t-1} + \beta_2 IDB_{it} + \beta_3 IDB_{i,t-1} + \beta_4 (Zscore)_{it} + \beta_5 (Zscore)_{i,t-1} \\ &+ \beta_6 (IFD)_{it} + \beta_7 (IFD)_{i,t-1} + \beta_8 (Tangibility)_{i,t} + \\ &+ \beta_9 (Tangibility)_{i,t-1} + X_{i,t} \delta + \mu_t + \tau_k + \gamma_f + \varepsilon_{it}. \end{split}$$

We then construct indexes of financial conditions for our ten countries as follows. First, we estimate an error-correction investment model including lagged fixed investment, lagged and contemporaneous sales growth; the error-correction term, and other controls. Alternatively, we expand the investment model by including lagged and contemporaneous of several key financial variables from our previous analysis- Leverage, our internal financing deficit indicator (IFD), the interest debt burden, the tangibility of assets and the firms' Z-score, as a measure of risk. Results are shown in Table 7. A key aspect of the model is the inclusion of time and firm effects to capture capital replacement costs. Also, the model predicts the existence of significant and negative error correction component. We then use the results for the investment equation for the construction of our financial conditions index.

Notice that about the financial variables included, only the IFD, Z-score and tangibility coefficients were found to be significant. Consistent with previous results, we used specification 2 in Table B.7, as leverage was found in previous sections to be caused by both tangibility and IFD. Accordingly, in the specification 1, lagged leverage is found to significantly explain investment when excluding these two of its determinants. In specification 3, we use profitability (ROAA) instead of IFD, and get similar results. For the variables of interest, the contemporaneous effects are significantly positive, and the lagged effects are significantly negative. However, the sum of both coefficients is found to be significantly different from zero and positive for Z-score and the IFD, the variables with the largest effects, indicating a positive relation between the index loadings and investment financial conditions. Accordingly, we propose the following financial conditions index (FCI) for nonfinancial firms:

$$FCIit = \widehat{\beta_4} \left(Zscore \right)_{it} + \widehat{\beta_5} \left(Zscore \right)_{i,t-1} + \widehat{\beta_6} \left(IFD \right)_{it} + \widehat{\beta_7} \left(IFD \right)_{i,t-1} + \widehat{\beta_8} \left(Tangibility \right)_{it} + \widehat{\beta_9} \left(Tangibility \right)_{i,t-1}.$$

FCI can be interpreted as the predicted *financial* value of the investment. In order to have a country index, we aggregate at country-time level by calculating percentile 25, 50 and 75 statistics for the index. Figures B.5 and B.6 show the resulting lags of country-time FCI pair as compared to gross fixed capital formation and GDP growth.

The index is constructed so that increasing/decreasing values imply improving/deteriorating financial conditions for investment. The figures convey, at the simple examination, the potential for a positive correlation. We further explore these patterns as follows. First, we estimate a simple first order panel VAR model including FCI, gross fixed capital formation, and
GDP growth, for the ten countries in the sample. As an initial step, and test for Granger causality. The results are shown in Table B.8.

Granger causality Wald tests indicate IFC to Granger cause both gross fixed capital formation and GDP growth. The reverse causality is not found. Also, GDP growth Granger causes gross fixed capital formation. At a final exercise, we show in Figure 9, resulting impulse-response functions assuming a Cholesky variance decomposition with ordering given by the obtained Granger criteria. A one-standard deviation positive shock in FCI results in an increase in both gross fixed capital formation and GDP future growth, which is significant and lasting for about 12 months, with 90% confidence levels. Thus, these preliminary results, albeit restricted about its simplicity and extent of the series, provides some evidence on the potential explanatory relevance of the financial conditions index constructed thus far, using firm level data. It is also worth to notice that the financial shock implicit in the exercise is common across countries, given the nature of the exercise. Thus, the real impulses obtained must be interpreted accordingly, as the average national real effects to a common financial shock.

7. CONCLUSIONS

In this article, we use a large dataset of nonfinancial firms from ten Latin American countries to assess leverage determinants and their dynamics. We then use that information to inform the specification of a new index of corporate financial conditions.

With regard to the first set of issues, our results seem to be consistent with elements of the two main theories, the tradeoff, and pecking order views. Regression results show the presence of significant adjustment costs. To the extent that firms have unobservable target levels, firms face low speed of adjustment. Furthermore, our results suggest that medium-sized firms in manufacturing and services tend to be significantly more leveraged than their small firms in these sectors, whereas very large and large companies in the services sector are significantly more leveraged than their counterparts in the medium-sized-firms group. Regarding the uniqueness indicator, we found that it affects leverage levels positively and significantly, only for firms in the primary and service sectors, which is evidence against the trade-off hypothesis. With regard to our dynamics determinants of leverage, we observe that a firm's leverage is significantly reduced in the face of rising interest rates, with feed-back effects. Also, firms' collateral resulted to be critical, as reductions in tangible assets bring about future volatility in the interest rates paid by the firms. When we separate firms according to the leverage level, it turns out that these effects are stronger for highly leveraged firms.

Dynamically, the risk seems to be associated with high leverage in the context of rate increases. It is manifested in higher rate volatility and reduced collateral levels, potential asset liquidation and rapid deleveraging. These dynamics are probably more likely in the context of policy rate changes and capital outflows. According to our results, segments most likely affected are medium size firms and large firms with high costs of liquidation as well as high sunk costs, especially in the service sector. Firms operating in markets with unique products would also suffer.

Our results ultimately suggest that traditional market-based indices of financial conditions could be complemented by corporate indicators. As mentioned, collateral levels, indicators of corporate distress and firms' cash flow positions are natural candidates for an index. To that end, we calculated a simple index of financial conditions in the corporate sector (FCI). Granger causality Wald tests indicate ICFC to Granger-cause both gross fixed capital formation and GDP growth. According to resulting impulse-response functions, a one-standard deviation positive shock in IFC results in an increase in both gross fixed capital formation and GDP future growths, which is significant and lasts for about 12 months. Thus, these preliminary evidence suggests the potential predictive relevance of the index proposed here.

	DESCRIPTIVE STATISTICS FOR THE	INITIAL LEV	ERAGE DETE	RMINAN	TS EXER	CISE		
Variable	Dasowiktion	Tube	Observations	Panels	Moon	Std.	Min	Mav
Auntune	nescription	adkr	CU0361 001101	(cmin)	unati	uev.	11111	YDIAI
Leverage	Total liabilities to total assets (%)	Firm-level	20,315	10,005	48.059	25.012	0.000	100.000
Intangibility	Intangible assets to total assets ($\%$)	Firm-level	17,457	8,969	5.182	13.580	0.000	99.948
Tangibility	Tangible assets to total assets $(\%)$	Firm-level	20,315	10,005	32.834	28.339	0.000	100.000
Internal financial déficit (IFD)	Operative expenditures (investment+working capital) minus the cash flow (which is assumed to be equal to 110% of the net income) to total assets (%)	Firm-level	20,315	10,005	-2.968	23.299	-99.885	99.819
Uniqueness	Costs of goods sold to operating revenue	Firm-level	13,359	3,960	65.389	22.346	0	199.633
Listed	(1): if the firm is publicly listed, (0): otherwise (source: Orbis and own calculations)	Firm-level	20,315	10,005	0.213	0.409	0	1
ROAA	Net income to average of total assets (%)	Firm-level	20,315	10,005	5.115	11.643	-96.000	98.425
Tax shield	Total taxes to total assets ($\%$)	Firm-level	16,643	8,744	3.474	5.324	0.000	99.579
Source: Orbis and	d own calculations.							

Table A.1

AnnexA

Table A.2

DESCRIPTIVE STATISTICS FOR THE PANEL VAR LEVERAGE DETERMINANTS EXERCISE

Max	100.000	99.080	85.797	94.073	115.096	78.814
Min	0.005	0.000	-92.576	0.000	0.013	0.001
Std. dev.	22.739	24.548	8.326	5.370	5.363	3.473
Mean	49.994	29.097	4.637	4.243	5.639	2.003
Panels (firms)	1,939	1,939	1,939	1,851	849	832
Observations	11,487	11,487	11,487	9,206	5,183	4,122
Type	Firm-level	Firm-level	Firm-level	Firm-level	Firm-level, proxy of financial conditions	Firm-level, proxy of financial conditions
Description	Total liabilities to total assets (%)	Tangible assets to total assets $(\%)$	Net income to average of total assets (%)	Total taxes to total assets ($\%)$	Interest paid to total liabilities (%)	3-year rolling standard devation of the implicit interest rate (with the exception of year 2008, in which we imputed a 2-year rolling standard deviation)
Variable	Leverage	Tangibility	ROAA	Tax shield	Imp. interest rate	SD of imp. interest rate

Source: Orbis and own calculations.

	FIRM'S CLASSIFI	CATION			
		A rellano-	Bond GMM	Panel vAR (only data for imp	v 849 firms with . interest rate)
Variable	Description	Observations	Panels (firms)	Observations	panels (firms)
By size (Orbis criteria):					
Very large and large	Firms that match at least one of the following conditions: operating revenue ≥ USD 13 M; total assets ≥ USD 26 M; employees ≥ 150.	11,849	5,747	5,131	831
Medium	Firms that match at least one of the following conditions: operating revenue ≥ USD 1.3 M; total assets ≥ USD 2.6 M; employees ≥ 15; not very large or large.	6,076	1,288	48	17
Small By industrial qualification	Firms that are not included in any of the above categories. (Bureau Van-Dijk main sector criteria):	2,390	2,970	4	1
Manufacturing	Firms classified in any of the following sectors: "Chemicals, rubber, plastics, non-metallic products," "Food, beverages, tobacco," "Machinery, equipment, furniture, recycling," "Metals and metal products," "Publishing, printing," "Textiles, wearing apparel, leather," "Wood, cork, paper."	6,110	2,568	2,170	336

Table A.3

Table A.3 (cont.) FIRM'S CLASSIFICATION

y 849 firms with . interest rate)	panels (firms)	302	62	6	147
Panel VAR (only data for imp	Observations	1,701	367	11	934
Bond $_{GMM}$	Panels (firms)	6,290	233	69	845
Arellano-	Observations	11,368	934	170	1,733
	Description	Firms classified in any of the following sectors: "Construction," "Education, health," "Hotels and restaurants," "Post and telecommunications," "Transport," "Wholesale and retail trade."	Firms classified in the following sector: "Primary sector."	Firms classified in the following sector: "Public administration and utilities."	Firms classified in the following sector: "Gas, water, electricity."
	Variable	Services	Primary	Public administration	Utilities

Annex B

Table B.1

	BASELI	INE GMM (BLUND)	ELL-BOND) REGRE	SSIONS FOR LEVER	AGE	
	Manufacturing	Services	Primary sector	Manufacturing	Services	Primary sector
Leverage (-1)	0.667°	0.644°	0.790°	0.664°	0.650°	0.761°
	(0.0417)	(0.0453)	(0.0924)	(0.0423)	(0.0437)	(0660.0)
Tangible assets	$0.0375^{ m b}$	-0.0481°	0.0389	$0.0377^{ m b}$	-0.0436°	0.0442
	(0.0164)	(0.0159)	(0.0379)	(0.0162)	(0.0159)	(0.0342)
IFD	0.0661°	0.0981°	0.0516^{a}	$0.0633^{ m c}$	0.0975°	0.0575^{a}
	(0.0124)	(0.0123)	(0.0311)	(0.0126)	(0.0124)	(0.0321)
Listed	$-2.061^{\rm b}$	$-2.157^{ m b}$	-3.435	$-2.180^{\rm b}$	-2.432°	-3.875^{a}
	(0.882)	(0.888)	(2.262)	(0.908)	(0.933)	(2.027)
Tax shield	0.0524	0.0577	0.0768	0.107^{a}	0.0903	0.183
	(0.0482)	(0.0662)	(0660.0)	(0.0568)	(0.0654)	(0.117)
Small	-1.499	-3.781°	-2.412	-1.012	-3.590°	-3.088
	(1.454)	(1.151)	(2.141)	(1.423)	(1.122)	(1.960)
Very large	0.931	2.230°	0.279	0.864	$1.820^{ m b}$	-0.453
	(0.974)	(0.830)	(2.075)	(0.944)	(0.815)	(2.014)
Uniqueness				0.0703°	0.0438°	0.114°
				(0.0170)	(0.0152)	(0.0403)
Constant	18.81°	24.66°	12.31^{a}	14.62°	22.46°	8.409
	(2.769)	(3.631)	(6.907)	(2.695)	(3.524)	(6.120)

Table B.1 (cont.)

BASELINE GMM (BLUNDELL-BOND) REGRESSIONS FOR LEVERAGE

	Manufacturing	Services	Primary sector	Manufacturing	Services	Primary sector
Country	Yes	Yes	Yes	Yes	Yes	Yes
Year	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
Z	3697	3774	606	3622	3562	601
Firms	835	1127	167	664	1053	165
J	74	68	65	75	69	66
Hansen	57.42	42.34	37.78	57.46	38.96	38.12
Hansen-p	0.0696	0.289	0.571	0.0691	0.426	0.555
AR1	-8.117	-7.124	-3.711	-8.123	-7.059	-3.552
AR1-p	4.79e-16	1.05e-12	0.000206	4.55e-16	$1.68e{-}12$	0.000382
AR2	1.056	-0.452	1.940	1.134	-0.386	2.004
ar2-p	0.291	0.651	0.0524	0.257	0.700	0.0451

represents the lagged value of the firm's leverage ratio (%); Listed is dummy variable for firms that participate in the stock market; Tangible and primary industry conglomerates. The Hansen is a test of the over-identifying restrictions for the GMM estimators. AR1 and AR2 are tests for the first-order and second-order serial correlation. N denotes the number of observations and I number of instruments. Country, Year Standard errors in parentheses ^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Columns 1, 2 and 3 represent the regressions for manufacturing, services, assets is the firm's tangible fixed assets (%); JFD is the firm's internal financing deficit to total assets (%); Tax shield is the and Industry denote if their respective dummy variables were introduced in the regressions. Variables are listed as follows: Lending(-1) firm's taxes to total assets(%); Small and Very large are dummies for small and very large firms according to Orbis disaggregation; and Uniqueness is the firm's cost of goods sold to operating revenue(%)

BASELINE REGRESSIONS FOR LEVERAGE

Primary sector (0.0360)(0.0542) 0.0692^{a} -0.201° -5.740^{b} (2.379)(0.0245)0.0721)(0.0813) 0.674° 0.174^{b} -3.277(2.311)-0.228(2.825) 0.0847^{c} (5.887) 15.59° (0.0411) -0.0404° (0.0146)(0.0393)(0.0490)(0.0128) -0.366° -5.445° 0.0277^{b} Services (1.123)0.0269 -4.150° (1.020)(0.968)(3.915) 3.072° 37.47° 0.490° Manufacturing(0.0376)(0.0494) 0.0416^{b} (0.0168) -0.310° (0.0183)(0.0414) -3.406° 0.0980^{b} -3.018^{b} (1.000)(1.305)0.0209(1.088) 0.0315^{a} 25.17° (3.185) 0.545° Primary sector (0.0349)(0.0671) -0.206° -7.186° (0.0720) 0.0666^{a} -4.889^{b} (2.479)0.0270(0.0524) 0.633° (2.294)(6.756)(3.183)0.525 24.23° (0.0319)(0.0377)(0.0145)-0.0153(0.0373) -0.328° -4.915° -5.705° Services (1.083)0.0403(1.007)(1.029)(3.981) 4.107° 38.56° 0.445° Manufacturing(0.0186)(0.0436) 0.0638° -0.365° 0.0846^{a} (0.0469)(0.0381) -3.340° -0.0293 -4.134° 0.481° (1.155)(1.340)(1.262)(3.806)29.23° Tangible assets Leverage(-1) Uniqueness Tax shield Very large Constant Listed Small ROAA

Table B.2 (cont.)

BASELINE REGRESSIONS FOR LEVERAGE

	Manufacturing	Services	Primary sector	Manufacturing	Services	Primary sector
Country	Yes	Yes	Yes	Yes	Yes	Yes
Year	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
Z	4,851	6,765	923	4,396	5,434	835
Firms	1,134	1,964	232	1,004	1,614	212
J	74	72	66	75	72	67
Hansen	60.94	57.24	32.86	64.62	42.21	30.72
Hansen-p	0.0370	0.0586	0.814	0.0181	0.418	0.879
AR1	-7.604	-9.476	-4.180	-8.370	-8.285	-4.278
AR1-p	2.88e-14	2.63e-21	0.0000292	5.74e-17	1.18e-16	0.0000189
AR2	0.863	0.824	0.332	1.300	0.335	1.279
ar2-p	0.388	0.410	0.740	0.194	0.738	0.201

(%); Small and Very large are dummies for small and very large firms according to Orbis disaggregation; and Uniqueness is the firm's cost of and primary industry conglomerates. The Hansen is a test of the over-identifying restrictions for the GMM estimators. AR1 and AR2 are tests represents the lagged value of the firm's leverage ratio (%); Listed is dummy variable for firms that participate in the stock market; ROA is firm's return on assets (%); Tangible assets is the firm's tangible fixed assets to total assets (%); Tax shield is the firm's taxes to total assets for the first-order and second-order serial correlation. N denotes the number of observations and I number of instruments. Country, Year Standard errors in parentheses ^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Columns 1, 2 and 3 represent the regressions for manufacturing, services, and Industry denote if their respective dummy variables were introduced in the regressions. Variables are listed as follows: Leverage (-1) goods sold to operating revenue (%).

PANEL VECTO AND THE IMPLICI	R AUTOREGRESSI IT INTEREST RATF LEVERA	ON (GMM ESTIMATIO E AS A PROXY OF FINA GE RATIO (TOTAL LL	N) FOR DETERMINA INCIAL CONDITIONS ABILITIES TO TOTAL	NTS OF CORPORATE I 5-BY GROUPS OF FIRM ASSETS)	FINANCING AS REGARDING
I			Response to		
Response of	ROAA $(t-I)$	Leverage $(t-1)$	Tangibility $(t-1)$	Imp. int. rate (t-1)	Tax shield $(t-1)$
	a) Firms w	ith a mean leverage ratic	lower than the median	(<51.02%)	
ROAA (t)	$0.3944^{ m c}$	-0.0149	-0.0201	0.0199	0.2484
	(0.0974)	(0.0650)	(0.0526)	(0.0326)	(0.1946)
Leverage (t)	-0.0975	0.9575°	-0.1012	-0.1698°	0.2877
	(0.1111)	(0.1104)	(0.0994)	(0.0394)	(0.2705)
Tangibility (t)	-0.0656	-0.0237	0.8316°	-0.0288	-0.1909
	(0.1021)	(0.1531)	(0.1202)	(0.0572)	(0.2430)
Imp. int. rate (t)	0.0353	-0.1505^{a}	-0.0575	$0.307^{ m b}$	-0.135
	(0.0831)	(0.0779)	(0.0588)	(0.1503)	(0.1705)
Tax shield (t)	0.0872°	-0.0323	0.0068	-0.0022	0.1519^{a}
	(0.0285)	(0.0242)	(0.0169)	(0.0115)	(0.0806)
Number of observatio	ns (N): 1,106				
Number of firms (N):	291				
Average number of ye	ars: 3.801				
Final GMM criterion C	2 (b): 4.45e–34				

		Table B	.3 (cont.) Response to		
- Response of	$ROAA \ (t-I)$	Leverage (t–1)	Tangibility (t-1)	Imp. int. rate (t-1)	$Tax \ shield \ (t-1)$
ROAA (t)	0.3704°	0.1111^{b}	-0.0573	-0.0228	0.1477
	(0.0991)	(0.0478)	(0.0457)	(0.0888)	(0.2037)
Leverage (t)	-0.2754^{b}	0.7171°	0.0527	0.1084	0.1586
	(0.1175)	(0.0813)	(0.0800)	(0.1474)	(0.3218)
Tangibility (t)	-0.1999^{a}	-0.1131	0.7985°	$-0.3252^{\rm b}$	0.1733
	(0.1125)	(0.0766)	(0.1139)	(0.1376)	(0.3237)
Imp. int. rate (t)	0.0166	-0.0926°	0.0513^{a}	0.2717°	0.0205
	(0.0401)	(0.0297)	(0.0312)	(0.0600)	(0.1033)
Tax shield (t)	0.0418	0.0129	0.0176	0.0034	0.4610°
	(0.0348)	(0.0209)	(0.0223)	(0.0272)	(0.1131)
Number of observatio	ns (N): 1,294				
Number of firms (N):	378				
Average number of ye	ars: 3.423				
Final GMM criterion (Q (b): 3.25e-34				
Initial weight matrix:	identity				
GMM weight matrix: r	obust				
^a $p < 0.10$, ^b $p < 0.05$, ^c $p <$ Arellano and Bover (199 calculated for each coun	0.01. Standard error 5), through the Heln try-year. This panel V	s in parenthesis. All vari nert procedure. All cour AR satisfies the stability	ables were transformed ttry effects were include condition proposed by J	using forward orthogon: d by subtracting the mea Hamilton (1994) and Lü	lization suggested by ns of each variable .kepohl (2005).

P-VAR VARIANCE DECOMPOSITION FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS- BY GROUPS OF FIRMS REGARDING LEVERAGE RATIO (TOTAL LIABILITIES TO TOTAL ASSETS)

a) Firms with a Mean Leverage Ratio Lower than the Median (<51.02%)

Response			Impulse variabl	le	
Variable	ROAA	Tangibility	Imp. int. rate	Leverage	Tax shield
ROAA	0.9498	0.0039	0.0073	0.0306	0.0084
Tangibility	0.0351	0.9611	0.0003	0.0016	0.0019
Imp. int. rate	0.0221	0.0302	0.7027	0.2429	0.0021
Leverage	0.0247	0.1939	0.0925	0.6845	0.0045
Tax shield	0.2340	0.0507	0.0215	0.1517	0.5421

b) Firms with a mean leverage ratio higher than the median or equal to the median (>51.02%)

Response			Impulse varia	ble	
Variable	Tax shield	Leverage	ROAA	Tangibility	Imp. int. rate
Tax shield	0.9533	0.0066	0.0066	0.0331	0.0004
Leverage	0.0247	0.7952	0.0703	0.1076	0.0022
ROAA	0.1584	0.1454	0.6583	0.0363	0.0016
Tangibility	0.0007	0.0365	0.0524	0.8926	0.0178
Imp. int. rate	0.0063	0.1539	0.0226	0.0438	0.7734

Percent of variation in the row variable (10 years ahead) explained by the column variable. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. The variables were sorted following Granger-Wald causality test criteria.

PANEL VECTOR AUTOREGRESSION (GMM ESTIMATION) FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD. OF IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS-BY GROUPS OF FIRMS REGARDING LEVERAGE RATIO (TOTAL LIABILITIES TO TOTAL ASSETS)

			Response to		
Response of a) Firms with a m	ROAA (t-1) ean leverage	Leverage (t-1)) e ratio lower	$\frac{Tangibility}{(t-1)}$ than the med	sD imp. int. rate (t-1) dian (<51.0	Tax shield (t-1)
ROAA (t)	0.4022°	0.0319	-0.0092	-0.0091	0.0779
	(0.1161)	(0.0619)	(0.0651)	(0.0844)	(0.3005)
Leverage (t)	-0.1334	0.9129°	-0.1269	-0.084	0.4354
	(0.1526)	(0.1216)	(0.1071)	(0.1146)	(0.4123)
Tangibility (t)	-0.1752	-0.0101	0.7109 ^c	-0.1038	0.0702
	(0.1646)	(0.1480)	(0.1326)	(0.1015)	(0.3314)
SD imp. int. rate (t)	0.0595	-0.0198	-0.0371	0.8741°	-0.0267
	(0.0656)	(0.0403)	(0.0278)	(0.1137)	(0.1305)
Tax shield (t)	0.0793^{a}	-0.0446ª	0.0074	0.0052	0.1797
	(0.0425)	(0.0258)	(0.0248)	(0.0188)	(0.1205)

Number of observations (N): 829

Number of firms (N): 243

Average number of years: 3.412

Final GMM criterion Q(b): 1.96e–33

PANEL VECTOR AUTOREGRESSION (GMM ESTIMATION) FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD. OF IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS-BY GROUPS OF FIRMS REGARDING LEVERAGE RATIO (TOTAL LIABILITIES TO TOTAL ASSETS)

	Response to						
Response of	ROAA $(t-1)$	Leverage (t-1))	Tangibility (t-1)	sD imp. int. rate (t-1)	Tax shield (t-1)		
b) Firms with a mean le	verage ratio	higher than (>51.02%)	the median)	or equal to a	the median		
ROAA (t)	0.3408°	0.1812^{b}	-0.0602	-0.0259	-0.3205		
	(0.1135)	(0.0729)	(0.0676)	(0.1260)	(0.3636)		
Leverage (t)	-0.1628	0.6629°	0.0587	-0.067	-0.2643		
	(0.1523)	(0.1035)	(0.1129)	(0.2171)	(0.5654)		
Tangibility (t)	-0.1585	-0.12	0.9836°	0.2273	0.2578		
	(0.1810)	(0.1177)	(0.1691)	(0.2232)	(0.6302)		
SD imp. int. rate (t)	-0.034	-0.0275	-0.0556^{b}	0.606°	0.1205		
	(0.0290)	(0.0201)	(0.0250)	(0.0978)	(0.0880)		
Tax shield (t)	0.0068	0.0179	-0.0068	-0.0535	$0.4555^{\rm b}$		
	(0.0446)	(0.0289)	(0.0338)	(0.0530)	(0.2131)		

Number of observations (N): 916

Number of firms (N): 294

Average number of years: 3.116

Final GMM criterion Q(b): 7.13e-34

Initial weight matrix: identity

GMM weight matrix: robust

^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Standard errors in parenthesis. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. This panel VAR satisfies the stability condition proposed by Hamilton (1994) and Lütkepohl (2005).

P-VAR VARIANCE DECOMPOSITION FOR DETERMINANTS OF CORPORATE FINANCING AND THE 3-YEAR ROLLING SD OF IMPLICIT INTEREST RATE AS A PROXY OF FINANCIAL CONDITIONS-BY GROUPS OF FIRMS REGARDING LEVERAGE RATIO (TOTAL LIABILITIES TO TOTAL ASSETS)

a) Firms with a Mean Leverage Ratio Lower than the Median (<51.02%)

Response variable	Impulse variable						
	ROAA	Leverage	Tangibility	sD imp. int. rate	Tax shield		
ROAA	0.9594	0.0328	0.0056	0.0006	0.0016		
Leverage	0.0066	0.8939	0.0868	0.0053	0.0074		
Tangibility	0.0310	0.0437	0.9130	0.0122	0.0001		
SD imp. int. rate	0.0180	0.0313	0.0900	0.8599	0.0008		
Tax shield	0.2751	0.1762	0.0232	0.0017	0.5238		

b) Firms with a Mean Leverage Ratio Higher than the Median or Equal to the Median (>51.02%)

Response variable	Impulse variable						
	Leverage	Tangibility	Tax shield	sD imp. int. rate	ROAA		
Leverage	0.7852	0.1904	0.0113	0.0010	0.0121		
Tangibility	0.0681	0.9095	0.0118	0.0069	0.0038		
Tax shield	0.0294	0.0198	0.9435	0.0071	0.0003		
SD imp. int. rate	0.0330	0.6676	0.0085	0.2872	0.0037		
ROAA	0.2300	0.0991	0.0943	0.0059	0.5708		

Percent of variation in the row variable (10 years ahead) explained by the column variable. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. The variables were sorted following Granger-Wald causality test criteria.

GMM (BLUNDELL-BOND) REGRESSIONS FOR INVESTMENT					
Percent of the change in fixed assets					
	(1)	(2)	(3)		
Investment (-1)	-0.0285	0.0734^{a}	-0.0524^{a}		
	(0.0303)	(0.0390)	(0.0292)		
Sales growth	27.89°	9.304 ^c	32.68°		
	(3.110)	(1.875)	(3.704)		
Sales growth (-1)	13.59^{b}	10.88^{a}	15.42 ^b		
	(6.929)	(6.554)	(6.674)		
Leverage	0.205				
	(0.134)				
Leverage (-1)	-0.292 ^b				
	(0.139)				
(k-y) (-2)	-6.999°	-5.957°	-7.913°		
	(1.411)	(1.227)	(1.679)		
Interest debt burden	0.374°	-0.00290	0.385 ^c		
	(0.0979)	(0.0619)	(0.113)		
Interest debt burden (-1)	-0.0316	-0.0228	-0.152		
	(0.105)	(0.0517)	(0.111)		
Z-score	0.487	10.53 ^c			
	(2.335)	(1.266)			
Z-score (-1)	0.418	-8.515 ^c			
	(2.351)	(1.260)			
Listed	1.271	1.112	1.786^{a}		
	(0.994)	(0.709)	(1.022)		

	1 1		-	-
1 0	h	0	к	1
1.0				

	(1)	(2)	(3)
Small	-9.489	0.242	-12.08
	(8.215)	(6.393)	(10.70)
Very large	0.492	-0.454	-3.940
	(4.020)	(6.046)	(4.352)
IFD		1.074°	
		(0.0289)	
IFD (-1)		-0.175°	
		(0.0409)	
Tangibility		0.278°	0.251°
		(0.0488)	(0.0650)
Tangibility (-1)		-0.258°	-0.226°
		(0.0425)	(0.0591)
Uniqueness		-0.0605^{b}	-0.0169
		(0.0305)	(0.0587)
Uniqueness (-1)		-0.115°	-0.131^{b}
		(0.0249)	(0.0545)
ROAA			-0.228^{b}
			(0.103)
ROAA (-1)			0.288°
			(0.103)
Constant	22.03 ^b	31.28°	34.89°
	(8.651)	(8.158)	(10.07)

Table B.7 (cont.)

GMM (BLUNDELL-BOND) REGRESSIONS FOR INVESTMENT

Percent of the change in fixed assets

Table B.7 (cont.)

	(1)	(2)	(3)
Country	Yes	Yes	Yes
Year	Yes	Yes	Yes
Industry	Yes	Yes	Yes
Ν	5443	3990	5172
N_g	1219	893	1080
J	74	78	76
Hansen	44.90	38.01	42.18
Hansen-p	0.0810	0.252	0.131
AR1	-9.738	-7.643	-9.547
AR1-p	2.07e-22	2.12e-14	1.33e-21
AR2	-0.989	0.751	-1.330
AR2-p	0.323	0.453	0.183

GMM (BLUNDELL-BOND) REGRESSIONS FOR INVESTMENT

Percent of the change in fixed assets

Standard errors in parentheses ^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Columns 1, 2 and 3 represent the regressions for manufacturing, services, and primary industry conglomerates. The Hansen is a test of the over-identifying restrictions for the GMM estimators. AR1 and AR2 are tests for the first-order and second-order serial correlation. N denotes the number of observations and I number of instruments. Country, Year and Industry denote if their respective dummy variables were introduced in the regressions. Variables are listed as follows: Investment represents the lagged value of the firm's fixed investment; Leverage is the firm's indebtness ratio; Interest debt burden is the ratio of interest paid to operating revenue (%); Sales growth is the annual variation of operating revenue; Listed is a dummy variable for firms that participate in the stock market; ROA is firm's return on assets (%); Z-score is the firm profitability deviation from its capital ratio divided by ROAA's standard deviation, this indicator is expressed in log-transformation; Tangibility assets is the firm's tangible fixed assets to total assets (%); IFD is the firm's internal financing deficit to total assets (%); k-y is the error correction term that reflects how firms adjust their capital towards a target; Small and very large are dummies for small and very large firms according to Orbis disaggregation; and Uniqueness is the firm's cost of goods sold to operating revenue (%).

	1	п.		TD.	0
La	D	10	e	в	ð

Response of	Response to				
	Gross fixed investment growth (t-1)	FC index (t-1)	GDP growth $(t-1)$		
Gross fixed investment growth(t)	-0.861 ^b	0.421 ^b	3.108 ^b		
	(0.350)	(0.194)	(1.497)		
FC index-country median (t)	0.196	-0.150	-0.536		
	(0.377)	(0.229)	(1.622)		
GDP growth (t)	-0.145	0.130^{b}	0.447		
	(0.0976)	(0.0551)	(0.403)		
Number of observations (N):	53				
Number of countries (N):	10				
Average number of years:	5.30				
Final GMM criterion Q(b): 3.04e-32	2				
Initial weight matrix: identity					
GMM weight matrix: robust					

PANEL VECTOR AUTOREGRESSION FOR FINANCIAL CONDITIONS INDEX AND MACROECONOMIC VARIABLES

^a p < 0.10, ^b p < 0.05, ^c p < 0.01. Standard errors in parenthesis. This panel VAR satisfies the stability condition proposed by Hamilton (1994) and Lütkepohl (2005).

Panel VAR-Granger causality Wald test

Ho: Excluded variable does not Granger-cause equation variable Ha: Excluded variable Granger-causes equation variable

Equation	Excluded	Chi-sq	df	Prob > Chi-sq
Gross fixed investment growth (%)	FC index–country median	4.714	1	0.030
	GDP growth (%)	4.310	1	0.038
	All	7.135	2	0.028
FC index-country median	Gross fixed investment growth (%)	0.270	1	0.603
	GDP growth (%)	0.109	1	0.741
	All	0.359	2	0.836
GDP growth (%)	Gross fixed investment growth (%)	2.217	1	0.136
	FC index-country median	5.578	1	0.018
	All	6.721	2	0.035

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE

Total liabilities to total assets



Firms with a mean leverage ratio lower than the median (<51.02%)

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE Total liabilities to total assets

Firms with a mean leverage ratio lower than the median (<51.02%)



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE Total liabilities to total assets



Firms with a mean leverage ratio lower than the median (<51.02%)

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE

Total liabilities to total assets

Firms with a mean leverage ratio higher than the median or equal to the median (>51.02%)



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE

Total liabilities to total assets



Firms with a mean leverage ratio higher than the median or equal to the median (>51.02%)

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE IMPLICIT INTEREST RATE Total liabilities to total assets

Firms with a mean leverage ratio higher than the median or equal to the median (>51.02%)



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE Total liabilities to total assets



Firms with a mean leverage ratio lower than the median (<51.02%)

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE Total liabilities to total assets

Total habilities to total assets



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE Total liabilities to total assets



Firms with a mean leverage ratio lower than the median (<51.02\%)

Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE

Total liabilities to total assets

Firms with a mean leverage ratio higher than the median or equal to the median (>51.02%)



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE

Total liabilities to total assets



Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.

Figure B.4 (cont.)

IMPULSE-RESPONSE FUNCTIONS FOR DETERMINANTS OF CORPORATE FINANCING AND THE SD OF IMPLICIT INTEREST RATE

Total liabilities to total assets





Impulse-response functions derived by Cholesky's variance decomposition. All variables were transformed using forward orthogonalization suggested by Arellano and Bover (1995), through the Helmert procedure. All country effects were included by subtracting the means of each variable calculated for each country-year. Variables were sorted following Granger-Wald causality test criteria. Confidence intervals were generated by a Monte-Carlo simulation with 1,000 repetitions.



Figure B.6



Selected ICFC quantiles for the ten sample countries, own calculations and WB-WDI





IMPULSE-RESPONSES OF THE PANEL VECTOR AUTOREGRESSION FOR FINANCIAL CONDITIONS INDEX AND MACROECONOMIC VARIABLES

References

- Abrigo, M. R., and I. Love (2016), "Estimation of Panel Vector Autoregression in Stata," *The Stata Journal*, Vol. 16, No. 3, pp. 778-804.
- Arellano, M., and S. Bond (1991), "Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations," *The Review of Economic Studies*, Vol. 58, No. 2, pp. 277-297.
- Arellano, M., and O. Bover (1995), "Another Look at the Instrumental Variable Estimation of Error-components Models," *Journal of Econometrics*, Vol. 68, No. 1, pp. 29-51.
- Balfoussia, Hiona, and Heather D. Gibson (2016), Firm Investment and Financial Conditions in the Euro Area: Evidence from Firm-level Data, Working Paper, No. 208, Bank of Greece.
- Blundell, R., and S. Bond (1998), "Initial Conditions and Moment Restrictions in Dynamic Panel Data Models," *Journal of Econometrics*, Vol. 87, No. 1, pp. 115-143.
- Bond, S., D. Harhoff, and J. van Reenen (1999), Investment, R&D and Financial Constraints in Britain and Germany, Working Paper, No. 99/5, Institute for Fiscal Studies.
- Bradley, M., G. A. Jarrell, and E. Kim (1984), "On the Existence of an Optimal Capital Structure: Theory and Evidence," *The Journal of Finance*, Vol. 39, No. 3, pp. 857-878.
- DeAngelo, H., and R. W. Masulis (1980), "Optimal Capital Structure under Corporate and Personal Taxation," *Journal of Financial Economics*, Vol. 8, No. 1, pp. 3-29.
- Fama, E. F., and K. R. French (2002), "Testing Trade-off and Pecking Order Predictions about Dividends and Debt," *Review* of *Financial Studies*, Vol. 15, No. 1, pp. 1-33.
- Flannery, M. J., and K. P. Rangan (2006), "Partial Adjustment toward Target Capital Structures," *Journal of Financial Economics*, Vol. 79, No. 3, pp. 469-506.
- Frank, M. Z., and V. K. Goyal (2007), "Trade-off and Pecking Order Theories of Debt," in B. Espen Eckbo, *Handbook of Empirical Corporate Finance*, Vol. 2 (2008), pp. 135-202.
- Fuertes, A., and J. M. Serena (2014), "Firms' Financial Soundness and Access to Capital Markets," *Financial Stability Review*, Vol. 27, Banco de España, pp. 113-132.
- Gilchrist, S., and C. Himmelberg (1998), *Investment, Fundamentals* and Finance, NBER Working Paper, No. 6652.

- Gilchrist, Simon, Raphael Schoenle, Jae W. Sim, and Egon Zakrajsek (2015), *Inflation Dynamics during the Financial Crisis*, Finance and Economics Discussion Series, No. 2015-012, Board of Governors of the Federal Reserve System, Washington, <http://dx.doi.org/10.17016/FEDS.2015.012>.
- Gomes, Joao F. (2001), "Financing Investment," American Economic Review, pp. 1263-1285.
- Hamilton, J. D. (1994), *Time Series Analysis*, Vol. 2, Princeton University Press, Princeton.
- Harris, M., and A. Raviv (1991), "The Theory of Capital Structure," *The Journal of Finance*, Vol. 46, No. 1, pp. 297-355.
- Hattori, M., and E. Takats (2015), *The Role of Debt Securities Markets*, BIS Paper, No. 83c.
- Hernando, Ignacio, and Carmen Martínez-Carrascal (2008), "The Impact of Financial Variables on Firms' Real Decisions: Evidence from Spanish Firm-level Data," *Journal of Macroeconomics*, Vol. 30, No. 1, pp. 543-561.
- Huang, R., and J. R. Ritter (2009), "Testing Theories of Capital Structure and Estimating the Speed of Adjustment," *Journal* of Financial and Quantitative Analysis, Vol. 44, No. 02, pp. 237-271.
- Jensen, M. C., and W. H. Meckling (1976), "Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure," *Journal of Financial Economics*, Vol. 3, No. 4, pp. 305-360.
- La Rocca, M., T. La Rocca, and A. Cariola (2011), "Capital Structure Decisions during a Firm's Life Cycle," *Small Business Economics*, Vol. 37, pp. 107-130.
- Lemmon, M. L., M. R. Roberts, and J. F. Zender (2008), "Back to the Beginning: Persistence and the Cross-section of Corporate Capital Structure," *The Journal of Finance*, Vol. 63, No. 4, pp. 1575-1608.
- López-García, J., and F. Sogorb-Mira (2008), "Testing Trade-off and Pecking Order Theories Financing SMEs," Small Business Economics, Vol. 31, No. 2, pp. 117-136.
- Love, I., and I. Zicchino (2006), "Financial Development and Dynamic Investment Behavior: Evidence from Panel VAR," *The Quarterly Review of Economics and Finance*, Vol. 46, No. 2, pp. 190-210.
- Lütkepohl, H. (2005), New Introduction to Multiple Time Series Analysis, Springer Science & Business Media.
- Michaelas, N., F. Chittenden and P. Poutziouris (1999), "Financial Policy and Capital Structure Choice in UK SMEs: Empirical Evidence from Company Panel Data," *SmallBusiness Economics*, Vol. 12, No. 2, pp. 113-130.
- Myers, S. C. (1977), "Determinants of Corporate Borrowing," *Journal* of Financial Economics, Vol. 5, No. 2, pp. 147-175.
- Myers, S. C. (1984), "The Capital Structure Puzzle," *The Journal of Finance*, Vol. 39, No. 3, pp. 574-592.
- Myers, S. C., and N. S. Majluf (1984), "Corporate Financing and Investment Decisions When Firms Have Information that Investors Do Not Have," *Journal of Financial Economics*, Vol. 13, No. 2, pp. 187-221.
- Rajan, R. G., and L. Zingales (1995), "What Do We Know about Capital Structure? Some Evidence from International Data," *The Journal of Finance*, Vol. 50, No. 5, pp. 1421-1460.
- Ramalho, J.S., and J. V. da Silva (2009), "A Two-part fractional Regression Model for the Financial Leverage Decisions of Micro, Small, Medium and Large Firms," *Quantitative Finance*, Vol. 9, No. 5, pp. 621-636.
- Serrasqueiro, Z., and A. Caetano (2015), "Trade-off Theory versus Pecking Order Theory: Capital Structure Decisions in a Peripheral Region of Portugal," *Journal of Business Economics* and Management, Vol. 16, No. 2, pp. 445-466.
- Shyam-Sunder, L., and S. C. Myers (1999), "Testing Static Tradeoff against Pecking Order Models of Capital Structure," *Journal* of Financial Economics, Vol. 51, No. 2, pp. 219-244.
- Smith, C. W., and R. M. Stulz (1985), "The Determinants of Firms' Hedging Policies," *Journal of Financial and Quantitative Anal*ysis, Vol. 20, No. 04, pp. 391-405.
- Sogorb-Mira, F. (2005), "How SME Uniqueness Affects Capital Structure: Evidence from a 1994-1998 Spanish Data Panel," *Small Business Economics*, Vol. 25, No. 5, pp. 447-457.
- Titman, S., and R. Wessels (1988), "The Determinants of Capital Structure Choice," *The Journal of Finance*, Vol. 43, No. 1, pp. 1-19.
- Windmeijer, F. (2005), "A Finite Sample Correction for the Variance of Linear Efficient Two-step GMM Estimators," *Journal* of Econometrics, Vol. 126, No. 1, pp. 25-51.

The Demand for Credit at the Individual Level: The Credit Registry (RCC) Meets the National Household Survey (ENAHO)

Nikita Céspedes Reynaga

Abstract

This article examines the demand for credit at the individual level in Peru. It uses a unique database resulting from merging the Credit Registry (RCC) and the National Household Survey (ENAHO). The data allows for ideally identifying the amount of credit and the interest rate as well as the characteristics of each credit granted in the Peruvian banking system. It also includes indicators of the supply of each credit, which is key for the identification of demand. The elasticity of the demand for credit relative to the interest rate is estimated using a two-step procedure proposed by Heckman (1979) and is approximately -0.29. This value means that a rise in the market interest rate by 1% implies a reduction in the demand for credit by 0.29%. This elasticity is slightly lower than the one provided by international evidence and is highly heterogeneous throughout credit types and features of individual debtors.

Keywords: demand for credit, balance sheet effect, heterogeneity. *JEL classification: E21, E44, E51, E52.*

N. Céspedes Reynaga, Banco Central de Reserva del Perú <nikitacr@gmail.com>. The author would like to thank Paul Castillo, Zenón Quispe, and participants at the Banco Central de Reserva del Perú's Economic Research Seminar for their valuable comments. The author is also grateful for the information provided by Eduardo Costa, Amílcar Vélez, Nancy Hidalgo, and Aníbal Galarreta. The opinions expressed in this paper are the sole responsibility of the author.

1. INTRODUCTION

redit as a funding mechanism for firms and households is crucial to a country's economic development. Study of the fundamentals influencing borrowing by households and firms has received considerable attention from academics and economic policymakers during the years following the international financial crisis that broke out in 2008. The reason behind such increased interest has been the growing participation of individuals in the credit market, allowing them to receive the benefits of said market, and at the same time exposing them to financial fluctuations. This is the case, for instance, of the 2008 financial crisis, the effects of which have spread beyond the business sector, extending to the household segment.¹This paper examines the characteristics of credit at the individual level in Peru while also estimating the demand for debt in this segment of economic agents. We believe the study is justified by the growing participation of households in Peru's formal credit market. Moreover, the Peruvian economy and its credit market have institutional and idiosyncratic characteristics, such as the dollarization of loans, its inflation targeting scheme and the economy's high level of exposure to external crises that make it different from others.

With respect to aggregate trends, household credit at the international level has been growing during recent decades (IMF, 2012), and Peru has seen a similar behavior.² Hence, between

¹ There is a large body of international literature on this subject suggesting business credit has positive effects on economic growth through higher investment and the resulting increased accumulation of physical capital. Meanwhile credit to households has a less clear impact on growth, functioning more as a mechanism that can improve households' wellbeing by the intertemporal smoothing of consumption during any adverse shocks they face (Hall, 1978).

² Diverse factors have contributed to the expansion of credit, among which stand out: low inflation and interest rates, higher income and wages within a context of strong economic growth,

2001 and 2016 consumer credit grew at an average annual rate of 19%, increasing as a percentage of GDP from 4.2% in 2001 to 14.8% in 2016 (Figure 1). This significant growth in credit has been enough to change the composition of credit between consumers and firms. Thus, in 2001 consumer loans accounted for 18% of total credit and in 2016 this figure had increased to 37%. In this regard, international evidence suggests that the significant growth of consumer loans as a proportion of total credit could represent a source of vulnerability for this segment of the population during adverse events, both for the financial system and households themselves (BIS, 2006; IMF, 2016). The latter point is another reason to study and understand the characteristics of the determinants of household debt in Peru.

Another useful aspect of this study is an estimation of the elasticity of demand for credit, which under stable financial conditions allows for measuring the necessary adjustments in monetary policy rates aimed at correcting deviations in inflation with respect to price stability levels through the credit transmission channel in line with Bernanke and Blinder (1988). In general terms, this elasticity captures the transmission to households of shifts in the financial system (credit supply shocks) as a result, for instance, of changes in Reserva Federal's monetary policy and external financial crises propagated through international credit restrictions. The latter being the case of the 2008-2009 global financial crisis that marked the beginning of higher financing costs for small economies like Peru.

This paper reveals new evidence regarding the importance of dollarization in Peru's debt market. It should be remembered that dollarization has been one of the (greatest) vulnerabilities in the Peruvian economy since the beginning of the nineties. Dollarization of credit reached historically high levels in December 1999 when loans in dollars represented 81.7% of total

opening of capital markets, larger capital flows and improved credit offerings under an environment of positive macroeconomic performance reflected in low country-risk levels, etcetera.



Note: Credit (balances) of deposit entities to private is shown, by type of credit. Source: Banco Central de Reserva del Perú. credit.³ Towards the end of 2015, and after a prolonged struggle in economic policy and macroeconomic stability terms, dollarization has fallen to around 30% of total credit. Given that dollarization is a phenomenon affecting only a small number of countries, empirical studies on foreign currency borrowing has been limited to a few particular cases ⁴ and no studies have been carried out for Perú.⁵

Literature examining credit dollarization at the individual level is very limited internationally, and in the case of Peru, nonexistent. For instance, Beer et al. (2010) analyze the borrowing behavior of Austrian households and estimate the influence of household characteristics, which are divided into subjective factors (e.g. risk perception, financial literacy and level of education) and objective factors (e.g., sociodemographic).

³ Corresponds to the dollarization ratio (%) of deposits firms to private banks (end of period).

⁴ Internationally, there are several authors who study foreign currency borrowing at the aggregate level, while a smaller number of papers characterize the demand for credit among firms. For instance, Brown et al. (2011) and Cowan et al. (2005) include enterprise level features in a theoretical model examining the borrowing behavior of small firms. Those models emphasize the role of institutional infrastructure and compliance, imperfect bank information and the monetary composition of income. Brown et al. (2011) consider various micro-level determinants of borrowing among firms in Bulgaria (employing enterprise level data for loans between 2003 and 2007). Their model includes supply features (bank characteristics) and demand determinants (firm characteristics) of loans in foreign currency. Their findings demonstrate that comparatively larger and older firms, as well as those with lower bailout costs in case of default, demand more loans in foreign currency. Moreover, banks grant loans in foreign currency mainly for fixed investments and long-term projects.

⁵ Data employed correspond to periods of high dollarization, even at the household credit level, meaning estimates from the study could be used to characterize the potential effects of external shocks on households' standards of living through the credit channel in domestic as well as foreign currencies.

According to their results, foreign currency borrowers tend to be risk seeking, older, financially literate and more affluent. Pellényi and Bilek (2009) present a study of survey data on foreign currency borrowing among households in Eastern Europe. They analyzed survey data collected in 2008 for Hungarian households and find that foreign currency borrowers tend to be less risk-loving and better aware of exchange rate risks.⁶

This paper studies the demand for loans among individuals using data disaggregated to the level of each loan. This procedure represents an advance in the literature on the credit market, especially in the case of Peru, for which no published papers are to be found on the subject.⁷

The study employs a unique database resulting from a merging of the National Household Survey (ENAHO) and the Credit Registry (RCC), which because it is an administrative registry allows for identifying without measurement errors the amount and interest rate of each individual bank loan by type of credit and currency, as well as the features of individual debtors. This consideration makes it possible to characterize the heterogeneity of credit according to the observable characteristics of individual debtors by currency type, age, income levels, region of residence, employment and informality, among others. After characterizing credit, we estimate the demand for credit at the individual level. This demand allows for identifying the sensitivity of credit to changes in interest rates after

⁶ There is also some recent literature that examines credit demand using data at the individual level, such as that of Fidrmuc et al. (2013) which studies the determinants of foreign currency borrowing in nine Eastern European countries and finds that a lack of confidence in local currency stability among households is an important consideration when taking out loans in foreign currency.

⁷ Papers on household credit in Peru include those of Cámara et al. (2013) and Alfageme and Ramírez-Rondán (2016), who use the ENAHO to perform a general study of the determinants of participation in the mortgage market.

controlling for the observable characteristics of demand and institutional features of credit supply. The estimation method consists of a two-step process (Heckman, 1979). In the first step, we estimate an equation for credit market participation and in the second an equation for credit demand that relates credit with interest rates and a group of relevant controls.

The results highlight a significant degree of heterogeneity of credit according to the observable features of individuals. One level of heterogeneity that stands out concerns an individual's income. Those with access to formal credit have high incomes. In line with the latter, credit in Lima is concentrated among middle-aged and better educated individuals, while informal workers are also seen to have access to formal credit. As for the elasticity of demand for credit relative to the interest rate, the estimation reveals that this is -0. 29, figure heterogeneous according to several observable features of individuals such as the type of credit, the currency in which it is granted, geographic region and informality. Moreover, the average elasticity found is lower than those estimated by the literature employing similar quality administrative data, which is consistent with the existence of an inelastic and uncompetitive credit market.

The rest of the paper is organized as follows. Section 2 presents data sources and explains the methodology for constructing the data. Section 3 discusses the heterogeneity of credit according to different categories of individuals, and Section 4 presents the model that justifies the credit demand equation. Section 5 shows the econometric estimation and Section 6 summarizes the main findings.

2. DATA

Data is taken from two sources. Firstly, there is administrative data for each loan granted to individuals by financial entities registered in the Credit Registry (RCC). This information is collected each month by the Superintendence of Banking and Insurance and the number of registries represents the whole population with loan obligations in the banking system. Information from the RCC corresponds to the credit balance of each individual by the banking institution. It is worth pointing out that this information discloses all loans held by an individual, identifying credit type and currency. The number of loans registered varies according to the month studied. Thus, in December 2014, for instance, 12.4 million loans were included, corresponding to a total of 5.7 million individuals with loans in the formal banking system.

The other source of information is the National Household Survey (ENAHO) conducted every year by the National Institute of Statistics and Information (INEI). This database collects information on diverse aspects, such as an individual's employment and personal data, that allow for identifying credit demand characteristics. The two databases are merged using the National Identity Document (DNI) and the names and surnames of each individual for data between 2008 and 2014 as common links, obtaining a total of 95 037 individuals in both databases. Considering that around 500 thousand individuals are registered in the ENAHO, the number with loans in the final sample during those years represents everyone in Peru that accessed formal credit in said period.

The credit sample in the final database is representative at the national level. This assertion is substantiated by comparing credit indicators estimated in the final database with those estimated in the RCC and the ENAHO. Hence, there is a similarity between the proportion of individuals reported in the final database and the corresponding value reported by the original data in the RCC (see Table 1). The RCC is used to estimate the share of individuals in the banking system with credit. As expected, the latter value is lower than the total share of individuals with credit in the banking system as well as other institutions (informal).

Tab	le I	

	RCC	ENAHO and RCC
Mortgage dollarization (no. of loans)	32.0	35.0
Mortgage dollarization (balances)	34.6	38.1
Consumer credit dollarization (no. of loans)	4.2	3.3
Consumer credit dollarization (balances)	7.7	5.9
Notes: the second column (RCC) corresponds to 2014. Sources: ENAHO, RCC, 2008-2014.		

CHARACTERISTICS OF CREDIT IN THE SAMPLE Percentages

Another useful indicator for controlling the quality of data employed is aggregate credit dollarization information. The latter makes it possible to verify that household debt dollarization trends reported by the BCR are similar to those reported in the final database obtained by merging the ENAHO and the RCC (Figure 2).

Estimating the demand for credit requires indicators regarding the interest rates on each loan. Although no direct measure of interest rates is available, in this paper we estimate the implicit interest rate on each credit using the yields of the loans, which in practice correspond to the monthly interest charged (accrued) by financial institutions on the loans they grant. The indicator for average interest rates calculated in this way is closely related to the interest rates published by the SBS, thereby upholding the known stylized facts for said indicator, such as, for instance, mortgage rates are lower, consumer loans have lower rates and average interest rates during the study period have followed a downward trend.



Note: Percentage of credit balances in foreign currency to total of credit. Sources: SBS, BCRP, ENAHO, and RCC, 2008-2014.

3. CHARACTERISTICS OF CREDIT AT THE HOUSEHOLD LEVEL

3.1 Descriptive Statistics

Credit and interest rates should be expressed in logs in the empirical model. This is particularly useful here because in order to guarantee the efficiency of model estimators. The first and second moments of the series employed must be well-defined. A casual inspection of the series suggests that these have a normal log distribution. The reason is that there is a considerable number of individuals with micro loans and a very small proportion with very large ones. Estimates for interest rates behave similarly, the use of logs, therefore, normalizes the series and guarantees stability in the variance of the estimators.⁸ With this taken into consideration, the descriptive statistics correspond to the log series.

Credit heterogeneity is noteworthy at the level of its principle moments, meaning the estimation should control for those average effects. Average mortgage credit is larger than credit to small firms and consumer credit. Moreover, there are differences in terms of loans denominated in domestic currency and those in foreign currency. The data in Table 2 also reveal that there is heterogeneity with respect to the observable characteristics of individuals such as age, income, and region of residence, among others. This heterogeneity found regarding loan size is also seen in terms of estimated implicit interest rates as shown in Table 3. These two stylized facts suggest that the regression estimated to measure the elasticity of the demand for credit should be controlled for the heterogeneity of demand.

3.2 Correlation between Interest Rates and Credit

Aggregate data suggest the likelihood of a negative correlation between credit and interest rates as illustrated in Figure 3 for data between 1992 and 2016. Nevertheless, this aggregate correlation might not be correct in all the study periods. The correlation is positive, for instance, in the years between 2004 and 2008. Estimation of aggregate demand for credit should also be corrected for the influence of macroeconomic type variables. Furthermore, although it is not documented, estimation of the demand for credit might be susceptible to aggregation biases. With these considerations in mind, we take into account that estimation of the demand for credit among agents can properly identify the elasticity we aim to calculate.

⁸ A comparison of the distribution of credit in levels as well as logs reveals that the log series has an approximately normal distribution as illustrated in Figures A.1 to A.4 in the Annex.

				Size of (credit in	logs Tybe of	^c Credit					
		Consur	nption			Smal	l firm			Mort	gage	
	D	ç	F	0	D	2	H	5	DQ	5	F(
	\overline{X}	SD	\bar{X}	SD	\overline{X}	SD	\bar{X}	SD	\bar{X}	SD	\bar{X}	SD
Average	7.91	2.36	7.33	3.10	8.54	2.23	12.06	2.51	11.57	1.45	11.48	1.96
Quintiles												
Ι	8.07	2.43	7.02	2.87	8.38	2.30	13.26	1.79	12.15	2.12	11.67	1.59
II	8.20	2.51	9.14	3.65	8.50	2.28	12.22	2.74	10.78	1.56	13.42	2.59
III	7.99	2.43	7.52	3.87	8.53	2.22	11.76	2.35	11.01	1.06	12.44	2.15
IV	7.85	2.31	6.96	3.38	8.53	2.16	11.45	2.23	11.42	1.54	11.22	2.04
Λ	7.88	2.32	7.36	2.90	8.79	2.18	12.12	2.52	11.71	1.46	11.47	1.88
Age												
17 to 24	7.56	2.07	7.21	3.10	8.19	2.26	13.1	2.35	11.26	1.46	10.91	1.16
25 to 34	7.60	2.24	6.78	2.91	8.49	2.24	12.56	2.30	11.71	1.13	12.73	2.01
35 to 44	7.94	2.37	7.60	3.10	8.54	2.22	11.04	2.47	11.57	1.50	11.27	1.73

Table 2

Monetaria, January-June, 2018

45 to 54	8.15	2.40	7.26	3.20	8.59	2.22	12.51	2.40	11.47	1.64	11.36	1.85
55 to 100	8.11	2.47	7.90	3.13	8.72	2.23	11.78	2.53	11.59	1.39	11.46	2.31
Region												
North Coast	7.80	2.25	7.04	3.49	8.18	2.12	11.09	1.91	11.26	1.31	11.41	1.78
Central Coast	7.73	1.99	6.68	3.24	8.29	2.03	10.97	3.00	10.39	0.88	10.93	1.77
South Coast	8.11	2.22	7.22	2.75	8.69	2.25	11.34	2.55	11.38	1.66	11.04	2.02
North Mountain Range	8.40	2.43	8.39	3.16	8.26	2.12	11.31	1.84	11.26	1.39	10.59	2.53
Central Mountain Range	8.13	2.15	7.81	3.99	8.34	2.17	12.04	1.97	11.54	1.33	11.98	2.01
South Mountain Range	8.18	2.47	7.08	2.96	8.78	2.29	12.03	2.54	11.31	1.41	12.18	2.20
Jungle	8.55	2.28	7.57	3.56	8.57	2.19	13.69	2.26	11.30	1.17	12.00	2.34
Lima metropolitan area	7.71	2.45	7.34	2.99	8.80	2.35	12.64	2.34	12.14	1.49	11.46	1.88
Without remittances	7.91	2.36	7.33	3.10	8.54	2.23	12.07	2.51	11.57	1.45	11.44	1.93
With remittances	8.07	2.45	6.74	3.22	8.92	2.40	10.85	I	I	I	15.44	0.29
Note: Shows the average \overline{X} all currency. Note that the log ave better characterizes the series Source: ENAHO, RCC, 2008-201	nd standa erage of c under cc 14.	ard devia rredit is sl nsiderati	tion (SD) hown giv on.	of credit en that th	per indi ne distrib	vidual. D ution of	C denotes credit is lo	domesti gnorma	c currenc l and in th	y and FC 1is case t	foreign he use of	ogs



Banking system credit to private sector (end of period) - credit in foreign currency (millions of dollars)

Note: Identify the credit demand requires credit supply indicators. This identification is not possible at macro level.

Source: Banco Central de Reserva del Perú.

The correlation between credit and interest rates is difficult to identify using a scatter plot between credit and interest rates as seen in Figure 4. The latter shows the correlations for all loans considered (consumer, small business, mortgage), differentiating between the loan denomination currency. These five figures, together with the descriptive statistics presented, help to suggest that an estimation of the demand for credit requires the inclusion of additional controls on the supply as well as the demand side.

4. THE DEMAND FOR CREDIT MODEL

Credit demand compares the size of the loan with the interest rate through a reduced form that can be deduced from a household optimization equation. This equation is the simplest case where households decide the amount of credit based on their fundamentals with respect to sources of income and different preferences represented by an aversion indicator, their level of impatience, and the interest rate they face. Formally, we follow the representation of the consumption-savings intertemporal choice model of Hall (1978), whose household optimization equation is as follows:

$$\max\sum_{t=1}^{\infty}oldsymbol{eta}^{t-1}Uig(c_tig),$$

subject to:

1

2

3

$$c_t + b_t = y_t + (1 + r_t)b_{t-1}; \quad \forall t = 1, 2, \dots$$

 $c_t \ge 0$



Note: Credit is on the y-axis, and the interest rate is on the x-axis. Variables are in logs. Sources: RCC and ENAHO, 2008-2014.

$$\lim_{t\to\infty}\left[\frac{1}{1+r_t}\right]^{t+1}b_t\to 0,$$

4

where c_t is consumption and b_t a household's bond holdings in period t, with $b_t < 0$ representing the size of a household's credit. The preferences of each household at every moment, which for simplicity sake we assume has only one member, are described by the following utility function $u_t = \frac{c_t^{1-\sigma}}{1-\sigma}$. We include the usual assumptions $u_c(\cdot) > 0$, i.e., consumption generates positive utility in the individuals. The budget constraints families face in each period capture the equivalence between resource funds and uses, $c_t + b_t = y_t + (1+r_t)b_{t-1}$. Household income is y_t , r_t the interest rate, β the subjective discount factor, and σ the risk aversion parameter. The last two equations (3 and 4) represent the positive consumption constraint and the trasversality condition, respectively.

The solution to this problem is a set of optimum values for consumption and the amount of credit for every value of t=1, 2, ..., which take the following values after considering a constant interest rate over time and an initial amount of debt (b_0) :

$$c_{t} = \left\{\beta(1+r)\right\}^{\frac{t}{\sigma}} \frac{\sum_{t=1}^{\infty} \left[\frac{1}{1+r_{t}}\right]^{t-1} y_{t}}{\sum_{t=1}^{\infty} \left[\frac{1}{1+r_{t}}\right]^{t-1} \beta^{\frac{t}{\sigma}} (1+r)^{\frac{t}{\sigma}-t+1}},$$
$$b_{t} = (1+r)^{t} b_{0} + \sum_{j=1}^{t-1} (1+r)^{t-j} \left\{y_{j} - c_{j}\right\}.$$

In the equation above, credit corresponds to a representative individual and is defined by the measure of household risk aversion, level of impatience, interest rate and income. Nevertheless, the empirical section uses a reduced form for credit

5

6

demand with different degrees of heterogeneity, which can be better justified if the heterogeneity of credit is explicit in the derivation of credit demand. To generate credit heterogeneity in the aforementioned equation it is only necessary to include the existence of different individuals with varying income levels. This idea can also be strengthened by including heterogeneity in the values for risk aversion and level of impatience, parameters that are considered heterogeneous by a large body of literature. Another item that can be used to generate heterogeneity is the interest rate.9 The data employed suggest that individuals access the credit market at heterogenous rates that change over time, which captures risk profiles at the individual level from the perspective of credit granting institutions. The econometric estimation considers these different levels of heterogeneity by including observable features for individuals, their income and interest rates.

4.1 Reduced Form of the Demand for Credit

The previous expression represents the solution for each household under ideal credit conditions. It is easy to deduce from this sequence that credit depends negatively on interest rates for all individuals whose current income levels are below their corresponding consumption $(y_j > c_j)$. It can also be seen that said dependence is non-linear. Another characteristic that can be inferred from the equation is that other determinants of credit are present, such as risk aversion and level of impatience. We, therefore, summarize the equation into a reduced form that linearly relates the log of credit and the interest rate.

Empirical estimation of this reduced form requires prior consideration of some specific features of Peru's credit market. One such consideration is participation in the credit market. In

⁹ Among the first papers to estimate preference heterogeneity and risk aversion in particular are: Barsky et al. (1997), Kimball et al. (2008) and Kimball et al. (2009).

practical terms the sample of individuals who access the credit market might be different from those who do not. If this is the case, the estimated parameters of the demand for credit equation might contain so-called sample selection bias. This problem is solved by including a Heckman correction, which basically suggests that the demand for credit should be estimated using a two-step approach. The first step consists of estimating the credit market participation equation using the whole sample, while the second corresponds to estimating the intensive demand for credit considering only the sample of individuals with credit.

Credit Market Participation

Participation in the credit market is only observed for those who manage to obtain a loan, and this only takes place after a credit assessment process that is not observed in the data. The data only shows individuals who participate in the credit market, which is denoted by $I_{ijl} = 1$. This event is related to a continuous and latent variable I^* that is determined by a set of variables, grouped in x, that identify participation in the credit market through the equation

$$I_{ijt}^* = \delta x_{ijt} + \varepsilon_{ijt}.$$

We can see that the data registers credit market participation $(I_{ijt} = 1)$ only if the latent variable is positive $(I_{ijt}^* > 0)$, where *i* denotes each individual and *t* the period. With this in mind, and assuming a normal distribution of random component ε_{ijt} , the probability of participation in the credit market is expressed as follows:

$$Pr(I_{ijt}=1) = Prob(\delta x_{ijt} + \varepsilon_{ijt} > 0) = \Phi(\delta x_{ijt}),$$

where as usual Φ represents the normal distribution characterizing the probit model.

7

Intensive Demand for Credit

In the second stage, we define the amount of credit (b), that depends on a set of variables divided into demand-side and supply-side. The equation to be estimated is as follows:

8
$$b_{ijt}^n = \alpha + \beta_r R_{ijt} + \beta x_{ijt} + \theta z_{jt} + \delta T_t + \gamma \lambda_{ijt} + v_{ijt},$$

where b_{ijt} is the demand for credit in period t for household i in bank j, and R_{ijt} is the interest rate. x_{ijt} are the controls representing different levels of heterogeneity among individuals and z_{ijt} are the controls per bank (j), and T_t captures aggregate variables that are known to affect the credit market. v is the error term that captures the determinants of credit that are not considered. The aforementioned specification includes the inverse Mills ratio λ_{ijt} , to correct the sample selection problem and also connect intensive demand with the estimation of extensive demand from the first stage.

The types of heterogeneity considered include observable features of individuals in terms of the level of education, age, and geographical region. We also include other less structural indicators for individuals captured in the type of employment (formal and informal) and by the shocks they experience, among which stand out employment, demographics, etc. Although there is only a small amount of literature on the subject, we believe the levels of heterogeneity employed capture probable differences in preferences (risk aversion, impatience, etc.) among households with respect to borrowing. This differentiation can be made according to the type of credit (consumer, mortgage) or the currency in which a loan is taken out, i.e. domestic or foreign. The latter separation allows for studying differences in the determinants of credit by type of currency.

One important aspect of the estimation is identifying the demand for debt achieved by including credit supply related variables measured for each credit granting institution in the estimation. The estimation contains identifiers of formal financial institutions using binary variables. 10

It is also important to mention that, in the case of Peru, credit supply characteristics should include the potential role of exchange rate variations and their influence on the interest rates at which financial institutions offer loans. By including a binary variable at the level of the main banks and their interaction with time and loan currency, we are implicitly controlling for the expected devaluation such institutions incorporate into their loans. Besides expected changes in the exchange rate, this interaction effect also captures the institutional characteristics of Peru's banking system. Among the latter stand out, for instance, the high interest rates charged by small financial entities, while larger institutions report lower rates as mentioned in Céspedes and Orrego (2014).

5. RESULTS OF THE ESTIMATIONS

5.1 Credit Market Participation

Participation in the credit market is represented using a discrete choice probit model where the explanatory variable takes a value of one if an individual has a loan and zero if not. Among the variables that determine access to credit, we have a set of indicators commonly used in the literature to capture different aspects of participation. Hence, we consider the following.

• Property rentals: Individuals with property rentals have a regular flow of income they can use to ensure loan repayments. This variable identifies credit market participation in the data as illustrated in Panel A of Figure 5. The latter figure shows that access to credit for individuals with property rentals is clearly higher than that of individuals without such income, a fact sustained on average for all age groups considered.

¹⁰ It includes 26 artificial variables. The first 25 correspond to the largest financial institutions, while the remaining financial entities (small ones) are grouped into variable 26.

- Remittances: External remittances received by workers constitute a source of income that could be used as a mechanism for hedging loans, particularly those in foreign currency. This variable has been widely used in the literature on foreign currency borrowing, for instance: Fidrmuc et al. (2013) show the importance of external remittances in the demand for loans in Eastern European countries.
- Age and age-squared: The use of an individual's age as a variable to identify their participation in the credit market obeys the shape of inverted U of the age of participation in the credit market. Along the same lines, age-squared captures lower credit market participation among young and old-aged people, while the middle-aged participate more in said market.
- Shocks faced by households: This variable captures one characteristic of the credit theory as an insurance mechanism for responding to the shocks a household faces. According to this argument, households smooth consumption by using the credit market to face adverse events at the expense of future income. A set of shocks faced by households are considered such as, for instance: demographic shocks, employment shocks, natural disasters, etc. The reported index takes into account that an individual has been exposed to one of the shocks during the last 12 months.
- Informality: We consider that formal employment among workers identifies credit market participation basically using the characteristics of the credit database, which is limited to formal bank credit. In our data, formal workers access the credit market, while those in informal employment exhibit a much smaller access as shown in Panel D of Figure 5.

The participation equation estimated includes an additional set of controls such as gender, marital status, and region of residence. The estimation results provide a good fit in econometric terms as displayed in Table 3. Note that the variables identifying the selection are significant and also have signs consistent with that shown in Figure 3. These results are comparable to the estimates of Alfageme and Ramírez-Rondán (2016), and Cámara et al. (2013), although the size of the differences could be related to



Note: Participation in the credit market (percentage of persons with credit in each category). Source: ENAHO and RCC, 2008-2014.

N. Céspedes

the inclusion of a set of variables identifying credit market participation, such as age, remittances, shocks and property rentals.

5.2 The Demand for Credit

The estimated elasticity of the demand for credit in Peruis -0.29. This figure is obtained after controlling for credit supply variables and Peruvian institutional characteristics as presented in Table 4. The latter table also presents estimated elasticities of the demand for credit using different specifications and estimation methods. This procedure highlights that the ordinary least squares estimator is not much different from the value estimated using Heckman's two-step method. The result that stands out is the fact that the elasticity of demand for credit is small, making it possible to conclude there is low credit market sensitivity among individuals during shocks channeled through interest rates. One of these events that occur relatively often are changes in Peruvian or US monetary policy. According to the results of this study, such changes would have had a modest impact on the demand for credit among individuals. International evidence on the value of this elasticity is mixed. On the one hand, Gross and Souleles (2002) use credit card records in the USA to find an elasticity of demand for credit of -1.3, which indicates a substantial reaction in credit demand (cards) to the interest rate. Meanwhile, another commonly referred to paper is that of Alessie et al. (2005), who use administrative data from a leading institution in Italy to find an elasticity of credit (instalment, revolving, personal loan) relative to the interest rate of -1.2 between 1995 and 1999. Nevertheless, this stance does not command full consensus because there is also literature suggesting a low elasticity of demand for credit. For instance, Ausubel (1991) includes one of the stylized facts most used for credit demand. This author employs administrative records and reveals that the demand for credit in the USA is rigid with respect to the interest rate and suggests that credit card holders rarely react to interest rate changes.

	\sim		
	Coefficients	z test	
Remittance (=0)	-0.0689^{b}	(-2.68)	
Property rentals	0.213 ^c	(23.58)	
Age	0.0842°	(101.10)	
Age x age	-0.000870°	(-100.48)	
Informal (=1)	-0.673°	(-138.24)	
Married	-0.0320°	(-5.29)	
Widowed	-0.171°	(-13.47)	
Divorced	0.0150	(0.51)	
Separated	-0.0378°	(-4.25)	
Single	-0.178°	(-24.01)	
Effect of shocks			
Reduced income	-0.0801°	(-13.09)	
Loss of assets/wealth	$-0.0374^{ m b}$	(-2.88)	
Both	-0.0437°	(-3.59)	
None	0.0127	(0.70)	
Central Coast	-0.192°	(-20.58)	
South Coast	0.0212^{a}	(2.05)	
North Mountain Range	-0.376°	(-31.63)	
Central Mountain Range	-0.390°	(-47.53)	
South Mountain Range	-0.0990°	(-12.14)	
Jungle	-0.210°	(-27.77)	
Lima metropolitan area	-0.249°	(-30.50)	
Constant	-2.102°	(-101.87)	
Mills (lambda)	-0.135 [°]	(-5.97)	
Rho	-0.10507		
Sigma	1.2845615		

Table 3

CREDIT MARKET PARTICIPATION EQUATION

Notes: corresponds to probit model estimates described in Equation 7. z statistic in parenthesis. ^a p < 0.05, ^b p < 0.01 and ^c p < 0.01. Source: ENAHO and RCC, 2008-2014. The low elasticity of demand for credit could also be related to Peru's banking structure, which is characterized by being concentrated in a small number of financial entities (Céspedes and Orrego, 2014; and Jopen, 2013). In this regard, some literature suggests that the elasticity of demand for credit with respect to the interest rate should be high in a competitive market.

CRED	IT DEMAN	D ESTIMAT	ES	
		Dependent ve	ariable: log(cre	dit)
	мсо (1)	мсо (2)	Heckman (3)	Heckman (4)
Interest rate (log)	$\overline{-0.362^{a}}$ (0.0066)	-0.295^{a} (0.0057)	$-0.307^{a} \\ (0.0034)$	-0.294^{a} (0.0035)
Demand characteristics				
Gender		~	✓	~
Age		~	✓	~
Age ²		~	✓	~
Education		~	✓	~
Parentage		~	✓	~
Marital status		~	✓	~
Economic sector		~	✓	~
Geographic region		~	✓	~
Supply characteristics				
Type of credit		~	✓	\checkmark
Type of currency		~	✓	~
Type of bank		~	✓	~
Year		~	✓	~
Type of bank×year×currency		~		~
Mills (lambda)			-0.233^{a}	-0.135 ^a
\mathbb{R}^2	0.05	0.66		
Number of observations	84,394	78,889	543,358	543,353
Prob > F	0.0	0.0		
Prob > χ^2			0.0	0.0
Note: standard error in parenthe Source: ENAHO and RCC, 2008-20	esis.)14.			

Table 4

The elasticity of demand for credit could be an indicator of competition in Peru's market. The reasoning behind this lies in the capacity banks have to pass the shocks they face on to households by changing interest rates, and this capability depends on the elasticity of the interest rate. Under such interpretation, financial institutions maintain high rates because lowering them does not significantly increase the demand for credit.

The recent strong economic growth experienced by Peru could be important in explaining the low elasticity of credit with respect to the interest rate. The significant expansion of household credit seen between 2001 and 2014 mostly reflects the aggressive placement policies of financial institutions in an environment of higher employment and wages. The growth of placements basically takes place on the extension side, i.e., the number of new loans granted rather than average loan size. Such facts are consistent with the greater financial inclusion experienced by the economy in those years, with a larger amount of institutions offering credit such as rural savings banks and cooperatives, among others, that enabled previously unattended sectors to participate. These new loans are potentially riskier and reflect the participation of high-risk individuals with credit profiles that accept the high interest rates offered by the banks. In this context, banks have few incentives to lower (the high) interest rates on their products because this would not substantially increase the demand for credit given the corresponding low elasticity.

Another factor considered as possibly explaining the low elasticity is the greater financial inclusion being seen in the Peruvian economy (arriving at economic sectors that did not previously have access to credit). This phenomenon has been observed, for instance, in the marginalized areas of Lima, and generally in different regions of Peru where there were practically no banks in previous decades. However, the scenario has changed considerably and nowadays the network of agencies and agents offering loans (rural savings banks, municipal savings banks, savings, and credit cooperatives, major bank branches, etc.) has widened along with access to credit (new loans), which has followed a similar path.

5.3 Heterogeneity of the Demand for Credit

The demand for credit is heterogeneous and depends on the credit market supply and demand side characteristics. Moreover, the literature has found that heterogeneity is also present in the elasticity of demand for credit with respect to the interest rate. In this section, we consider several levels of heterogeneity basically related to the institutional order of Peru's economy that could sustain the heterogeneity of the transmission of interest rate shocks to household credit. This heterogeneity takes place according to the type of currency in which loans are granted, according to the region where they are granted and type of loan. We also consider the possibility that the elasticity of demand for credit changes over time.

Formally, Equation 8 is modified by including the effects of interaction between the interest rate and a set of artificial variables (D_{ijt}^l) that take a value of one at each level of heterogeneity considered, the resulting equation is written as follows:

9
$$b_{ijt}^n = \alpha + \sum_{l=1}^Q \beta_r^l D_{ijt}^l \times R_{ijt} + \beta x_{ijt} + \theta z_{jt} + \delta T_t + \gamma \lambda_{ijt} + v_{ijt},$$

where Q levels of heterogeneity are considered. In this specification, indexes associated with interaction (β_r^l) are the elasticities for each level of heterogeneity.

5.3.1 Heterogeneity by Type of Loan

The demand for credit is particularly heterogeneous according to the type of credit. The data allow for disaggregating up to three types of credit: consumer loans, mortgages, and credit to small and micro enterprises. The estimates suggest that the elasticity of the demand for credit differs according to the type of credit; consumer loans being the most elastic with an elasticity of close to -0.40, while mortgage loans are the least elastic.

5.3.2 Dollarization and Credit Demand

The type of currency can play an important role in the transmission of interest rate changes to credit. In this respect, dollarization of household credit in Peru's economy is around 30% and very heterogenous according to different observable categories among individuals such as income, age, and region of residence, among others, as documented in Céspedes (2017). The estimations in this section suggest that the elasticity of demand for credit is heterogeneous according to loan currency, those denominated in foreign currency being more elastic, while those in domestic currency are less so (Figure 6, Panel B).

The higher sensitivity of foreign currency borrowing could respond to a greater exposure of personal loans to interest rate movements. For instance, changes in international rates originating from, among other reasons, adverse global events could have a larger pass-through to households' foreign currency denominated credit, while such effects would be smaller in the case of loans in domestic currency.

However, the estimation methodology that has been implemented in this study might overestimate the sensitivity of interest rates to credit. This could be the case, for instance, of the exchange rate and the registration method used in the RCC by the SBS. In this regard, the RCC expresses loans in domestic currency using, in the case of loans in foreign currency, the official exchange rate at any given day (end-of-month for accounting purposes) for all loans and all financial institutions. Nevertheless, each financial entity uses a specific exchange rate that captures the expected exchange rate devaluation and is included in the interest rate they charge on loans, especially for individuals whose income is in domestic currency and have loans in foreign currency. The suggested overestimation would take place because the interest rate series employed has a lower variance by considering an average exchange rate and



Note: The estimates of β_r^l from equation 9 are showed. There are also showed the confidence intervals. LIC is the lower limit of confidence of the credit demand elasticity, while LSC is the upper limit. Source: ENAHO and RCC, 2008-2014.

a specific date, while interest rate variance would be larger if, ideally, it was possible to include the exchange rate used by each institution on the date loans are paid. Given that this volatility is not controlled in the regressions this would result in a larger elasticity of loans in foreign currency.

5.3.3 Changes in the Demand for Credit

By estimating the elasticity of demand for credit over time we find that this parameter increased in absolute value between 2008 and 2012 and exhibited a downward trend in 2013 and 2014 (Figure 6, Panel C). This result could be evidence that the pass-through of interest rate changes to credit has been related to the credit cycle, recalling that credit expanded at the highest rates between 2008 and 2012, and has slowed since 2012.

5.3.4 Credit Demand and Income

Participation in the credit market depends on the position in the income distribution. In this section, we have also found that the elasticity of the interest rate is lower in high income and better educated households, as well as among those with formal employment. These elasticities are statistically significant as illustrated in panels D, E, and F of Figure 6.

5.4 New and Old Loans

The demand for credit estimated in the previous section encompasses all loans registered in the RCC, while loan size corresponds to their balances. From the viewpoint of monetary policy, the loans that capture the transmission of changes in monetary policy would be new ones. We identify new loans using the credit panel in consecutive periods. To be specific, we distinguish the new loans in each month, identifying individuals with credit who did not have loans or were not registered in the RCC during the immediately preceding month. Note that individuals identified as having new loans might have had some type of credit in the past. We find that the new loans identified maintain a negative correlation with the corresponding interest rates, similar to that shown in Figure 4. Moreover, the new loans sample is small compared to the total sample, and said sample becomes even smaller if the different types of loan and individuals' characteristics are included, meaning the elasticity of demand for credit estimated for the previously mentioned categories (loan type, an individual's age, etc.) would only have high standard errors and be inaccurate with new loans. In light of the aforementioned considerations, the elasticity of credit is estimated with new loans and compared with the elasticities estimated using the full sample.



Note: Credit is on the y-axis and the interest rate on the x-axis. Variables are expressed in logs.

Source: ENAHO and RCC, 2008-2014.

Credit demand for new loans is estimated using the same procedure described in the previous section, i.e., employing a binary variable that identifies new and old loans. As a result of this procedure, we find that the elasticity of demand for new loans is lower in absolute value (-0.17) than the elasticity reported for old loans (-0.30). This gap between the elasticity of new and old loans is similar using different methods (ordinary leas squares, Heckman). Furthermore, the elasticity increases when medium length loans and the oldest ones are included. The rationale of these outcomes lies in the particular characteristics of Peru's credit market, where interest rates on loans change over time even after being stipulated in a contract. Furthermore, there is a secondary market for purchasing debt, and each repo transaction qualifies as a new loan, although it would be difficult to identify them from RCC data. In sum, an important percentage of loans deemed to be old are in fact new ones and, therefore, the elasticity of demand for such loans in the secondary market should be higher.

6. CONCLUSIONS

Demand for credit at the individual level is an equation seldom estimated for an economy. The reason for this is that it is necessary to know credit and interest rates along with credit demand and supply features, but databases with this type of indicators are scarce at the international level. In this paper, we construct a new database that allows for observing the aforementioned variables by merging the National Household Survey with the Credit Registry for the period 2008-2014. The resulting database enabled us to examine 73,000 individual debtors.

Households' demand for credit is estimated using a twostep procedure proposed by Heckman (1979). The first step estimates extensive credit demand and the second intensive credit demand. The results highlight that participation in the credit market is determined by the number of property rentals households have, the remittances they receive, the size of the shocks they face, and their informality status. The latter characteristic is particularly important because it reveals a significant participation of informal individuals in formal bank credit. On this point, it is interesting to delve deeper into the reasons why informal workers participate in the formal credit market.

With respect to intensive demand for credit, it stands out that there is an elasticity of demand of approximately -0.29, figure slightly lower than that reported by the small number of international studies related to this paper.

The elasticity of demand for credit is found to be heterogeneous in the estimation after controlling for the heterogeneity of credit demanders (individuals) and the heterogeneity of credit suppliers (banks). This evidence suggests that fixed effects at the individual and bank level not only impact average credit but also the elasticity of demand for credit. This heterogeneity is found according to loan type, currency denomination (domestic or foreign), individuals' income and level of education, among others.

Finally, it is important to highlight that the findings provide a first look at the heterogeneity of the demand for credit at the individual level in Peru. In general, the results of the study could be useful for assessing the transmission of the effects of the variables determining interest rates on individuals' credit and through the same channel on their consumption and standard of living. In particular, and considering the high and persistent dollarization in Peru's economy, the data shown in this article should be taken into account as arguments for explaining the pass-through of exchange rate shocks to credit at the individual level.






Source: ENAHO and RCC, 2008-2014.



Source: ENAHO and RCC, 2008-2014.





References

- Alessie Rob, Stefan Hochguertel, and Guglielmo Weber (2005), "Consumer Credit: Evidence from Italian Micro Data," *Journal of the European Economic Association*, Vol. 3, No. 1, March, pp. 144-178, <https://www.jstor.org/stable/40004946>.
- Alfageme, Augusta, and Nelson Ramírez-Rondán (2016), Acceso a servicios financieros de los hogares en el Perú, Documento de Trabajo, No. 2016-015, Banco Central de Reserva del Perú.
- Ausubel, Lawrence M. (1991), "The Failure of Competition in the Credit Card Market," *The American Economic Review*, Vol. 81, No. 1, March, pp. 50-81.
- Bank for International Settlements (2006), *Housing Finance in the Global Financial Market*, CGFS Papers, Working Group Report, No. 26, January, 45 pages.
- Barsky, Robert B., F. Thomas Juster, Miles S. Kimball, and Matthew D. Shapiro (1997), "Preference Parameters and Behavioral Heterogeneity: An Experimental Approach in the Health and Retirement Study," *The Quarterly Journal of Economics*, Vol. 112, No. 2, May, pp. 537-579, https://doi.org/10.1162/003355397555280>.
- Beer Christian, Steven Ongena, and Marcel Peter (2010), "Borrowing in Foreign Currency: Austrian Households as Carry Traders," *Journal of Banking & Finance*, Vol. 34, No. 9, pp. 2198-2211.
- Bernanke, Ben S., and Alan S. Blinder (1988), "Credit, Money, and Aggregate Demand," *The American Economic Review*, Vol. 78, No. 2, pp. 435-439, https://www.jstor.org/stable/1818164>.
- Brown Martin, Steven Ongena, and Pinar Yeşin (2011), "Foreign Currency Borrowing by Small Firms in the Transition Economies," *Journal of Financial Intermediation*, Vol. 20, No. 3, July, pp. 285-302, https://doi.org/10.1016/j.jfi.2010.12.001>.
- Cámara, Noelia, Ximena Peña, and David Tuesta (2013), Determinantes de la inclusión financiera en Perú, Working Paper, No.13/31, Economic Research Department, Banco BBVA.
- Céspedes-Reynaga, Nikita, and Fabrizio Orrego (2014), *Competencia de intermediarios financieros en Perú*, Working Paper, No. 2014-010, Banco Central de Reserva del Perú, August.
- Céspedes-Reynaga, Nikita (2017), La heterogeneidad de la dolarización de créditos a nivel de personas, Working Paper, No. 2017-008, Banco Central de Reserva del Perú, September.

- Cowan, Kevin, Erwin Hansen and Luis Óscar Herrera (2005), Currency Mismatches, Balance Sheet Effects and Hedging in Chilean Nonfinancial Corporations, Research Department Working Papers, No. 521, Inter-American Development Bank.
- Fidrmuc, Jarko, Mariya Hake, and Helmut Stix (2013), "Households' Foreign Currency Borrowing in Central and Eastern Europe," *Journal of Banking & Finance*, Vol. 37, No. 6, June, pp. 1880-1897, https://doi.org/10.1016/j.jbankfin.2012.06.018>.
- Gross, David B., and Nicholas S. Souleles (2002), "Do Liquidity Constraints and Interest Rates Matter for Consumer Behavior? Evidence from Credit Card Data," *Quarterly Journal of Economics*, Vol. 117, No. 1, pp. 149-185, <DOI: 10.1162/003355302753399472>.
- Hall, Robert (1978), "Stochastic Implications of the Life-cycle Permanent Income Hypothesis: Theory and Evidence," *Journal of Political Economy*, Vol. 86, No. 6, December, pp. 971-987, http://dx.doi.org/10.1086/260724>.
- Heckman, James J. (1979), "Sample Selection Bias as a Specification Error," *Econometrica*, Vol. 47, No. 1, January, pp. 153-161, <DOI: 10.2307/1912352>.
- International Monetary Fund (2012), "Dealing with Household Debt," *World Economic Outlook*, Chapter 3, pp. 89-124.
- International Monetary Fund (2016), "Household Credit Growth in Emerging Market Countries," *Global Financial Stability Report*, Chapter 2, pp. 46-73, September.
- Jopen, G. (2013), "Poder de mercado, intermediación financiera y banca: un enfoque de organización industrial," *Economía*, Vol. 36, No. 71, Fondo Editorial de la Pontificia Universidad Católica del Perú, pp. 75-106.
- Kimball, Miles S., Claudia R. Sahm, and Matthew D. Shapiro (2008), "Imputing Risk Tolerance from Survey Responses," *Journal of the American Statistical Association*, Vol. 103, No. 483, pp. 1028-1038, https://doi.org/10.1198/016214508000000139>.
- Kimball, Miles S., Claudia R. Sahm, and Matthew D. Shapiro (2009),
 "Risk Preferences in the PSID: Individual Imputations and Family Covariation," *The American Economic Review*, Vol. 99, No. 2, May, pp. 363-368, <DOI:10.1257/aer.99.2.363>.
- Pellényi, Gábor, and Péter Bilek (2009), *Foreign Currency Borrowing: The Case of Hungary*, Working Paper FINESS D.5.4, German Institute for Economic Research, Berlin, November.

Inventory Adjustments to Demand Shocks under Flexible Specifications

Carlos R. Barrera Chaupis

Abstract

The relations among growth rates in GDP and four aggregate demand components associated with inventory management are approximated by a neural VAR model with t-Student disturbances and an ARCH covariance matrix. The estimation sample corresponds to Peru's market-based growth experience (1993Q1-2010Q1). The main finding is that a positive shock to private demand growth will contemporaneously generate a more than proportional increase in production growth. This amplifier impact effect is consistent with the cycle of inventories and the average incidence of the inventory investment growth inside the production growth during the last four recessions.

Keywords: time series models, neural networks, inventories, production smoothing, business fluctuations.

JEL classification: C32, C45, E22, E23, E32.

C. R. Barrera Chaupis <carlos.barrera@bcrp.gob.pe>, Research Division, Banco Central de Reserva del Perú. The author would like to thank Nikita Céspedes and Marco Vega for their comments and suggestions.

1. INTRODUCTION

Since the 1940s the *inventory cycles* of Metzler (1941) have been recognized as a predominant characteristic of economic cycles (Blinder and Maccini, 1991). Their importance has been confirmed as lengthier expansions observed recently in the world economy came to an end with the United States (US) financial crisis (2007-2008) and the temporary collapse of international trade (see Alessandria et al., 2010), which also impacted the demand associated to commodity exports.

The sustained expansions seen since the start of the 1990s, particularly in the developed world, fostered growth in emerging economies due to greater commercial and financial openness. In general terms, this expansion was characterized by 1) a decreasing and eventually low world inflation, an environment that had not been observed since the 1960s; and 2) a reduction in the GDP growth volatility.

The literature on inventories have justifiably become relevant because they provide an explanation for the observed phenomenon of sustained and stable expansions, also known as the *Great Moderation*. According to this *explanatory history*, the continuous development of information technologies, communications, and sales forecasting techniques have fostered improvements in inventory management accompanied by a consequent reduction in the volatility of *inventory variation* (in other words, inventory investment or flow of inventory balances), which explains the reduction in the volatility of United States GDP and its corresponding growth rates (see, for instance, Kahn et al., 2002).

The prolonged economic expansion in US that started in 1991 was followed by a very short recession during the first few months of 2001, in which massive inventory liquidation was in contrast to the smooth movements observed previously, even before the prolonged expansion. For Kahn and McConnell (2002), this massive liquidation did not show that improvements in inventory management had been tenuous, but that firms had predicted falling sales long before they appeared, wich allow them to drastically reduce their inventories and thereby avoid excessive accumulation. Predicting the fall in sales allowed them to reduce their production in advance and then ration inventories according to demand, maintaining inventory-to-sales-ratios close to desired values.¹

To characterize the stabilization observed in the US durable goods sector, Kahn (2008) points to two key facts: 1) a significant reduction in the volatility of output growth and 2) a more modest reduction in the volatility of sales growth. To characterize the stabilization of aggregate output in Australia, Simon (2001) also highlights two key facts: 1) changes in the inventories cycle, and 2) declines in underlying output volatility. By dismissing an increase in structural stability (the previously mentioned explanatory history), Simon (2001) explains the second key fact through a decline in the volatility of *productivity shocks* (supply shocks) that hit the economy, but leaves the source of such shocks as an open question.² In any case, the sectoral decomposition (by productive sectors) provides an explanation for the Great Moderation, which is complementary to that based on a decomposition of GDP growth by type of expenditure,³ and both lines of work emphasize the unconditional variance of GDP.

¹ The objectives of predicting sales and remaining close to the desired ratio imply that movements in inventories amplify business cycle fluctuations. Despite this, the *average* contribution of inventories to the volatility of GDP growth in the USA (its *average incidence*) is smaller. The model in the following section encompasses those contributions.

² Later in this paper it will be seen that Simon's underlying output (2001) is actually an *aggregate demand excluding inventories*, meaning it is inappropriate to decompose it with an output function in order to estimate *productivity shocks*.

³ Eggers and Ioannides (2006) point to a decline in the importance within GDP of relatively more volatile sectors (agriculture and manufacturing) in favor of other less volatile ones (financial and services) as *the* explanation for the Great Moderation. Davis

The recent financial crisis in USA (2007-2008) affected demand associated with exports as part of the inventory cycle in the economic cycle, generating an unparalleled collapse and recovery of international trade. The literature has highlighted the role of private domestic demand and the inventories mechanism (Alessandria et al., 2010), as well as the private domestic demand of a country's main trading partner (Eaton et al., 2011), the latter being the most important determinant of external demand for exports.

In this context, although towards the start of 2010 it was too early to outline a general description of the turning point stemming from the US crisis in 2007-2008, the experience of Peru up until then might be illustrative of the inventory cycle in an emerging economy, despite having only a few business cycles, i.e., recorded under market conditions (Barrera, 2009). Moreover, economic relations between inventory growth during GDP growth shocks and three aggregate demand components (public domestic demand, private domestic demand and external demand for exports) stand out as being the least studied in Peru.

Table 1 quantifies the importance of inventory change as a percentage of GDP variation in the four recessions observed in Peru prior to that generated as a consequence of the US crisis in 2007-2008.⁴

The average of these coefficients is 230.4%, with a variation range of [100.9, 466.6]. As a reference, the average for the USA is 87%, with a variation range of [2, 232] according to the calculations made by Blinder and Maccini (1991) with the eight recessions recorded during 1948-1982. Firstly, this confirms that shifts in inventory investment have contributed by amplifying the recessive phases of the Peruvian economy since the start of the 1990s, especially the most recent one. Secondly,

and Kahn (2008) seek a more complete explanatory theory, with several interacting factors.

⁴ The units employed are peak to trough changes in the *four-quarter* average percentage variations (percentage variations in four-quarter moving averages expressed in millions of 1994 nuevos soles).

Table 1

	Change in fo average percent (peak-tr	our-quarter tage variations rough)	Inventory
Reference variable: nonprimary GDP (peak-trough dates)	Real GDP (1)	Inventory investment (2)	investment to real GDP (2/1)
Sample: 1992M12-2007M12 ^a			
(1) 1995M7-1996M10	-2.4	-2.4	100.9
(2) 1997M12-1999M8	-1.7	-3.5	212.1
(3) 2000M8-2001M8	-1.8	-2.6	141.9
(4) 2003M3-2004M6	-0.7	-3.3	466.6
Average (1-4)	-1.6	-2.9	230.4
Memo: 2008Q2-2009Q2	-3.3	-12.2	373.9

AVERAGE INCIDENCE OF INVENTORY CHANGES

Inventory investment and recessions since 1990

^a Four-quarter average percentage variations were the units employed to identify business cycles in Peru's economy using the Bry-Boschan approach (see Barrera, 2009).

Source: Author's calculations using data at levels from the Banco Central de Reserva del Perú.

and in contrast to the Great Moderation observed in US business cycles, Peru has seen a phenomenon of demoderation, at least since the third recession recorded.⁵

This paper aims to explain why the *demoderation* phenomenon takes place in Peru. To that end we quantitatively approximate the dynamic relations (potentially asymmetric) between inventory growth, GDP growth and three aggregate demand components (domestic public demand, domestic private demand and, especially, external demand for exports) during

Note that the coefficients for USA are calculated using peak to trough flows in billions of 1982 dollars, making them indirectly comparable with coefficients for Peru.

Peru's market-based growth experience between the first quarter of 1993 and the first quarter of 2010 (1993Q1-2010Q1).

The second section describes the data that will be used to obtain empirical results. These data allow us to outline what we tentatively and temporarily call the *stylized facts* regarding the use of inventories. In principle, inventories serve to buffer the effect of demand shocks on manufacturing operations, although they can also be used for *other objectives* that would explain the demoderation phenomenon in Peru. The third section presents a conceptual framework with respect to production and inventory decisions to provide a qualitative explanation for the demoderation phenomenon. The fourth section proposes a flexible nonstructural model to approximate the dynamic asymmetric relations among GDP and inventories and three aggregate demand sources, as well as a structural model to decompose the covariance matrices of the last period in the sample (final period T=2010Q1). The fifth section describes the results in terms of conditional covariance and impulse responses in an attempt to provide an explanation for the demoderation phenomenon in Peru. The sixth section gives the conclusions.

2. DATA AND STYLIZED FACTS: AGGREGATE FLOW OF INVENTORIES IN PERU

Data used in this study are taken from the Banco Central de Reserva del Perú and are available on its website under the title "Economic Statistics" at the following links: All Sets, Economic Activity and GDP expenditure, at <https://estadisticas.bcrp. gob.pe/estadisticas/series/trimestrales/pbi-gasto>. Figures are originally expressed in real 2007 soles.

2.1 Aggregate Data of Inventory, Production, and Demand

Aggregate production and inventories of firms in an economy will obviously respond to different types of demand shocks.

Thus, aggregate demand excluding inventory investment (*AgDem*) can be decomposed into:

- 1) real exports, goods and nonfinancial services (XDem);
- 2) public sector: real consumption and investment, goods and nonfinancial services (*PuDem*); and
- *3)* private sector: real consumption and investment, goods and nonfinancial services (*PrDem*).

Figure 1 illustrates the four-quarter average percentage variations of those three components, and this data transformation will be used throughout the study.⁶ To represent the scenario it was not sufficient to use the variance of the aggregate AgDem: fluctuations in PuDem were aimed at offsetting those in PrDem on several occasions (anticyclical policies) since the start of the 1990s (with a weak quantitative impact though), and since 1996 aimed to offset short term fluctuation in XDem (partially and at their discretion). Only since 2001, as comprehensive financial constraints on the public sector imposed during the economic stabilization were lifted, the frequency of these more focused countercyclical policies increased. These constraints consisted of continuous fiscal efforts to build public revenues to enable more effective medium-term anticyclical policies, which fostered a larger quantitative impact of fluctuations in PuDem during the sharp fluctuation in *XDem* caused by the crisis in the USA in 2007-2008.

It is also important to consider two other endogenous variables: *1*) real calibrated balance of inventories (*BInv*), and *2*) real gross domestic product (GDP).

⁶ One reason for not following the rules established in the literature for real cycles is due to the fact that data at levels contains a significant measurement error component, while these variations have a very low signal-noise ratio. Thus, Section 3 only provides a qualitative explanation that allows for structurally interpreting the empirical results in Section 5.



del Perú.

Figure 2 shows the same type of variations for these two variables, together with those of AgDem. It can be seen that AgDem and GDP grow at very similar rates. Meanwhile, the growth of *BInv* remained relatively close to aggregate demand since the reversal of the period of inventory overaccumulation at the end of 1994⁷ and until the end of 1998. Later, three overaccumulations of growing magnitude are observed, the first ending

⁷ This episode of overaccumulation reflected, at first, the recovery of production recorded by the success of the stabilization program (inflation decreased drastically, although still at double-digit levels), as well as the optimistic outlook for the economy before the end of the internal war, in the second half of 1992.





Source: author's calculations, using data at levels from the Banco Central de Reserva del Perú.

at the peak of 2000Q2; the second at that of 2003Q2; and the third at that of 2008Q1.⁸

Although only aggregate data is available for inventory investment, *DInv*, the increasing amplitude of *BInv* growth cycles might be explained by the growing participation of goods-inprocess inventories in the whole *DInv*, particularly in traditional export sectors.

2.2 Stylized Facts for Aggregate Inventory Flows

The stylized facts on the relation between inventory investment, sales and production in Peru are presented similarly to

⁸ The third reversal reaches negative rates of around 12% between 2009Q4 and 2010Q1.

in the literature on US inventories. We attempt to explain two stylized facts: 1) why is production more or less volatile than sales? and 2) why are inventory investment and sales not negatively correlated?⁹

One approach to these stylized facts comes from the unconditional sample moments of *four-quarter average percentage variations*, with quarterly frequency, of different GDP components by type of expenditure (one subaggregate of which is aggregate demand excluding inventory investment, *AgDem*). Table 2 presents the mean and standard deviation of these changes, as well as their correlations with inventory investment variation (*DInv*) and calibrated inventory stocks (*BInv*)¹⁰ for two sample subperiods: before and during the period following the international financial crisis arising from the US crisis in 2007-2008.

In terms of standard deviations, production in Peru is *less* variable than sales (demand) for all *AgDem* components except private consumption (in both subperiods). Might there be incentives to use inventories as a buffer against positive demand shocks and maintain smooth production growth?

Given the extreme values in the means and standard deviations of *DInv* variation, the changes of calibrated inventory stocks, *BInv*, is a more stable indicator. This is confirmed in its correlations with the variance of all expenditure components.

Correlations with *DInv* variation reveal that variance in inventory investment and sales (demand) are positively correlated

⁹ A third stylized fact emerges from recent improvements in the quality of inventory investment statistics for developed countries: The most volatile components of inventory investment are not finished goods inventories of the manufacturing sector, but rather its commodity inventories, as well as retail trade inventories (see Blinder and Maccini, 1991).

¹⁰ High levels of volatility in inventory investment growth rates in Table 2 justify such calibration (see Annex A) and explain emphasis on the relations between growth rates of aggregate production, aggregate demand and a *calibrated* sequence of inventory stocks.

Table 2

GDP BY TYPE OF EXPENDITURE (GOODS AND SERVICES)

Four-quarter average percentage variations

		Mean	(<i>W</i>) 1		•	Standard de	viation (.	S)	-	Corr with I)(Inv) (R	(1.	0	Jorr with B	(Inv) (R.	(2
	19.	95-2007	20	08-2010	199.	5-2007	2008	8-2010	199.	5-2007	200,	8-2010	1995	-2007	2005	-2010
	m()	%m(GDP)	m()	% m(GDP)	s()	%s(GDP)	s()	%s(GDP)	r()	% n(GDP)	r()	%r(GDP)	r()	%r(GDP)	r()	% r(GDP)
I. Global demand (1+2)	5.0	108.2	6.1	96.7	4.3	124.2	6.4	156.6	0.11	120.4	-0.02	-124.5	0.53	111.5	0.97	100.9
Ib. Global demand (1b+2)	5.0	107.3	7.3	117.1	4.1	118.4	5.3	128.7	0.11	120.8	-0.11	-613.8	0.50	104.5	0.98	101.9
1. Domestic demand with $D(\ln v)$	4.5	97.5	6.4	102.3	4.9	144.3	6.6	161.6	0.14	154.9	-0.02	-103.9	0.58	121.5	76.0	100.8
1b. Domestic demand without $D(Inv)$	4.5	96.2	7.9	126.8	4.7	136.0	5.3	128.8	0.14	157.9	-0.13	-686.9	0.55	116.0	0.98	101.6
a. Private consumption	4.1	87.5	6.2	99.4	3.0	88.3	2.8	69.3	0.08	94.4	-0.15	-836.6	0.58	121.8	0.99	102.7
b. Public consumption	4.9	104.6	7.7	123.1	5.0	146.5	5.4	130.8	0.13	147.8	0.27	1,437.7	0.11	23.6	-0.95	-98.8
c. Gross domestic investment	6.3	136.2	7.6	121.0	13.0	380.1	20.5	500.8	0.17	186.9	0.00	20.7	0.57	120.6	0.97	100.3
Gross fixed investment (GFI)	6.0	129.5	13.8	221.0	11.8	344.1	15.9	388.8	0.18	201.7	-0.12	-638.0	0.52	109.5	0.98	101.6

Table 2 (cont.)

		Mean	(W)		-	Standard de	viation (S)	-	Corr with I	H) (unl)C	(1	-	Corr with S	(Inv) (R2	
	199	ə5-2007	20	08-2010	199.	5-2007	2008	8-2010	199.	5-2007	200	8-2010	,66 I	5-2007	2008	-2010
	m()	% m(GDP)	m()	% m(GDP)	s()	%s(GDP)	<i>s()</i>	%s(GDP)	r()	$\% \eta(GDP)$	r()	% r(GDP)	r()	%or(GDP)	r()	% r(GDP)
- Private	7.3	156.4	10.5	166.9	13.4	391.6	17.7	431.6	0.16	183.8	-0.13	-678.1	0.54	113.2	0.98	102.1
– Public	1.6	35.4	32.5	519.5	11.6	338.5	9.2	225.1	0.15	173.4	-0.06	-340.0	0.15	31.0	0.77	79.5
Stocks variance (D(Inv))	-83.3	-1,792.2	0.8	12.9	583.3	17,036.7	319.7	7,804.7	1.00	1, 122.8	1.00	5,408.0	0.16	34.5	-0.23	-23.5
Memo: inventory stocks (S(Inv))	2.7	57.7	6.1	96.9	7.6	220.6	24.4	595.5	0.16	184.0	-0.23	-1227.3	1.00	210.5	1.00	103.8
2. Exports	8.4	181.2	4.3	68.4	4.3	125.4	5.4	132.6	-0.13	-142.9	-0.05	-258.9	-0.32	-67.9	0.97	100.8
II. Global supply (3+4)	5.0	108.2	6.1	96.7	4.3	124.2	6.4	156.6	0.11	120.4	-0.02	-124.5	0.53	111.5	0.97	100.9
3. GDP	4.7	100.0	6.3	100.0	3.4	100.0	4.1	100.0	0.09	100.0	-0.02	-100.0	0.48	100.0	0.96	100.0
- Goods	5.0	107.7	5.2	83.7	3.8	110.5	5.1	124.0	0.03	34.1	-0.01	-58.0	0.37	77.2	0.96	99.9
- Services	4.5	96.0	6.8	108.9	3.4	98.8	3.6	86.9	0.12	136.0	-0.03	-136.1	0.52	109.5	0.96	100.1
4. Imports	7.4	159.6	5.6	89.5	10.3	299.7	17.0	415.8	0.13	145.3	-0.03	-163.1	0.55	116.4	0.98	101.4
Source: Author's	s calculi	ations using	g data	at levels fr	om the]	Banco Cei	ntral de	Reserva d	lel Perí	í.						

in the period 1995-2007 for all components of AgDem except exports. In period 2008-2010, they are negatively correlated, except public consumption. The size of all the correlations with *DInv* variation is close to zero due to the large proportion of noise present in *DInv*.

Correlations with the variation of *BInv* are more informative: changes in calibrated inventory stocks and sales (demand) are positively correlated in period 1995-2007 for all components of *AgDem* except exports. They are also positively correlated in period 2008-2010 except for public consumption. The magnitudes of this second group of correlations take values far from zero due to a clearer sign in calibrated stocks *BInv* (a small proportion of noise).

Why are variations of *BInv* and that of demand not negatively correlated? Could there be additional incentives for accumulating inventories at a *higher* rate than the minimum needed to cover positive shocks in demand growth and thereby slow or stabilize production growth?

If firms' main incentive for holding inventories is to meet positive shocks in demand growth and thereby smooth the evolution of production in order to leverage complementary opportunities such as, for instance, low input prices, firms are said to be producing to build stocks. In this case, changes in BInv allow them to control supply in response to demand fluctuations. Nevertheless, successive periods with higher than expected demand growth rates lead to increased production to meet part of the unanticipated demand and even achieve additional growth in BInv. This extra incentive for larger BInv growth stems from a need to increase a nonfinancial asset that offsets higher short-term borrowing incurred to cover production when demand is growing just in case this demand growth reverts (successive periods with lower expectations regarding demand growth could have symmetric effects). Hence, BInv and production operate in a coordinated manner, but with different periods, where BInv can be more than just the main instrument for offsetting demand shocks over the short-term.

Finally, although the stylized facts favor these hypotheses, it is worth questioning their suitability. Should we consider the description of those *unconditional* moments as a correct description of the stylized facts for the relations between aggregate demand growth, on the one hand, and growth in *BInv* and GDP, on the other? According to the variance decomposition theorem, the conditional variance of an available data set is less than unconditional variance. A more general theory sets forth that conditional covariance is different to unconditional covariance (which is also valid for the correlations). Therefore, the unconditional moments can only provide a preliminary description. In this regard, this paper aims to determine whether the conditional moments of the data provide evidence for the presence of the demoderation phenomenon in Peru.

3. GENERAL THEORETICAL MODEL WITH HETEROSCEDASTICITY

Sensier (2003) presents a model that encompasses those of Blanchard (1983), Blinder (1986), Eichenbaum (1989), Kahn (1987), and Ramey (1991) based on the model of Callen et al. (1990) and Cuthbertson and Gasparro (1993). Being I_t a vector of M levels of inventories held by a representative firm by type of good k, for instance, if M=3, k=1 (finished goods); k=2(work-in-progress) and k=3 (raw materials), denominated in units of some finished consumer good that serves as a numeral. Moreover, the vector of functions for its corresponding desired levels is defined as

1
$$I_t^* = I^* \left(\underbrace{S_t, z_t^I, h_t^S, r_t^H}_{+, -, -, +, -, -} \right),$$

where S_t is the vector of sales in period t^{II} of M types of goods (to the market and to the firm's transfer pricing area), z_t^I is a

¹¹ It is feasible to interpret this function for desired inventory levels in period *t* as dependent on sales in period *t*, whatever this level

vector of technological change factors in period t for inventory control procedures for M types of goods, r_l^H is the financial-tax benefit of holding inventories as an asset¹² in period tand h_l^S is a vector of M standard deviations in period t of the prediction error (one period ahead) of each component of vector of sales S_t , conditional to all data available up to current period t. The signs under each variable suggest the direction of dependence in comparative statics (Callen et al., 1990, and Cuthbertson and Gasparro, 1993). The costs or losses incurred for moving away from desired levels are defined as the function

$$C_t^A = C^A (I_t - I_t^*),$$

that has been named *accelerator* in the literature.¹³ The physical cost of holding inventories, which includes renting warehouse space, maintaining a suitable environment for preserving the qualities of the goods (for instance, refrigeration), transport equipment upkeep and man-hours for operating it, among others, is defined as the vector of functions

$$C_t^m = C^m(I_t, \delta),$$

where δ is the vector with M rates of depreciation (maximum effective decrease allowed) of each good k held in the firm's inventories (for instance, δ^f is the component corresponding

2

3

may be (including a predicted or expected level and elaborated with data available in any previous period t-s, where s > 0). The literature has typically considered it as dependent on expected sales for period t (see, for instance, Sensier, 2003; Cuthbertson and Gasparro, 1993; Blinder and Maccini, 1991).

¹² See Sensier (1993). Callen et al. (1990) treat it as a unitary financial cost for holding inventories.

¹³ For instance, the sum of quadratic terms corresponding to each good *k*, multiplying each one by a coefficient $b_k/2$.

to finished goods).¹⁴ The cost of producing finished goods is defined as the function

where v_t is the marginal cost term that varies over time¹⁵ and P_t is the level of production.¹⁶ To simplify, from now on we assume that a firm only holds inventories of finished goods $(I_t = I_t^f)$, meaning all the vectors mentioned previously in this section are scalar.

The *inventories restriction* establishes a relation between production, sales, and finished goods inventory flows

$$P_t = S_t + \Delta I_t^f,$$

- ¹⁴ In Blinder (1982, 1986a, 1986b) and Sensier (2003), C_t^m is a quadratic function in I_t^f without a constant and with coefficient $e_2/2$ for the quadratic term. Eichenbaum (1989) use a quadratic function, but with coefficient e_{1t} for the linear term (that varies over time). Here the physical holding costs depend on depreciation rates (that can vary over time).
- ¹⁵ In Eichenbaum (1989) it is a stochastic shock to the marginal cost of producing P_t in order for the model to encompass the motive for production cost smoothing of Blanchard (1983) and West (1990), such as for instance a shock to relative factor prices. In general, it can be any variable that affects a firm's intertemporal production decisions, such as financial or liquidity position (Cuthbertson and Gasparro, 1993; Sensier, 2003) or a one step ahead sales forecast error (Sensier, 2003 also uses production forecasts in her estimates).
- ¹⁶ In Blanchard (1983), Eichenbaum (1989), Sensier (2003) and West (1990), C_t^P is a quadratic function in P_t without a constant and with coefficients v_t for the linear terms and a/2 for the quadratic term. If *a* is positive, marginal production costs are rising and the model encompasses the production smoothing motive of Blinder (1986a); if *a* is negative, the model includes the case considered by Ramey (1991).

5

that is normally used to obtain total flows (billed and unbilled) of finished goods sales. The historic sequence of inventory flows can be used to obtain inventory stocks, for instance, of finished goods,

$$G \qquad I_t^f = \left(1 - \delta^f\right) I_{t-1}^f + I_t^f,$$

i.e., an equation of perpetual inventories where δ^f is the depreciation rate of finished goods inventories.

Under these assumptions, the firm maximizes the conditional expectation of the present value of real benefit at time t, Π_t , with respect to the decision variable sequence, $\{I_{t+j}^f\}_{j=0}^{\infty}$, given predetermined variables I_{t+j-1}^f and sequences of the best forecasts of $\{S_{t+j}, z_{t+j}^I, r_{t+j}, h_{t+j}^S, v_{t+j}\}_{j=0}^{\infty}$ for the whole period considered in the present value, $[t, t+1, \dots, \infty)$.

$$\begin{aligned} \mathbf{T} & E_t \left[\Pi_t \right] \! = \! \left[\Sigma_{J=0}^{\infty} \beta^j \left\{ S_{t+j} - C^A \left(I_{t+j}^f - I^* \left(S_{t+j}, z_{t+j}^I, r_{t+j}, h_{t+j}^S \right) \right) - C^m \left(I_{t+j}^f, \delta^f \right) \! - C^P \left(c_{t+j}, S_{t+j} + \Delta I_{t+j}^f \right) \right\} \right] \! , \end{aligned}$$

where β is the discount factor and $E_t[.] \equiv E_t[.|\Omega_t]$ is the conditional expectation operator for full relevant data set Ω_t available for a firm at time *t* when it is going to determine the optimal sequence $\{\tilde{I}_{t+j}^f\}_{j=0}^\infty$. We have assumed that the sales income function is concave and that cost functions are all convex, meaning the first order condition (Euler equation) is the necessary and sufficient condition for an optimal.¹⁷

Eichenbaum (1989) solves this problem for a set of specific parameters where the first order condition gives a necessary

¹⁷ This formula should include benefits stemming from all production and financial operations conducted by a firm, or at least those associated to different types of inventories. For in-

and sufficient condition. After appropriate algebraic manipulation, he obtains the condition for the optimal plan of inventory stocks $\left\{\tilde{I}_{l+j}^{f}\right\}_{j=0}^{\infty}$, according to which:

- *I*) I_t^f depends positively on expected future sales $\{S_{t+j}^f\}_{j=0}^{\infty}$: Inventories are held for production smoothing;
- 2) I_t^f depends negatively on current sales S_t^f : Because marginal production costs are increasing, there is a margin above which firms would rather cover their sales with inventories than increase production;
- 3) I_t^f depends negatively on a current stochastic shock to marginal production costs v_t : When marginal production costs are high, firms would rather meet current sales with current inventories than increase production;
- 4) I_t^f depends positively on future shocks to marginal productions costs $\{v_{t+j}\}_{j=1}^{\infty}$: Firms would rather build up inventories with current production when current marginal production costs are low compared to future ones, and therefore meet future sales out of those stocks of inventories instead of future production; and
- 5) I_t^f depends negatively on the linear coefficients (present and future) of inventory holding costs, $\{e_{1t+j}\}_{j=0}^{\infty}$ (see note 14).

Formulating the problem of a representative firm assumes that the variables are stationary. Given that production and aggregate sales are not stationary, the problem must be rewritten

stance, costs associated to the factors of production for work in progress separated from finished goods, net benefits resulting from production operations for work-in-progress as well as financial operations such as purchases-sales of raw materials (the *inventories restriction* would be modified accordingly). The simple specification in terms of real benefits avoids considering the possibility for accounting part of the financial-speculative activities the corporate sector may perform with different types of inventories it holds. through an appropriate normalization or alternatively by using a two-step approach proposed by Callen et al. (1990): *1*) propose a linear cointegration relationship between the *nonstationary* level of inventories and the determinants of a desired level of inventories; and 2) use the cointegration error sequence to minimize total costs $C_t^T = C_t^m + C_t^P$ for each period as a function of inventory stocks.

The theoretical framework provides a qualitative explanation for the relation between the level of inventories and its determinants, although as mentioned previously, we will only use average percentage variations (*var*%) in the following sections.¹⁸

4. PROPOSED VARNN-ARCH MODELS

A family of dynamic models are immune to heteroskedasticity problems and are appropriate for both the conceptual model of the previous section and most models used in macroeconomics, where the aim is for the conditional means to be correctly calculated despite the presence of outliers and high variance episodes (Hamilton, 2008).

4.1 Conditional Means

First, we describe the models to be estimated for the conditional means. The first model for those moments is a typical linear multivalued function of VAR(K, p) models,

$$y_t = A_0 + A_1 y_{t-1} + \ldots + A_p y_{t-p} + \varepsilon_t = A_0 + \sum_{j=1}^p A_j y_{t-j} + \varepsilon_t,$$
$$\varepsilon_t \mid \Omega_{t-1} \sim N(0, \Sigma_t),$$

8

¹⁸ Another justification for this can be found in the properties of elasticities ε_i of a scalar function that depends on *n* variables, $z_t = z(x_t^1, ..., x_t^n)$, or $var\% z_t = \sum_{i=1}^n (var\% x_t^i) \varepsilon_i$. The property is applicable to any of the functions used under this theoretical framework (including Euler conditions).

where $y'_t \equiv \{y_{1t}, y_{2t}, ..., y_{Kt}\}$ and $\varepsilon'_t \equiv \{\varepsilon_{1t}, \varepsilon_{2t}, ..., \varepsilon_{Kt}\}$ are vectors of K stationary variables, $\Omega_{t-1} \equiv \{y'_{t-1}, y'_{t-2}, ..., y'_{t-p}\}$ is the relevant data set and $\Sigma_t \equiv [\sigma_t^{ij}]$ is matrix $K \times K$ of conditional covariances of period $t(\sigma_t^{ij} = \sigma^{ij} \text{ for VAR}(K, p) \text{ models}).$

The second group of nonlinear VAR models generalizes the model of Equation 8:

$$y_t = A_0 + g(\Omega_{t-1}) + \varepsilon_t \quad \varepsilon_t \mid \Omega_{t-1} \sim N(0, \Sigma_t),$$

where it is usual to postulate a specific nonlinear multivalued function g(.), for instance, choosing (somewhat arbitrarily) the smooth transition function, VSTVAR, or the self-excited threshold function SETVAR (see Granger and Teräsvirta, 1993).

Instead of assuming a priori the knowledge of function g(.), a hypothesis that is taken as a premise in modern macroeconomics, here we employ a more general assumption: the existence of unknown nonlinear patterns in the data. Hence, we propose using flexible dynamic models (neural networks) whose main property is precisely a high capacity for approximating those patterns in the data. In this context, we choose a network architecture named *multilayer perceptron* (MLP).¹⁹ Its dynamic version (VARNN-perceptron or VARMLP) will be used to obtain an approximation (global) of the nonlinear multivalued g(.), the one that best adjusts to nonlinear patterns in the data.²⁰ According to that architecture, this is made possible by combining a finite number of basic structured nonlinear Hfunctions in a multilayer graph,

¹⁹ See Dorffner (1996). This architecture of *artificial neural networks* (ANN) are used in temporal series (also known as *feedforward* ANN; see Kuan and Liu, 1995).

²⁰ A Taylor approximation requires a specific function and an approximation point.

10
$$g(\Omega_{t-1}) \approx \beta_0 + \Sigma_{i=1}^H \beta_i h_i(\Omega_{t-1}) = \Sigma_{i=1}^H \beta_i \Psi_i \Big(\Delta_{0,i} + \Sigma_{j=1}^p \Delta_i(j) y_{t-j} \Big),$$

where *H* units h_i are denoted *hidden units*, each one of which is a multivalued linear function Ψ_i whose components are bounded functions.²¹

4.2 Conditional Covariances

Second, we describe the family of models for the conditional covariance matrices of the model we will finally estimate. This is the family of multivariate ARCH models, of which the most well-known are VECH, BEKK and exponential. The VECH model is the most general,

$$vech(\Sigma_t) = c + \Sigma_{h=1}^p C_h vech(\varepsilon_{t-h}\varepsilon'_{t-h}) + \Sigma_{k=1}^q B_k vech(\Sigma_{t-k})$$

where using a *vech* operator (that stacks elements above and below the square matrix diagonal) gives c as a vector of order $[K(K+1)/2] \times 1$ and $\{C_h\}$, $\{B_k\}$ are matrices of order $[K(K+1)/2] \times [K(K+1)/2]$. As mentioned in Ding and Engle (2001), their generality goes hand in hand with their reduced parsimony and the difficulty of imposing restrictions that ensure a sequence of positively defined matrices $\{\Sigma_t\}$ (except when imposing $\{C_h\}$ and $\{B_k\}$ diagonals).

The BEKK model is a restricted version of the VECH model that generates a sequence of positively defined $\{\Sigma_t\}$ matrices by imposing a quadratic parameter structure,

11

²¹ Shachmurove (2002) mentions that a major advantage of ANNs is their ability to analyze complex patters quickly with a high degree of accuracy and without making assumptions about the distribution of the data. Among the disadvantages are the fact they tend to over-fit data and lack a standard structured method for choosing, developing, training and evaluating an ANN.

where C, $\{D_h\}$ and $\{E_h\}$ are matrices $K \times K$ and only C is a lower triangle. Engle and Kroner (1995) provide the conditions by which a BEKK model encompasses all diagonal VECH models with a sequence of positively defined matrices $\{\Sigma_t\}$ and almost all VECH models with a set of positively defined matrices $\{\Sigma_t\}$. These conditions eliminate all redundant representations (that are observed as equivalent).

The possibility of asymmetries in conditional covariances has been taken into account through two strategies. The first imposes specific restrictions not necessarily substantiated by the data (for instance, those proposed in Ebrahim, 2000; see Annex B in Barrera, 2010) while the second, proposed by Kawakatsu (2006), uses specific unrestricted parameterization, which we will use to adapt the model for this study.

Kawakatsu (2006) proposes a generalization of the asymmetric model of Nelson (1991) to the multivariate case that manages to maintain the generality of the VECH representation through an innovative parametric structure that generates a sequence of positively defined $\{\Sigma_t\}$ matrices without the sensitive simplifications of Ebrahim (2000). Using a VECH representation, Kawakatsu (2006) proposes

$$\begin{aligned} \operatorname{vech}(\log(\Sigma_{t})) - c_{0} &= \Sigma_{h=1}^{p} C_{t}^{*} \varepsilon_{t-h} + \Sigma_{h=1}^{p} C_{t}^{**} \left(|\varepsilon_{t-h}| - E\left\{ |\varepsilon_{t-h}| \right\} \right) \\ &+ \Sigma_{k=1}^{q} B_{k} \left(\operatorname{vech}\left(\log(\Sigma_{t-k}) \right) - c_{0} \right), \end{aligned}$$

where $log(\Sigma_t)$ is the matrix logarithm of Σ_t , $vech(log(\Sigma_t))$ and $c_0 \equiv vech(C)$ are vectors $[K(K+1)/2] \times 1$, C is a symmetrical matrix $K \times K$ and matrices $\{C_h^*\}$, $\{C_h^{**}\}$ and $\{B_k\}$ have dimensions $[K(K+1)/2] \times K$, $[K(K+1)/2] \times K$ and

13

 $[K(K+1)/2] \times [K(K+1)/2]$, respectively. Matrices $\{C_h^{**}\}$ cap-

ture the *leverage effects* in the conditional covariance process.

Using the matrix logarithm transformation of the covariance matrix (symmetrical) means it is not necessary for $log(\Sigma_t)$ to be positively defined (or to impose any condition). Applying the exponential matrix (inverse) operation to that transformed space gives a covariance matrix that is symmetric and therefore positively defined. This allows any dynamic to be specified for this matrix, always generating a positively defined sequence of $\{\Sigma_t\}$ matrices.

If *T* is the number of observations, where $y'_t = \{y_{1t}, y_{2t}, ..., y_{Kt}\}$ is the transposed vector of *K* variables and Θ is the column vector of all the parameters, the normal multivariate conditional density of $y_t | \Omega_{t-1}$ can be written as:

14
$$f\left(y_{t} \mid \Omega_{t-1}; \Theta\right) = (2\pi)^{-\frac{K}{2}} |\Sigma_{t}|^{-\frac{1}{2}} exp\left(-\frac{1}{2} \left(\varepsilon_{t}^{\prime} \Sigma_{t}^{-1} \varepsilon_{t}\right)\right);$$

and log-likelihood function $l_Q = \Sigma_{t=1}^T l_t$, is obtained, where $l_t \equiv log(y_t | \Omega_{t-1}; \Theta)$. For comparison purposes, the contribution of observation *t* to this log-likelihood function is

15
$$l_t = -\frac{1}{2} \Big(K log(2\pi) + log(|\Sigma_t|) + \varepsilon_t' \Sigma_t^{-1} \varepsilon_t \Big).$$

In the case of the exponential matrix model of Kawakatsu (2006), this expression can be written as

16
$$l_t = -\frac{1}{2} \Big(Klog(2\pi) + logm(\Big| e^{logm(\Sigma_t)} \Big| \Big) + \varepsilon'_t \Big(e^{logm(\Sigma_t)} \Big) \varepsilon_t \Big).$$

Using the following exponential matrix and matrix logarithm properties:

1) For all square matrix A, $(e^A)^{-1} = e^{-A}$.

2) For all symmetrical matrix S, $logm(|e^S|) = traza(S)$.

We obtain

$$l_{t} = -\frac{1}{2} \Big(Klog(2\pi) + traza(logm(\Sigma_{t})) + \varepsilon_{t}'(e^{-logm(\Sigma_{t})})\varepsilon_{t} \Big)$$

By adding the exponential matrix of Kawakatsu (2006) to the proposed nonstructural modelling, which includes a multivariate Student's *t* distribution, all the parameters are robust to the presence of atypical observations without imposing specific restrictions not necessarily substantiated by the data. This model is estimated for the case of Peru with 65 quarterly data for the period 1994Q1-2010Q1.²² Al the variables are expressed as *four-quarter average percentage variations*.

Estimation of the dynamic flexible econometric model is feasible, despite computing restrictions, if the over-parameterization problem is addressed. The latter is common in neural network models and can reduce their usefulness for predictive purposes. Annex B describes the *maximum penalized likelihood* method for solving this problem and the associated reduced number of degrees of freedom.

4.3 A Contemporaneous Structure

We proposed a structural model for covariance matrix decomposition for the final period t = T of the nonstructural VARNN-ARCH model estimated (although the following discussion is applicable to the covariance matrix of any period t). Using decomposition *AB*, matrix (*I*–*A*) is triangular and matrix *B* is dimension diagonal k = 5. Ordering of the structural model $y_t = \{XDem_t, PuDem_t, PrDem_t, BInv_t, GDP_t\}$ should be taken into account for interpreting its coefficients: The most

²² The possibility of including the period of high inflation and its subsequent stabilization was rejected due to considerable fluctuations in relative prices. With the lags in conditional means and lags in conditional covariances, the estimated sample of conditional covariances includes 41 observations (2000Q1-2010Q1).

exogeneous shocks correspond to those of the growth rates of $\{XDem_t, PuDem_t, PrDem_t\}$, in response to which follows the compensatory action of the shock to the growth rate of $\{BInv_t\}$ (according to prevailing incentives), all of which finally determines the shock in the growth rate of $\{GDP_t\}$.

Expected values or signs of coefficients a_{ii} in the matrix (*I*-A) come from the theoretical model described in Section 3. We postulate that there are contemporaneous relations among shocks to the three aggregate demand components: It is anticipated that $\{PuDem_t\}$ fulfills some type of compensatory function in response to shocks in $\{PrDem_t\}$ and $\{XDem_t\}$ (*inverse* relations reflected in *positive* coefficients immediately below the main diagonal of the submatrix (1:3,1:3) of (I-A); see Table 3). Moreover, shocks in all three components affect firms' inventory and production decisions. If a firm's only incentive for holding inventories was production smoothing, the contemporaneous relations between $\{BInv_t\}$ and the three aggregate demand components would be inverse and reflected in positive coefficients in the fourth row of (I-A). However, if there are additional incentives for increasing $\{BInv_t\}$, these relations might be *direct* (*negative* coefficients in said row). Furthermore, while production smoothing, $\{GDP_t\}$, would free it from demand shocks (coefficients in the fifth row would be null), additional incentives would generate direct relations between supply shocks²³ and all the rest (negative coefficients in that row).24

²³ As mentioned in Section 3, production shocks encompass marginal costs shocks (for instance, in the relative prices of factors of production) and technology shocks (investments that improve capital assets), but also include shocks to production processes (problems of logistics such as, for instance, cuts in energy supplies for manufacturing or mining, or shortages in inputs such as water for agricultural production, etc.).

²⁴ Section 2 does not mention that growth of $\{AgDem_t\}$ is a weighted average of the growth of the first three components of the vector

		MATRIX (I	[- A)		
Affect a structural		Stri	uctural shock	s of	
shock in:	XDem	PuDem	PrDem	BInv	GDP
XDem	1	0	0	0	0
PuDem	a_{21}	1	0	0	0
PrDem	a ₃₁	a ₃₂	1	0	0
Binv	a ₄₁	a_{42}	a ₄₃	1	0
GDP	a_{51}	a_{52}	a ₅₃	a ₅₄	1

Table 3

5. RESULTS

From an econometric and statistical standpoint, it is worth questioning the relevance of using such general assumptions, performing statistical tests to validate the need for them, either individually or jointly. The answer, however, should consider the need to nest simpler hypotheses within the proposed model, a consideration that has proved hard to find in the literature consulted on the penalized likelihood (see Annex B).

The results of the proposed general observation tool that imposes a minimum number of maintained assumptions (with the additional cost associated with their estimation) are shown below. Another significant product of this tool is the availability of conditional covariances estimates (conditional variances indicate periods of greater uncertainty for each variable in the model).

5.1 Nonstructural VARNN-ARCH Model

Figure 3 shows the conditional variance of four-quarter average percentage variations for each of the three aggregate

of endogenous variables.





demand components, inventories, and production (i.e., units are squared variations).

It can be seen that the conditional variances of *PrDem* and GDP change over time, while that of *BInv*, *PuDem*, and *XDem* appear as pseudo-constants due to the variation range of conditional variances that clearly change over time.²⁵

Conditional variances of *PrDem* and GDP tend to rise contemporaneously, standing out the more recent jumps in uncertainty. Meanwhile, the sequence of conditional variances for GDP tends to be smaller than the sequence corresponding

²⁵ These two wide ranges of variation could reflect the need to separate *quanta* from relative prices inherent to original units (1994 nuevos soles) for including them in larger sized models (and difficult to estimate). In any case, all conditional moments of the model estimated are so with respect to the small number of included variables.

to *PrDem*, which reflects the existence of a degree of production stabilization with respect to *PrDem* that is attributable to inventory management and is more noteworthy in the event of jumps in the uncertainty of *PrDem*. As for pseudo constant conditional variances over time, that of the *PuDem* is greater than that of *BInv*, and that one is in turn larger than that of *XDem*. Given that these pseudo constants tend to be larger than the variances that change over time (*PrDem* and GDP), production stabilization is performed for each of those aggregate demand components.

With respect to the estimated conditional variance sequence for *AgDem*, which has been added to the previous figures, we calculated it based on the conditional covariances submatrix of its three components (*PrDem*, *PuDem*, and *XDem*).

The conditional variance of AgDem confirms the possibility that motivated this study: That it is smaller than the conditional variance of GDP (except in one quarter subsequent to the recent period of maximum uncertainty) and with a relative magnitude of around one to four (during the period of low lower uncertainty). This result is in contrast to results obtained with unconditional variances (see Table 2), explained by the impact of conditional covariances among their three components.

To conclude, aggregate management of inventories leads to production stabilization through mechanisms that are reflected in the conditional covariances of variances in all three components of AgDem (PrDem, PuDem, and XDem). The evolution of all 15 different entries in the conditional covariance matrix (standardized) is presented in Annex C. Two out of the three covariances that intervene in calculating AgDem variance are negative, (PrDem, XDem) and (PuDem, XDem), which contributes to the sequence of the variance of AgDem being closer to abscissa (see Figure 4).

Covariances (*BInv*, *PuDem*) and (*BInv*, *XDem*) are negative, reflecting expected inverse relations when there are no other incentives for holding inventories except GDP smoothing. Covariance (*PrDem*, *BInv*) is the only positive one, reflecting the expected direct relations when there are additional incentives for *BInv* growth. **Figure 4**



CONDITIONAL VARIANCES OF THE AGGREGATE DEMAND COMPONENTS.

5.2 Structural VARNN-ARCH Model: **Contemporaneous Structure**

Table 4 displays the coefficients estimated for the matrices of AB decomposition of the conditional covariance matrix estimated for the last sample period (T=2010Q1). Note that items below the diagonal in (I - A) have the opposite sign to those of the corresponding items of A, while items different to zero in matrix B (its diagonal) are shown as a column vector.

All the parameters estimated in the matrix (I-A) for period Tof the sample are statistically equal to zero, except the parameter that measures the *positive* impact of the *PrDem* structural shock on GDP (-1.207 in the table). Estimates in period T of the sample reveal that the contemporaneous relations between BInv and AgDem components are statistically equal to zero. Therefore, GDP growth smoothing is not the only incentive

		1	1		
	2	h		ρ	4
-	u	~	-	\sim	-

					I-A		
			1	2	3	4	5
		В	XDem	PuDem	PrDem	BInv	GDP
1	XDem	3.671	1.000				
		(1.985)					
2	PuDem	5.361	0.053	1.000			
		(1.679)	(0.123)				
3	PrDem	3.332	0.059	(0.047)	1.000		
		(1.112)	(0.074)	(0.118)			
4	BInv	4.040	0.044	0.172	(0.076)	1.000	
		(0.521)	(0.117)	(0.188)	(0.115)		
5	GDP	1.822	(0.013)	(0.002)	(1.207)	(0.008)	1.000
		(1.335)	(0.016)	(0.026)	(0.016)	(0.033)	

ESTIMATED CONTEMPORANEOUS RELATIONS SpVARNN-ARCH with five variables

for increasing *BInv* in that period, meaning there are possibly *additional incentives* for it. The only parameter statistically different from zero is consistent with the presence of additional incentives, which according to the macroeconomic context of that period means that negative shocks in *PrDem* growth are reflected in decreases in production growth measured in GDP.

On the basis of these contemporaneous relations we obtain the response functions for any variable *i* after a 1% change in any variable *j* (*impulse response functions*), denoted as $FRI[j \rightarrow i]$. Impulse response functions (IRF) were calculated as the difference between two projections that are not based on a stationary state: a projection with the structural shock from period *T*, the last period of the sample, and a projection without this shock (see Koop et al., 1996).
The IRFs do not generally show asymmetries in the sign or magnitude of shocks, although the scale of contemporaneous impacts (responses in period T) dominate the scale of the rest of the sequence (responses in periods T+h, $h \neq 0$). For this reason, IRFs are presented in 2×2 subgraph matrices: IRFs in the first row include contemporaneous impacts, while those in the second row exclude them.²⁶

5.3 Impulse Responses in the PrDem

Figure 5 displays IRFs for estimated increases in *BInv* and GDP after a shock of 1% in *PrDem* growth. This positive structural shock in *PrDem* causes GDP growth to increase at the time of impact, it falls soon after and then continues to decline slowly towards zero. Meanwhile, *BInv* growth increases upon impact, continues increasing very slowly and then falls 10 quarters ahead.

Considering the relative magnitudes, a positive structural shock in *PrDem* growth is initially absorbed by a sharp increase in the GDP growth and a slight increase in *BInv* growth (which is followed by a delayed smaller decrease 10 quarters ahead). This behavior is in disagreement with simple intuitive inventory management, but consistent with *additional incentives* for raising the growth of *BInv*, such as lags in the adjustment of the aggregate production process and induced price changes that maximize private profits (high current prices with respect to the marginal production costs of stocked goods, not necessarily finished goods).

The model estimated captures here the episodes where inventory investment amplifies the response of GDP to large negative demand shocks (during the recessive phases of Peru's

²⁶ The first row of graphs includes the value of the coefficient corresponding to the estimated contemporaneous impact in matrix A (Table 4), which is typically greater (in absolute value) than the contemporaneous impact in the corresponding IRF due to the way it was calculated.



IMPULSE-RESPONSE FUNCTIONS FOR ESTIMATED INCREASES IN BINV AND GDP AFTER A SHOCK OF 1% IN PRDEM GROWTH

economy since the start of the 1990s, particularly the most recent one), the *demoderation* phenomenon mentioned in Section 1.

Limitations to inventory statistics in Peru²⁷ make it necessary to postpone a strict comparison of a new hypothesis

²⁷ Barrera (2009) employs 12-month average percentage changes to date the phases of business cycles in Peru's economy with monthly periodicity. Using those units avoids problems for measuring real monthly levels, making the monthly dates for peaks

expounded in the literature that the recent international crises explain most of the recent fluctuations in the inventory cycle (especially in exportable primary products) and therefore in the activity of an increasingly globalized economy such as Peru's (see Alessandria et al., 2010).²⁸ This paper provides indirect evidence to support this hypothesis.

5.4 Impulse Responses in PuDem

Figure 6 displays IRFs for the estimated growth in *BInv* and GDP after a shock of 1% in *PuDem* growth. In response to a positive structural shock in *PuDem* growth, *BInv* growth falls upon impact and subsequently remains unchanged until it increases 10 quarters ahead. On the other hand, GDP growth increases upon impact, then rises very slightly and falls lightly after which it exhibits a series of small falls and rebounds with the zero line as a ceiling.

Given the relative magnitudes, an increase in *PuDem* growth is absorbed by a significant fall in *BInv* growth and a small increase in GDP growth. The tendering process associated with government expenditure, very different on aggregate from the private expenditure process, can explain this behavior more in line with intuitive inventory management, but opposite to that resulting from a shock in *PrDem* (of the same sign).

and troughs more robust. Given those dates, if the coefficients (inventory investment)/GDP of recessionary phases in Peru are calculated using real quarterly flows in millions of 1994 soles, only the coefficient corresponding to the recession between December 1997 and August 1999 (1997M12-1999M8) will be valid.

²⁸ Disaggregating inventory investment into its typical components (inputs, work-in-progress and finished goods) is not feasible with data for Peru, and even less so with its external trade components (exports, imports and nontradeable goods). The latter disaggregation is used by Alessandria et al. (2010) for USA.





5.5 Impulse Responses in XDem

Figure 7 displays IRFs for the estimated growth in *BInv* and GDP after a shock of 1% in *XDem* growth. In response to a positive structural shock in *XDem* growth, *BInv* growth falls upon impact and then remains unchanged until it increases marginally 10 quarters ahead. Meanwhile, GDP growth decreases almost unnoticeably and subsequently posts a series of modest increases and decreases.

Figure 7





Given the relative magnitudes, an increase in *XDem* growth is absorbed by a fall in *BInv* growth as well as an imperceptible drop in GDP growth. With respect *BInv* growth, the response is qualitatively similar to the response to a positive structural shock in *PuDem*, meaning it is not possible to reject that the way sales of goods and services are conducted abroad has similar effects to those that stem from the way sales are made to the federal government on aggregate inventory management. In both cases, the magnitude of responses in GDP growth reflects the fact that GDP growth is not the main adjustment channel. Nonetheless, decreases in GDP growth in response to the shock in *XDem* can be understood as the impact of mining production dynamics (where production is reduced when external prices are high).

5.6 Observations

IRF calculations employ a projection without a shock that is not based on a stationary state. A comparison of this projection with the recent execution of aggregate demand components for the 2010Q2-2010Q4 (out of sample) was not encouraging, reflecting that the propagation of shocks during the last two years point to a scenario of an economic slowdown in the medium term.

IRF patterns do not follow a smooth transition as in the over-parameterized linear VAR models. For instance, those for *BInv* are reflected upon impact as well as 10 quarters after the shock to any component of *AgDem* (although with different signs), which is explained by different ways for contracting or demanding goods and services.²⁹ This lack of a smooth transition is normally obtained when exclusion restrictions are imposed (parsimony) on the parameters of a linear VAR model (see Lütkepohl, 2005). It could also result from the penalized log-likelihood (see Annex B) used when parsimoniously estimating a VARNN-ARCH model.

²⁹ Another explanation is that mechanisms associated to aggregate inventory management are not reflected so much in their conditional means (that serves to quantify them) as in their conditional second moments. In structural terms, more comprehensive inventory management includes risk factors associated to profits and losses. In econometric terms, it is possible that maximization of the penalized log-likelihood reflects the dominance of changes in the conditional covariance matrix over the quadratic errors of the conditional mean vector.

Finally, the absence of asymmetries in shock response with different signs or magnitudes might be a preliminary but robust result. Optimization of the penalized log-likelihood of a neural network model (see Annex B) is equivalent to a learning process, and this could be lengthy. Due to computing time restrictions, the optimization must be truncated after a large number of iterations, without the network having detected asymmetries. However, the *t*-Student distribution allows for discarding spurious asymmetries in conditional means, making it possible to state that the neural network has still not detected asymmetries in the data because they are not evident.

6. CONCLUSIONS AND OUTLOOKS

This paper econometrically approximates the potentially significant nonlinear effects (asymmetries) that inventory management exerts on production dynamics considering that its volatility varies over time. To that end, we decompose aggregate demand into three components (domestic public, domestic private and external).

The most important results are shown in terms of conditional covariances. Covariances (BInv, PuDem) and (BInv, XDem) are negative, reflecting the expected inverse relations when there are no incentives for holding inventories except production smoothing. Covariance (BInv, PrDem) is the only positive one, reflecting the expected direct relation when there are additional incentives besides smoothing GDP growth. In terms of contemporaneous relations, the only parameter statistically different from zero is consistent with the presence of such additional incentives. This parameter indicates that a positive shock in PrDem will be mainly absorbed by a more than proportional increase in the production rate shock, meaning there is an amplifier effect (demoderating) of demand shocks on the evolution of production that is explained by the inventory cycle. In fact, some of this faster production rate will be used for increased inventory accumulation, which will probably allow for maximizing profits when current prices are high with respect to the marginal production costs of stocked goods.

Another incentive for holding inventories stems from the need to have a nonfinancial asset that allows for offsetting shortterm borrowing incurred to cover production when demand is growing in the event such increased demand reverts. Precisely, given the symmetry found in IRFs, a negative shock in PrDem will be offset mainly by a slower rate of production, as well by decreases in the growth of inventory stocks (although to a lesser extent). This result might be consistent with inventory management that takes into account lags in the adjustment of the aggregate production process, as well as changes induced in prices that maximize private profits, particularly when current prices are high compared to the marginal production costs of stocked goods (not necessarily finished goods). In this regard, there are indications that the amplifier effect (demoderating) could be explained by the inventory cycle of raw materials or work-in-progress (although we do not have the data to prove this more specific hypothesis).

The model estimated partly captures episodes around the turning points of GDP in which inventory investment amplifies the response of GDP to large demand shocks. This paper, therefore, provides indirect evidence to support the hypothesis that recent international crises mostly explain recent fluctuations in the inventory cycle (especially for commodity exports) and therefore in the activity of an increasingly globalized economy such as Peru's (see Alessandria et al., 2010). This would provisionally explain the demoderation described in Section 1, particularly in the average incidence of inventory investment growth on real GDP growth during four recently observed recessions in Peru (before that generated as a consequence of the US crisis in 2007-2008; see Table 1).

There is clearly a need to include other potentially relevant variables (some of which are not available for Peru's economy, such as disaggregated inventory investment in raw materials, work-in-progress and finished goods). Given the absence of such disaggregated data, the results of this inventory investment model with aggregate data regarding production *stabilization* could represent a reference for more complete models that manage to include inventories of work-in-progress and raw materials (separate from finished goods) in conditional covariance index modelling. This would provide more appropriate evaluation of production stabilization in terms of conditional second moments, as well as an improvement in the capability of representing the structure of relationships in conditional means and, therefore, in the model's predictive capacity.

ANNEX

Annex A. Aggregate Stock of Inventories according to the Perpetual Inventory Method

In Peru's experience, shifts in inventory investment have contributed (amplified) recessionary phases since the start of the 1990s. With the US financial crisis (2007-2008), this amplification is more noteworthy, unfolding a demoderation phenomenon in contrast to the Great Moderation observed in the business cycles of the US economy (see introductory discussion). In any case, highly volatile inventory investment growth rates in Peru (see Table 2 in the main text) reveals the need for using a calibrated series of aggregate inventory stocks instead of a series of changes in inventory.

This annex explains the assumptions employed for calibrating a series for the aggregate stock of inventories. This is obtained based on inventory changes data through two quantitative assumptions: 1) initial inventory stocks and 2) the depreciation rate. Figure A.1 presents a set of alternative sequences with initial stocks of between 2,000 million and 13,500 million 1994 soles for the first quarter of 1990, as well as quarterly depreciation rates of between 0.0% and 3.6% (a rate of 2.4% corresponds to that of physical capital that depreciates in 10 years).







All these inventory stock sequences indicate that, before the international crisis of 2008 affected most economies in the region (2008Q3), Peru had been registering significant inventory accumulation that reached a peak in 2008Q4, just after the initial impact of the crisis was perceived in financial variables such as the exchange rate and interest rates (August 2008). In terms of inventory stocks, the impact of the crisis is evident since the start of 2009 in the form of an unprecedented deaccumulation in the available sample (1990Q4-2010Q1).

All deaccumulations associated to the financial crises of 1995, 1998-1999 and 2001 appear small in size and generally affect the evolution of inventory stocks cumulatively, for instance, when assuming a depreciation rate higher than that for physical capital (for instance, with a quarterly rate of 3.6%) and 2,000 million or 4,500 million of initial stock. If we wish to reduce the preponderance of the sharp accumulation and later deaccumulation of inventories associated to the international crisis of 2008 in the sample, the initial stock can be raised slightly to above 5,000 million, which would be qualitatively compatible with high inventory levels expected to be registered at the start of the 1990s.³⁰ This paper explicitly addresses the conditionality of all the results with respect to these two quantitative assumptions: *1*) initial inventory stock and *2*) depreciation rate.³¹

³⁰ Fujino (1960) refers to high levels of inventory stocks of finished goods as a percentage of demand in some Japanese industries in 1950 or 1951 due to speculation under the setting of the Korean war (June 1950 to July 1951). Japan provided military, logistical and medical support to the allied forces led by USA.

³¹ The results shown use a calibrated sequence of inventory stocks that assumes an initial balance of 2,000 million 1994 nuevos soles and a null depreciation rate (perpetual inventories).

Annex B. Estimation via Penalized Maximum Likelihood

Estimation of multiple time series models typically finds the problem of over parameterization unsurmountable. The usual strategies for tackling this problem have been elimination algorithms with stepwise and a data criteria sequence, thereby achieving parsimonious models.

Based on statistical applications to penalized regression problems in chemistry and biology (molecule and genotype structures), the literature on parameter shrinkage has re(emerged); in it a penalization function in them, is included which is added to the function that typically optimized in parameter estimation (GLS, GMM or MV).³²

In the case of MV estimation, the loss function minimized is the negative of log-likelihood, which we denote as $L(\theta)$, where θ is a parameter vector. In a system with multiple variables, this vector θ can be decomposed into two blocks: interceptors α and all other parameters β , to define the penalized loss function as

B.1
$$g(\theta) \equiv L(\theta) + P_{\lambda}(\beta),$$

where $P_{\lambda}(\beta)$ is one of the three penalized functions available in the literature (see McCann and Welsch, 2006, and Ulbricht and Tutz, 2007) that depend on tuning parameters λ_i (positive):

³² The typical MCO estimator minimizes $SSE(\tilde{\beta}) \equiv (y - x\tilde{\beta})'(y - x\tilde{\beta})$. To avoid a potential problem of multicollinearity, the ridge estimator $\tilde{\beta} \equiv [x'x + \lambda Q]^{-1}x'y$ was devised to minimize $SSER(\tilde{\beta}) \equiv SSE(\tilde{\beta}) + \lambda \tilde{\beta}' Q \tilde{\beta}$, where Q should be a positively defined arbitrary matrix and $\lambda > 0$ so the MCO estimator *regularizes* (see Firinguetti and Rubio, 2000, for references and a generalization). Returning to our context, a parsimonious estimator belongs to this same family of estimators because Q = I obtains the penalized version of $SSE(\tilde{\beta})$.

- 1) Lasso or L1 (strong zeros; Tibshirani, 1996), $P_{\lambda}(\beta) \equiv \lambda \sum_{i=1}^{q} |\beta_i|.$
- 2) Ridge or *L*2 (against over-parameterization), $P_{\lambda}(\beta) \equiv \lambda \sum_{i=1}^{q} \beta_{i}^{2}$.
- 3) Elastic network (*L1* and *L2*), $P_{\lambda}(\beta) \equiv \lambda_1 \sum_{i=1}^{q} |\beta_i| + \lambda_2 \sum_{i=1}^{q} \beta_i^2$.

The most direct reason for optimizing this new loss function is clearly that of estimating the parameters at the same time as selecting the specification (Fan and Li, 1999). This model selection is apparently more direct than the alternative of performing a series of hypothesis tests. Nonetheless, the main motivation is to reduce the mean squared error (MSE) of the sample. One well-known econometric result is that the MV estimator over-estimates the length of the true parameter vector when the regressors are not orthogonal amongst each other, causing significant bias in the MV estimator. Minimizing this bias led to the family of *ridge* estimators (see Fomby et al., 1984, pp. 300-302 and references), specifically an MV estimator with restrictions or penalties.

However, similarly to the *ridge* family of estimators (see note 29), it is necessary to determine tuning parameters $\lambda > 0$ through a set of estimations for different values of λ .³³

B.1 Tuning λ Parameters in VARNN-ARCH Models

We define the estimator we will use as

B.2
$$\theta(\lambda) \equiv \arg\min\{g(\theta)\}.$$

³³ The complexity of the resulting optimization problem for each fixed value of λ is considerably greater, meaning addressing it various times to fill a grid and thereby select the tuning parameters (and associated β parameters) is extremely costly in computational terms. For the simple case of a *lasso* regression, a group of algorithms has been proposed (see Wu and Lange, 2008).

Tuning parameters λ are basically Lagrange multipliers and are usually determined in such way that the asymptotic mean squared error (MSE) of estimator $\theta(\lambda \neq 0)$ is less than the asymptotic variance of the estimator of MV, $\theta(\lambda = 0)$. This determination is direct in a simple problem such as a linear regression, but generally requires, in the case of the *elastic network*, a search algorithm in an \mathbb{R}^2_{++} mesh with simulation at each point of it, a procedure too computationally costly for a VARNN-ARCH model.

The alternative is to define its optimization as a *weak appren*tice, i.e., (λ_1, λ_2) with large values to force small changes in each maximum likelihood iteration and thereby obtain more stable estimates (Ulbricht and Tutz, 2007).³⁴ The advantage of this likelihood penalization is that neural network training and pruning is performed in parallel, meaning the neural network can adapt for minimizing errors associated with pruning (see Reed, 1993). This alternative was the first to be used for a VAR-NN-ARCH model, without managing to converge after a large number of iterations.

After forcing very small changes with large values for (λ_1, λ_2) , we used ad hoc values based on the proposals of Fan and Li (1999), i.e.,

B.3
$$\lambda_i = \sqrt{2\log(nparam)},$$

where *nparam* is the total number of θ parameters in the model. This strategy did not manage convergence for an even higher number of iterations (three million). The results reported in this version of the paper use nonstructural parameters of the VARNN-ARCH estimated using this strategy.

³⁴ In fact, the nonlinear classification problems that are typical applications of neural networks, optimization of the objective function $L(\theta)$ is established around a set of desirable values, defining these regularization penalties and fixing parameters (λ_1, λ_2) through other criteria. See Jaakkola (2006).

B.2 Alternative to a Single Tuning Parameter

Finally, results were obtained with the truncated maximum (reaching the maximum number of iterations without converging) of the penalized likelihood function for a lasso function using the value of the previous equation for the single tuning parameter. These results have allowed for estimating the proposed contemporaneous structure and performing *provisional* tests on it (they would not be provisional if the required convergence had been achieved), which has been reflected in a lack of accuracy of the projections generated. Although convergence has not been produced after a prohibitive number of iterations, in this subsection we present an alternative tuning strategy proposed by Wang et al. (2007).

Wang et al. (2007) propose discarding the lasso penalty with a single tuning parameter due to the potentially significant bias it generates and using multiple tuning parameters, in fact, one for each parameter of the unpenalized likelihood function.

 $\lambda_i = \frac{\log(nparam)}{nparam\lambda_i}.$

Greater parametric complexity in the penalization function proposed in Wang et al. (2007) is addressed through a profitable strategy for estimating in a first stage all the tuning parameters for optimizing the unpenalized likelihood, and then using said estimates in a second stage of penalized likelihood optimization. Another advantage of this strategy is that it solves the problem of a lack of asymptotic characteristics required for performing statistical tests when there is only one tuning parameter.

B.4



Annex C. Evolution of Conditional Covariances (Standardized)











References

- Alessandria, George, Joseph Kaboski, and Virgiliu Midrigan (2010) "The Great Trade Collapse of 2008-09: An Inventory Adjustment?," *IMF Economic Review*, Vol. 58, No. 2, December, pp. 254-294.
- Barrera, Carlos (2009), Ciclos sectoriales de los negocios en el Perú e indicadores anticipados para el crecimiento del PIB no primario, Documento de Trabajo, No. 2009-013, Banco Central de Reserva del Perú.
- Barrera, Carlos (2010), *Redes neurales para predecir el tipo de cambio diario*, Documento de Trabajo, No. 2010-01, Banco Central de Reserva del Perú.
- Barrera, Carlos (2011), Impacto amplificador del ajuste de inventarios ante choques de demanda según especificaciones flexibles, Documento de Trabajo, No. 2011-009, Banco Central de Reserva del Perú.
- Blanchard, Olivier J. (1983), "The Production and Inventory Behavior of the American Automobile Industry," *Journal of Political Economy*, Vol. 91, No. 3, June, pp. 365-400.
- Blinder, Alan S. (1982), "Inventories and Sticky Prices: More on the Microfoundations of Macroeconomics," *American Economic Review*, Vol. 72, No. 3, June, pp. 334-348, https://www.jstor.org/stable/1831536>.
- Blinder, Alan S. (1986a), "Can the Production Smoothing Model of Inventory Behavior Be Saved?," *The Quarterly Journal of Economics*, Vol. 101, No. 3, August, pp. 431-453, https://doi.org/10.2307/1885691>.
- Blinder, Alan S. (1986b), "More on the Speed of Adjustment in Inventory Models," *Journal of Money, Credit, and Banking*, Vol. 18, No. 3, August, pp. 355-365, <DOI: 10.2307/1992386>.
- Blinder, Alan S., and Louis J. Maccini (1991), "Taking Stock: A Critical Assessment of Recent Research on Inventories," *Journal of Economic Perspectives*, Vol. 5, No. 1, pp. 73-96, <DOI: 10.1257/jep.5.1.73>.
- Callen, T. S., S. G. Hall, and S. G. B. Henry (1990), "Manufacturing Stocks: Expectations, Risk and Co-integration," *The Economic Journal*, Vol. 100, No. 402, September, pp. 756-772, <DOI: 10.2307/2233657>.

- Cuthbertson, Keith, and David Gasparro (1993), "The Determinants of Manufacturing Inventories in the UK," *The Economic Journal*, Vol. 103, No. 421, November, pp. 1479-1492, <DOI: 10.2307/2234478>.
- Davis, Steven J., and James A. Kahn (2008), "Interpreting the Great Moderation: Changes in the Volatility of Economic Activity at the Macro and Micro Levels," *Journal of Economic Perspectives*, Vol. 22, No. 4, pp. 155-180, <DOI: 10.1257/jep.22.4.155>.
- Ding, Zhuanxin, and Robert F. Engle (2001), "Large-scale Conditional Covariance Matrix Modeling, Estimation, and Testing," Academia Economic Papers, Vol. 29, No. 2, pp. 157-184.
- Dorffner, Georg (1996), "Neural Networks for Time Series Processing," *Neural Network World*, Vol. 4, No. 6, pp. 447-468.
- Eaton, Jonathan, Samuel Kortum, Brent Neiman, and John Romalis (2011), *Trade and the Global Recession*, NBER Working Paper, No. 16666, January, http://www.nber.org/papers/w16666>.
- Ebrahim, Shafiq K. (2000), Volatility Transmission between Foreign Exchange and Money Markets, Working Paper, No. 2000-16, Bank of Canada.
- Eggers, Andrew, and Yannis M. Ioannides (2006), "The Role of Output Composition in the Stabilization of US Output Growth," *Journal of Macroeconomics*, Vol. 28, No. 3, pp. 585-595, https://doi.org/10.1016/j.jmacro.2004.09.007>.
- Eichenbaum, Martin (1989), "Some Empirical Evidence on the Production Level and Production Cost Smoothing Models of Inventory Investment," *The American Economic Review*, Vol. 79, No. 4, September, pp. 853-864, https://www.jstor.org/stable/1827937>.
- Engle, Robert F., and Kenneth F. Kroner (1995), "Multivariate Simultaneous Generalized ARCH," *Econometric Theory*, Vol. 11, No. 1, February, pp. 122-150, https://doi.org/10.1017/S0266466600009063>.
- Fan, Jianqing, and Runze Li (1999), Variable Selection via Penalized Likelihood, unpublished working paper, Department of Statistics, UCLA and UNC.
- Firinguetti, Luis, and Hernán Rubio (2000), A Note on the Moments of Stochastic Shrinkage Parameters in Ridge Regression, Documentos de Trabajo, No. 65, Banco Central de Chile,.
- Fomby, Thomas B., Carter Hill, and Stanley R. Johnson (1984), Advanced Econometric Methods, Springer-Verlag, 624 pages.

- Fujino, Shozaburo (1960), "Some Aspects of Inventory Cycles," *The Review of Economics and Statistics*, Vol. 42, No. 2, May, pp. 203-209, <DOI: 10.2307/1926540>.
- Granger, Clive W. J., and Timo Teräsvirta (1993), *Modelling Nonlinear Economic Relations*, Oxford University Press, 200 pages.
- Hamilton, James (2008), "Macroeconomics and ARCH," paper presented at the Festschrift Conference in Honor of Robert F. Engle, San Diego (June 21).
- Jaakkola, Tommi (2006), Machine learning 6.867, Fall 2006, MIT OpenCourseWare.
- Kahn, James A. (1987), "Inventories and the Volatility of Production," *The American Economic Review*, Vol. 77, No. 4, September, pp. 667-679, https://www.jstor.org/stable/1814538>.
- Kahn, James A. (2008), Durable Goods Inventories and the Great Moderation, Federal Reserve Bank of New York Staff Reports, No. 325, May.
- Kahn, James A., Margaret M. McConnell, and Gabriel Pérez-Quirós (2002), "On the Causes of the Increased Stability of the US Economy," *Economic Policy Review*, Vol. 8, No. 1, May, pp. 183-202.
- Kahn, James A., and Margaret M. McConnell (2002), "Has Inventory Volatility Returned? A Look at the Current Cycle," *Current Issues in Economics and Finance*, Federal Reserve Bank of New York, Vol. 8, No. 5, May, pp. 1-6.
- Kawakatsu, Hiroyuki (2006), "Matrix Exponential GARCH," Journal of Econometrics, Vol. 134, No. 1, September, pp. 95-128, https://doi.org/10.1016/j.jeconom.2005.06.023>.
- Koop, Gary, M. Hashem Pesaran, and Simon M. Potter (1996), "Impulse Response Analysis in Nonlinear Multivariate Models," *Journal of Econometrics*, Vol. 74, No. 1, September, pp. 119-147, https://doi.org/10.1016/0304-4076(95)01753-4>.
- Kuan, Chung-Ming, and Tung Liu (1995), "Forecasting Exchange Rates Using Feedforward and Recurrent Neural Networks," *Journal of Applied Econometrics*, Vol. 10, No. 4, October, pp. 347-364, <https://doi.org/10.1002/jae.3950100403>.
- Lütkepohl, Helmut (2005), New Introduction to Multiple Time Series Analysis, Springer-Verlag, Berlin, 764 pages.
- McCann, Lauren, and Roy Welsch (2006), Simultaneous Robust Estimation and Variable Selection, unpublished working paper, Department of Statistics, MIT.

- McCarthy, Johathan, and Egon Zakrajšek (2007), "Inventory Dynamics and Business Cycles: What Has Changed?," *Journal of Money, Credit and Banking*, Vol. 39, No. 2-3, March-April, pp. 591-613, https://doi.org/10.1111/j.0022-2879.2007.00037>.
- McConnell, Margaret M., and Gabriel Pérez-Quirós (2000), "Output Fluctuations in the United States: What Has Changed since the Early 1980s?," *The American Economic Review*, Vol. 90, No. 5, December, pp. 1464-1476, <https://www.jstor. org/stable/2677860>.
- Metzler, Lloyd A. (1941), "The Nature and Stability of Inventory Cycles," *Review of Economics and Statistics*, Vol. 23, No. 3, August, pp. 113-129, <DOI: 10.2307/1927555>.
- Nelson, Daniel B. (1991), "Conditional Heteroskedasticity in Asset Returns: A New Approach," *Econometrica*, Vol. 59, No. 2, March, pp. 347-370, <DOI: 10.2307/2938260>.
- Ramey, Valerie A. (1991), "Nonconvex Costs and the Behavior of Inventories," *Journal of Political Economy*, Vol. 99, No. 2, April, pp. 306-334.
- Reed, Russell D. (1993), "Pruning Algorithms: A Survey," *IEEE Transactions on Neural Networks*, Vol. 4, No. 5, September, pp. 740 747, <DOI: 10.1109/72.248452>.
- Shachmurove, Yochanan (2002), Applying Artificial Neural Networks to Business, Economics and Finance, CARESS Working Paper, No. 02-08, Dept. of Economics, University of Pennsylvania.
- Sensier, Marianne (2003), "Inventories and Asymmetric Business Cycle Fluctuations in the UK: A Structural Approach," *Applied Economics*, Vol. 35, No. 4, pp. 387-402, https://doi.org/10.1080/00036840210128735>.
- Simon, John (2001), The Decline in Australian Output Volatility, Research Discussion Paper, Economic Research, No. 2001-01, Reserve Bank of Australia.
- Tibshirani, Robert (1996), "Regression Shrinkage and Selection via the Lasso," *Journal of the Royal Statistical Society*, Series B (Methodological), Vol. 58, No. 1, January, pp. 267-288, https://www.jstor.org/stable/2346178>.
- Ulbricht, Jan, and Gerhard Tutz (2007), *Boosting Correlation-based Penalization in Generalized Linear Models*, Technical Report, Department of Statistics, University of Munich, No. 009.2007.

- Wang, Hansheng, Guodong Li, and Guohua Jiang (2007), "Robust Regression Shrinkage and Consistent Variable Selection Through the LAD-Lasso," *Journal of Business and Economics Statistics*, Vol. 25, No. 3, July, pp. 347-355, https://doi.or g/10.1198/07350010600000251>.
- West, Kenneth D. (1990), "The Sources of Fluctuations in Aggregate Inventories and GNP," *The Quarterly Journal of Economics*, Vol. 105, No. 4, November, pp. 939-971, <DOI: 10.2307/2937880>.
- Wu, Tong Tong, and Kenneth Lange (2008), "Coordinate Descent Algorithms for Lasso Penalized Regression," Annals of Applied Statistics, Vol. 2, No.1, pp. 224-244.

CEMLA MEMBERS

Associated

Banco Central de Bolivia Banco Central do Brasil Banco Central de Chile Banco Central de Costa Rica Banco Central de Cuba Banco Central del Ecuador Banco de Guatemala Banco Central de Honduras Banco de México Banco Central de Nicaragua Banco Central del Paraguay Banco Central de Reserva del Perú Banco de la República (Colombia) Banco Central de la República Argentina Banco Central de la República Dominicana

Banco Central de Reserva de El Salvador Banco Central del Uruguay Banco Central de Venezuela Bank of Guyana Bank of Jamaica Banque de la République d'Haïti Cayman Islands Monetary Authority Central Bank of Barbados The Central Bank of The Bahamas Central Bank of Belize Central Bank of Trinidad and Tobago Centrale Bank van Aruba Centrale Bank van Curacao en Sint Maarten Centrale Bank van Suriname Eastern Caribbean Central Bank

Collaborating

Central banks

Banca d'Italia Banco de España Banco de Portugal Bangko Sentral ng Pilipinas Bank of Canada Banque de France Deutsche Bundesbank (Germany) European Central Bank

> Federal Reserve System (United States of America)

Sveriges Riksbank (Sweden) Swiss National Bank

Other institutions

Banco Centroamericano de Integración Económica Banco Latinoamericano de Comercio Exterior, S. A. CAF-Banco de Desarrollo de América Latina Comisión Nacional de Bancos y Seguros (Honduras) Fondo Latinoamericano de Reservas Superintendencia de Bancos (Dominican Republic) Superintendencia de Bancos de Guatemala Superintendencia de Bancos de Panamá Superintendencia de Bancos y Seguros (Ecuador) Superintendencia del Sistema Financiero (El Salvador)

> Turks and Caicos Islands Financial Services Commission

CEMLA

Assembly

Associated Central Banks (vox et votum) Collaborating Members (vox)

Board of Governors 2017-2019

President Mario Bergara, Banco Central del Uruguay

Associate Members

John A. Rolle, The Central Bank of The Bahamas Ilan Goldfajn, Banco Central do Brasil Mario Marcel, Banco Central de Chile Julio Roberto Suárez Guerra, Banco de Guatemala Alejandro Díaz de León Carrillo, Banco de México Calixto José Ortega Sánchez, Banco Central de Venezuela

Collaborating Members

Pablo Hernández de Cos, Banco de España Jerome Powell, The Federal Reserve System

Management

Fernando Tenjo Galarza General Director

Dalmir Louzada Deputy General Director

Monetaria is published twice a year (Spring, Autumn) in Spanish and English (two volumes) by the Centro de Estudios Monetarios Latinoamericanos, Durango 54, colonia Roma Norte, delegación Cuauhtémoc, 06700 Ciudad de México. Printed in Master Copy, S. A. de C. V., Calle Plásticos núm. 84, local 2, Ala sur, colonia Industrial Alce Blanco, 53370, Naucalpan de Juárez, Estado de México. ISSN 0185-1136 (300 copies).

CENTER FOR LATIN AMERICAN MONETARY STUDIES

Regional Association of Central Banks

www.cemla.org

